UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

[X] QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended April 30, 2002

OR

[_] TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from _____ to _____

Commission file number 0-27022

OPTICAL CABLE CORPORATION (Exact name of registrant as specified in its charter)

Virginia (State or other jurisdiction of incorporation or organization) 54-1237042 (I.R.S. Employer Identification No.)

5290 Concourse Drive Roanoke, Virginia 24019 (Address of principal executive offices, including zip code)

(540) 265-0690 (Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days.

(1)	TEP	Δ	NO	(2)	res	Δ	NO

As of June 7, 2002, 55,431,279 shares of the registrant's Common Stock, no par value, were outstanding.

PART I. FINANCIAL INFORMATION

	Item 1.	Financial Statements	
		Condensed Balance Sheets - April 30, 2002 and October 31, 2001	2
		Condensed Statements of Operations - Three Months and Six Months Ended April 30, 2002 and 2001	3
		Condensed Statement of Changes in Stockholders' Equity - Six Months Ended April 30, 2002	4
		Condensed Statements of Cash Flows - Six Months Ended April 30, 2002 and 2001	5
		Condensed Notes to Condensed Financial Statements	6-12
	Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	13-25
	Item 3.	Quantitative and Qualitative Disclosures About Market Risk	26
PART II.	OTHER IN	FORMATION	
	Item 1.	Legal Proceedings	27
	Item 4.	Submission of Matters to a Vote of Security Holders	28
	Item 5.	Other Information	29
	Item 6.	Exhibits and Reports on Form 8-K	29

SIGNATURES

Page

Item 1. Financial Statements

OPTICAL CABLE CORPORATION Condensed Balance Sheets

Assets April 10, 2002 October 31, 2001 Current assets: Trade accounts receivables frade accounts receivables for 070,2006 at April 30, 2002 and 5572,853 at October 31, 2001 Income taxes refundable of 670,2006 at April 30, 2002 and 5572,853 at October 31, 2001 Three taxes refundable of 670,000 Inventories prepaid expenses of 670,000 Inventories prepaid expenses total current isolities: Current liabilities Total assets Current isolities (22,004,164 28,811,974 220,04,164 28,811,974 220,04,164 28,811,974 220,04,164 28,811,974 220,04,164 28,811,974 220,04,164 28,811,974 220,04,164 28,811,974 220,07,113 36,733 220,713 33,801 Current liabilities Total assets Current liabilities Current liabilities Total current liabilities for a for a payable to bank corrent once payable to bank conserved and outstanding Common stock, no par value, authorized 100,000,000 shares; instal index and contingencies \$,474 28,577,857 27,865,228 28,583,331 27,865,228 28,583,331 27,865,228		(Unaudited)			
Cash and cash equivalents \$ 133,155 \$ 2,087,608 Trada accounts receivable, net of allowance for doubtful accounts of \$702,086 at April 30, 2002 and \$572,853 at October 31, 2001 7,979,203 10,678,214 Income taxes refundable 440,500 371,656 Other receivables 34,543 35,018 Income meployees, net of allowance for uncollectible advances of \$70,000 34,543 35,018 Prepaid expenses 234,441 185,831 Deferred income taxes 222,004,164 28,811,974 Other assets, net 270,711 367,469 Property and equipment, net 21,207,712 12,267,712 Deferred income taxes 835,321,973 \$ 42,798,297 Current liabilities 3,019,983 5,537,313 Accounts payable and accrued expenses 3,665,470 14,606,516 Noncurrent notes payable to bank 2,607,403 Accure taxets anding 3,665,470 14,606,516 Noncurrent note payable to bank 2,605,470 14,606,516 Noncurrent note payable, authorized 1,000,000 shares; issued and outstanding 55,431,279 shares 5,474	Assets		April 30,		October 31, 2001
Cash and cash equivalents \$ 133,155 \$ 2,087,608 Trada accounts receivable, net of allowance for doubtful accounts of \$702,086 at April 30, 2002 and \$572,853 at October 31, 2001 7,979,203 10,678,214 Income taxes refundable 440,500 371,656 Other receivables 34,543 35,018 Income meployees, net of allowance for uncollectible advances of \$70,000 34,543 35,018 Prepaid expenses 234,441 185,831 Deferred income taxes 222,004,164 28,811,974 Other assets, net 270,711 367,469 Property and equipment, net 21,207,712 12,267,712 Deferred income taxes 835,321,973 \$ 42,798,297 Current liabilities 3,019,983 5,537,313 Accounts payable and accrued expenses 3,665,470 14,606,516 Noncurrent notes payable to bank 2,607,403 Accure taxets anding 3,665,470 14,606,516 Noncurrent note payable to bank 2,605,470 14,606,516 Noncurrent note payable, authorized 1,000,000 shares; issued and outstanding 55,431,279 shares 5,474					
Trade accounts receivable, net of allowance for doubtful accounts of \$702,086 at April 30, 2002 and \$572,853 at October 31, 2001 Income taxes refundable 7,979,203 10,678,214 Income taxes refundable 978,835 1,108,007 Other receivables 440,500 371,656 Due from employees, net of allowance for uncollectible advances of \$70,000 34,543 35,018 Inventories 11,921,107 14,084,931 Prepaid expenses 224,044,164 28,811,974 Other assets 222,004,164 28,811,974 Other assets 222,004,164 28,811,974 Property and equipment, net 12,207,125 12,665,053 Deferred income taxes 33,973 93,801 Current notes payable to bank \$ Accounts payable and accrued expenses Accrued compensation and payroll taxes 3,603,470 14,606,516 Noncurrent note payable to bank \$ Noncurrent liabilities 3,603,470 14,606,516 Noncurrent liabilities 3,603,470 14,606,516 Noncurrent liabilities 6,738,642 14,933,069 Shareholders' equity:		Ś	133,155	S	2 . 087 . 608
Income taxes refundable 978,835 1,108,007 Other receivables 440,500 371,656 Due from employees, net of allowance for uncollectible advances of \$70,000 34,543 35,018 Inventories 11,921,107 14,084,931 Propaid expenses 234,441 185,831 Deferred income taxes 220,04,164 28,11,974 Other assets 22,004,164 28,811,974 Other assets 22,07,125 12,685,053 Deferred income taxes 833,973 933,801 Total current assets 9,35,321,973 \$ 42,798,297 Liabilities and Stockholders' Equity 5 - \$ 8,271,000 Current notes payable to bank Accounts payable and accrued expenses Accounts payable and accrued expenses 3,019,983 5537,313 Noncurrent note payable to bank 2,970,236 - 14,606,516 Other liabilities 3,603,470 14,606,516 - Noncurrent note payable to bank 2,970,236 - - Total current liabilities - - - - Noncurrent note payable to bank 2,970,236 - - - <		т	100,100	т	2,000,0000
Other receivables 440,500 371,656 Due from employees, net of allowance for uncollectible advances of 570,000 34,543 35,018 Inventories 34,543 35,018 Prepaid expenses 234,441 185,831 Deferred income taxes 234,441 185,831 Deferred income taxes 234,441 185,831 Total current assets 234,441 185,831 Other assets, net 22,004,164 28,611,974 Property and equipment, net 12,207,125 12,685,053 Deferred income taxes \$35,321,973 \$ 42,798,297 Liabilities: \$3,019,983 \$5,737,313 Current hotes payable to bank \$ 3,019,983 \$5,537,313 Accoued compensation and payroll taxes \$3,019,983 \$26,553 Total current liabilities 3,605,470 14,606,516 Noncurrent note payable to bank \$3,605,470 14,606,516 Noncurrent note payable to bank \$3,019,963 326,553 Total current liabilities \$6,738,642 14,933,069 Current notes payable to bank \$6,738,6			7,979,203		10,678,214
Due from employees, net of allowance for uncollectible advances of \$70,000 34,543 35,018 Inventories 11,921,107 14,084,931 Prepaid expenses 234,441 185,931 Deferred income taxes 282,380 260,709 Total current assets 22,004,164 28,811,974 Other assets, net 22,004,164 28,811,974 Property and equipment, net 270,711 367,469 Promerty and equipment, net 12,207,125 12,685,053 Deferred income taxes 33,973 933,801 Total assets 53,321,973 542,798,297 Liabilities and Stockholders' Equity 53,321,973 542,798,297 Current liabilities 5,019,983 5,537,313 Accourse payable to bank \$ Accourse payable to bank \$ Accourse triabilities 3,605,470 14,666,516 Other liabilities 6,738,642 14,933,069 Total current liabilities 6,738,642 14,933,069 Shareholders' equity:			978,835		1,108,007
of \$70,000 34,543 35,018 Inventories 11,921,107 14,084,931 Deferred income taxes 234,441 185,831 Deferred income taxes 234,441 185,831 Total current assets 234,441 185,831 Other assets, net 270,071 367,469 Property and equipment, net 12,207,125 12,685,053 Deferred income taxes 839,973 933,801 Total assets \$35,321,973 \$ 42,798,297 Liabilities and Stockholders' Equity Current liabilities Current notes payable and accrued expenses 3,605,470 14,606,516 Accounts payable and accrued expenses 3,605,470 14,606,516 Noncurrent note payable to bank \$ \$ 555,487 758,203 Total current liabilities Shareholders' equity: \$ 6,738,642 14,933,069 Properred stock, no par value, authorized 1,000,000 shares; issued and outstanding 5,474 Commitments and contigencies 28,597,867 27,865,228 Commitments and cou			440,500		371 , 656
Prepaid expenses 234,441 185,831 Deferred income taxes 282,380 260,709 Total current assets 22,004,164 28,811,974 Other assets, net 270,711 367,469 Property and equipment, net 23,973 933,801 Deferred income taxes 839,973 933,801 Current liabilities \$35,321,973 \$42,798,297 Liabilities \$3,019,983 5,537,313 Current notes payable to bank \$					05.040
Prepaid expenses 234,441 185,831 Deferred income taxes 282,380 260,709 Total current assets 22,004,164 28,811,974 Other assets, net 270,711 367,469 Property and equipment, net 23,973 933,801 Deferred income taxes 839,973 933,801 Current liabilities \$35,321,973 \$42,798,297 Liabilities \$3,019,983 5,537,313 Current notes payable to bank \$			34,543		35,018
Total current assets 22,004,164 28,811,974 Other assets, net 270,711 367,469 Property and equipment, net 12,207,125 12,685,053 Deferred income taxes 839,973 933,801 Total assets \$ 35,321,973 \$ 42,798,297 Liabilities and Stockholders' Equity \$ 35,321,973 \$ 42,798,297 Current liabilities \$ 3,019,983 5,537,313 Accounts payable and accrued expenses 3,019,983 5,537,313 Accounts payable and accrued expenses 3,605,470 14,606,516 Noncurrent note payable to bank \$ 162,936 326,553 Total current liabilities 3,605,470 14,606,516 Noncurrent note payable to bank \$ 6,738,642 14,933,069 Total liabilities 6,738,642 14,933,069 Shareholders' equity: 5,474 Preferred stock, no par value, authorized 10,000,000 shares; none issued and outstanding 55,431,279 shares 5,474 Issued and outstanding 55,431,279 shares 28,577,857 27,865,228 Common stock, no par value, authorized 100,000,000 shares; issued and outsta					105 021
Total current assets 22,004,164 28,811,974 Other assets, net 270,711 367,469 Property and equipment, net 12,207,125 12,685,053 Deferred income taxes 839,973 933,801 Total assets \$ 35,321,973 \$ 42,798,297 Liabilities and Stockholders' Equity \$ 35,321,973 \$ 42,798,297 Current liabilities \$ 3,019,983 5,537,313 Accounts payable and accrued expenses 3,019,983 5,537,313 Accounts payable and accrued expenses 3,605,470 14,606,516 Noncurrent note payable to bank \$ 162,936 326,553 Total current liabilities 3,605,470 14,606,516 Noncurrent note payable to bank \$ 6,738,642 14,933,069 Total liabilities 6,738,642 14,933,069 Shareholders' equity: 5,474 Preferred stock, no par value, authorized 10,000,000 shares; none issued and outstanding 55,431,279 shares 5,474 Issued and outstanding 55,431,279 shares 28,577,857 27,865,228 Common stock, no par value, authorized 100,000,000 shares; issued and outsta					10J,031 260 709
Total current assets 22,004,164 28,811,974 Other assets, net property and equipment, net Deferred income taxes 367,469 367,469 Deferred income taxes 333,973 933,801 Total assets 833,973 \$ 35,321,973 \$ 42,798,297 Liabilities and Stockholders' Equity 5 35,321,973 \$ 42,798,297 Current liabilities: \$ 35,321,973 \$ 42,798,297 Current notes payable to bank Accounts payable and accured expenses Accrued compensation and payroll taxes \$ \$ 8,271,000 Noncurrent note payable to bank Accounts payable to bank \$ 2,970,236 Total current liabilities 3,605,470 14,606,516 Noncurrent note payable to bank 2,970,236 Total liabilities 6,738,642 14,933,069 Shareholders' equity: Freferred stock, no par value, authorized 10,000,000 shares; none issued and outstanding 5,474 Common stock, no par value, authorized 100,000,000 shares; issued and outstanding 55,431,279 shares 5,474 Total shareholders' equity 28,583,331 27,865,228 Commitments and contingencies 28,583,331	Defeifed fillome taxes				
Total assets35,321,97342,798,297Liabilities and Stockholders' Equity	Total current assets		22,004,164		28,811,974
Total assets35,321,97342,798,297Liabilities and Stockholders' Equity	Other assets, net		270,711		367,469
Total assets35,321,97342,798,297Liabilities and Stockholders' Equity	Property and equipment, net		12,207,125		12,685,053
Total assets35,321,97342,798,297Liabilities and Stockholders' Equity	Deferred income taxes		839,973		933,801
Liabilities and Stockholders' Equity Current liabilities: Current notes payable to bank Accounts payable and accrued expenses Accrued compensation and payroll taxes Total current liabilities Total current liabilities Total liabilities Total liabilities Total liabilities Total shareholders' equity Total liabilities and shareholders' equity Total liabilities and shareholders' equity Total liabilities and shareholders' equity Liabilities and shareholders' equity Current notes payable to bank Current notes payable to bank Current note payable to bank Current note payable to bank Current liabilities Total liabilities Total liabilities Total liabilities Total liabilities and shareholders' equity Current notes Total liabilities and shareholders' equity Current notes Current liabilities and shareholders' equity Current notes Current notes Current notes Current liabilities and shareholders' equity Current notes Current liabilities and shareholders' equity Current notes Current notes Current liabilities and shareholders' equity Current notes Current notes Current notes Current liabilities and shareholders' equity Current notes Current liabilities Current notes Current n	Total assets	 \$	35,321,973	 \$	42,798,297
Current notes payable to bank Accounts payable and accrued expenses Accrued compensation and payroll taxes\$ \$ 8,271,000 3,019,983\$,537,313 5,537,313 5,85,487Total current liabilities3,605,47014,606,516 2,970,236Noncurrent note payable to bank Other liabilities3,605,47014,606,516 2,970,236Total liabilities6,738,64214,933,069Shareholders' equity: Preferred stock, no par value, authorized 1,000,000 shares; issued and outstanding Common stock, no par value, authorized 100,000 shares; issued and outstanding 55,431,279 sharesTotal shareholders' equity28,583,33127,865,228Total shareholders' equity\$ 35,321,973\$ 42,798,297	Liabilities and Stockholders' Equity	==			
Current notes payable to bank Accounts payable and accrued expenses Accrued compensation and payroll taxes\$ \$ 8,271,000 3,019,983\$,537,313 5,537,313 798,203Total current liabilities3,605,47014,606,516 2,970,236 Noncurrent note payable to bank Other liabilities3,605,47014,606,516 2,970,236 Total liabilities6,738,64214,933,069Shareholders' equity: Preferred stock, no par value, authorized 1,000,000 shares; issued and outstanding Common stock, no par value, authorized 100,000 shares; issued and outstanding 55,431,279 shares 28,577,857 27,865,228Total shareholders' equity28,583,33127,865,228Total liabilities and shareholders' equity\$ 35,321,973\$ 42,798,297	Current liabilities.				
Accounts payable and accrued expenses Accrued compensation and payroll taxes3,019,983 585,4875,537,313 798,203Total current liabilities3,605,470 2,970,23614,606,516 2,970,236Noncurrent note payable to bank Other liabilities3,605,470 2,970,23614,606,516 2,970,236Total liabilities6,738,642 2,970,23614,933,069Shareholders' equity: Preferred stock, no par value, authorized 1,000,000 shares; none issued and outstanding Common stock, no par value, authorized 100,000,000 shares; issued and outstanding 55,431,279 shares Retained earningsTotal shareholders' equity28,577,857 27,865,22827,865,228Total liabilities and shareholders' equity\$ 35,321,973 \$ 42,798,29742,798,297		Ś		Ś	8.271.000
Total current liabilities3,605,47014,606,516Noncurrent note payable to bank2,970,236Other liabilities162,936326,553Total liabilities6,738,64214,933,069Shareholders' equity:6,738,64214,933,069Preferred stock, no par value, authorized 1,000,000 shares; none issued and outstandingCommon stock, no par value, authorized 100,000,000 shares; issued and outstanding 55,431,279 shares5,474Retained earningsTotal shareholders' equity28,583,33127,865,228Commitments and contingenciesTotal liabilities and shareholders' equity\$ 35,321,973\$ 42,798,297		т	3.019.983	Ŧ	5,537,313
Total current liabilities3,605,47014,606,516Noncurrent note payable to bank2,970,236Other liabilities162,936326,553Total liabilities6,738,64214,933,069Shareholders' equity:6,738,64214,933,069Preferred stock, no par value, authorized 1,000,000 shares; none issued and outstandingCommon stock, no par value, authorized 100,000,000 shares; issued and outstanding 55,431,279 shares5,474Retained earningsTotal shareholders' equity28,577,85727,865,228Commitments and contingenciesTotal liabilities and shareholders' equity\$ 35,321,973\$ 42,798,297			585,487		798,203
Total liabilities6,738,64214,933,069Shareholders' equity: Preferred stock, no par value, authorized 1,000,000 shares; none issued and outstanding Common stock, no par value, authorized 100,000,000 shares; issued and outstanding 55,431,279 sharesRetained earnings5,474Total shareholders' equity28,583,33127,865,228Total shareholders' equity\$ 35,321,973\$ 42,798,297					
Total liabilities6,738,64214,933,069Shareholders' equity: Preferred stock, no par value, authorized 1,000,000 shares; none issued and outstanding Common stock, no par value, authorized 100,000,000 shares; issued and outstanding 55,431,279 sharesRetained earnings5,474Total shareholders' equity28,583,33127,865,228Total shareholders' equity\$ 35,321,973\$ 42,798,297	Total current liabilities		3,605,470		14,606,516
Total liabilities6,738,64214,933,069Shareholders' equity: Preferred stock, no par value, authorized 1,000,000 shares; none issued and outstanding Common stock, no par value, authorized 100,000,000 shares; issued and outstanding 55,431,279 sharesRetained earnings5,474Total shareholders' equity28,583,33127,865,228Total shareholders' equity\$ 35,321,973\$ 42,798,297	Noncurrent note payable to bank		2,970,236		
Total liabilities6,738,64214,933,069Shareholders' equity: Preferred stock, no par value, authorized 1,000,000 shares; none issued and outstanding Common stock, no par value, authorized 100,000,000 shares; issued and outstanding 55,431,279 sharesRetained earnings5,474Total shareholders' equity28,583,33127,865,228Total shareholders' equity\$ 35,321,973\$ 42,798,297	Other liabilities		162,936		326,553
Shareholders' equity: Preferred stock, no par value, authorized 1,000,000 shares; none issued and outstanding Common stock, no par value, authorized 100,000,000 shares; issued and outstanding 55,431,279 shares Retained earnings Total shareholders' equity Commitments and contingencies Total liabilities and shareholders' equity \$ 35,321,973 \$ 42,798,297	Total liabilities		6,738,642		14,933,069
Preferred stock, no par value, authorized 1,000,000 shares; none issued and outstanding Common stock, no par value, authorized 100,000,000 shares; issued and outstanding 55,431,279 shares 5,474 Retained earnings 28,577,857 27,865,228 Total shareholders' equity 28,583,331 27,865,228 Commitments and contingencies 28,583,331 27,865,228 Total liabilities and shareholders' equity \$ 35,321,973 \$ 42,798,297	Shareholders' equity:				
issued and outstanding Common stock, no par value, authorized 100,000,000 shares; issued and outstanding 55,431,279 shares 5,474 Retained earnings 28,577,857 27,865,228 Total shareholders' equity 28,583,331 27,865,228 Commitments and contingencies 28,583,331 27,865,228 Total liabilities and shareholders' equity \$ 35,321,973 \$ 42,798,297	1 1				
Common stock, no par value, authorized 100,000,000 shares; issued and outstanding 55,431,279 shares 5,474 Retained earnings 28,577,857 27,865,228 Total shareholders' equity 28,583,331 27,865,228 Commitments and contingencies 20,500 \$ 35,321,973 \$ 42,798,297					
Retained earnings28,577,85727,865,228Total shareholders' equity28,583,33127,865,228Commitments and contingencies28,583,33127,865,228Total liabilities and shareholders' equity\$ 35,321,973\$ 42,798,297					
Total shareholders' equity 28,583,331 27,865,228 Commitments and contingencies Total liabilities and shareholders' equity \$ 35,321,973 \$ 42,798,297			5,474		
Total shareholders' equity 28,583,331 27,865,228 Commitments and contingencies Total liabilities and shareholders' equity \$ 35,321,973 \$ 42,798,297	Retained earnings		28,577,857		27,865,228
Commitments and contingencies Total liabilities and shareholders' equity \$ 35,321,973 \$ 42,798,297	Total shareholders' equity				
Total liabilities and shareholders' equity \$ 35,321,973 \$ 42,798,297			,,,		_ , , 220
	Total liabilities and shareholders' equity	ş ==			

See accompanying condensed notes to condensed financial statements.

OPTICAL CABLE CORPORATION Condensed Statements of Operations (Unaudited)

	Three Months Ended April 30,			Six Months Ended April 30,				
	20	02 		2001		2002		2001
Net sales Cost of goods sold			9			22,307,039 14,202,980		
Gross profit Selling, general and administrative expenses	,		7	,828,473		8,104,059 6,899,895		15,706,742 7,674,120
Income from operations	21	4,344		,082,167		1,204,164		8,032,622
Other income (expense): Losses on trading securities, net Interest income Interest expense Other, net		 3,494 2,139) 983		9,066 (98,213)		20,646 (126,843) 1,167		24,830
Other expense, net	(4	7 , 662)	(5	,236,430)		(105,030)		(9,363,213)
Income (loss) before income tax expense Income tax expense			2			1,099,134 386,505		(1,330,591) 2,441,950
Net income (loss)	\$ 10	8,177	\$ (3	,657,928)	\$	712,629	\$	(3,772,541)
Net income (loss) per share: Basic and diluted	\$ =======		\$ =====	(0.06)		0.01	\$	(0.07)

See accompanying condensed notes to condensed financial statements.

OPTICAL CABLE CORPORATION Condensed Statement of Changes in Shareholders' Equity (Unaudited)

	Six Months Ended April 30, 2002								
	Common Stock				Retained		Total Stockholders'	rs'	
	Shares		Amount		Earnings		Equity		
Balances at October 31, 2001	55,431,279	\$		\$	27,865,228	\$	27,865,228		
Stock-based compensation Net income			5,474		 712,629		5,474 712,629		
Balances at April 30, 2002	55,431,279	\$	5,474	 \$ == =	28,577,857	\$	28,583,331		

See accompanying condensed notes to condensed financial statements.

OPTICAL CABLE CORPORATION Condensed Statements of Cash Flows (Unaudited)

		Six Months Ended April 30,		
		2002		2001
Cash flows from operating activities.				
Cash flows from operating activities: Net income (loss)	\$	712 620	ċ	(3,772,541)
Adjustments to reconcile net income (loss) to net cash provided	Ŷ	112,029	Ŷ	(3,772,341)
by operating activities:				
Depreciation, amortization and accretion		597,057		475.861
Bad debt expense		133,263		475,861 494,203
Deferred income tax expense		72,157		698.561
Stock-based compensation expense		5,474		698,561 68,385
Unrealized gains on trading securities, net				(183,191)
Decrease in trading securities				13,135,671
Decrease in payable to investment broker related to securities				13,133,071
trading				(4,550,275)
(Increase) decrease in:				(4,550,275)
Trade accounts receivable		2,565,748		(5,064,599)
Income taxes refundable		129,172		1,162,118
Other receivables		(68,844)		(22,001)
Due from employees		475		(22,001)
Inventories		2.163.824		975 (1,245,391)
Prepaid expenses		(48,610)		(21,783)
Increase (decrease) in:		(40,010)		(21, 103)
Accounts payable and accrued expenses		(2,644,513)		878,292
Accrued compensation and payroll taxes				
Income taxes payable		(212, /10)		(101,203)
Income taxes payable	_			(101,205) 444,071
Net cash provided by operating activities		3,405,116		2,397,151
	-			
Cash flows from investing activities:		(150 560)		(1 100 101)
Purchase of property and equipment		(152,563)		(1,183,171)
Increase in cash surrender value of life insurance				(54,085)
Receipt of cash surrender value of life insurance		367,469		
Net cash provided by (used in) investing activities	_	214,906		(1,237,256)
Cash flows from financing activities:	_			
Repayment of notes payable to bank, net		(5,300,764)		
Payments for financing costs		(273,711)		
Repurchase of common stock		(2,3,711)		 (1,578,109)
Proceeds from exercise of employee stock options				288,178
Freedo from exercise of employee stock options	_			
Net cash used in financing activities				(1,289,931)
Net decrease in cash and cash equivalents		(1,954,453)		(130,036)
Cash and cash equivalents at beginning of period		2,087,608		1,458,896
Cash and cash equivalents at end of period	\$	133,155	\$	1,328,860
	-			

See accompanying condensed notes to condensed financial statements.

(1) General

The accompanying unaudited condensed financial statements of Optical Cable Corporation (the Company) have been prepared in accordance with accounting principles generally accepted in the United States of America for interim financial reporting information and the instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all of the information and notes required by accounting principles generally accepted in the United States of America for complete financial statements. In the opinion of management, all material adjustments (consisting of normal recurring accruals) considered necessary for a fair presentation have been included. Operating results for the six months ended April 30, 2002 are not necessarily indicative of the results that may be expected for the fiscal year ending October 31, 2002. The unaudited condensed financial statements and condensed notes are presented as permitted by Form 10-Q and do not contain certain information included in the Company's annual financial statements and notes. For further information, refer to the financial statements and notes thereto included in the Company's annual report on Form 10-K for the fiscal year ended October 31, 2001.

(2) Allowance for Doubtful Accounts for Trade Accounts Receivable

A summary of changes in the allowance for doubtful accounts for trade accounts receivable for the six months ended April 30, 2002 and 2001 follows:

	Six Months Ended April 30,				
	2002	2001			
Balance at beginning of period Bad debt expense Losses charged to allowance Recoveries added to allowance	\$ 572,853 133,263 (4,030) 	\$ 1,909,069 494,203 (2,191,105) 983			
Balance at end of period	\$ 702,086	\$ 213,150			

One of the Company's two major distributors filed for liquidation under bankruptcy laws in January 2001. As of October 31, 2000, the Company specifically reserved approximately \$1,772,000 for estimated uncollectible accounts receivable from this distributor. As of January 31, 2001, the Company wrote off that \$1,772,000 reserve, as well as an additional bad debt reserve related to this distributor of approximately \$419,000 incurred during the first quarter of fiscal year 2001, for a total write-off of approximately \$2,191,000 for estimated uncollectible accounts receivable from this distributor for the six months ended April 30, 2001. There were no net sales attributed to this distributor subsequent to the first quarter of fiscal year 2001.

6

(3) Inventories

Inventories at April 30, 2002 and October 31, 2001, consisted of the following:

		April 30, 2002		October 31, 2001
Finished goods Work in process Raw materials Production supplies	Ş	5,215,605 1,938,160 4,704,434 62,908	Ş	4,328,379 3,064,975 6,641,985 49,592
	 S ===	11,921,107	\$ ===	14,084,931

(4) Notes Payable to Bank

Under a loan agreement with its bank dated March 10, 1999 and amended on October 30, 2001, the Company had a \$5 million secured revolving line of credit and a \$4.5 million secured revolving line of credit. As of October 31, 2001, the Company had combined outstanding borrowings under these lines of credit in the amount of \$8,271,000 with \$1,229,000 unused and available.

The lines of credit bore interest at 1.50% above the monthly LIBOR rate (3.79% as of October 31, 2001) and were equally and ratably collateralized by the Company's accounts receivable, contract rights, inventory, furniture and fixtures, machinery and equipment and general intangibles.

On April 25, 2002, the Company closed its new credit facility with Wachovia Bank, National Association (formerly First Union National Bank). The new three-year credit facility provides up to a maximum of \$25.0 million and replaces the Company's previous \$9.5 million credit facility described above.

The new credit facility bears interest at three-quarters of one percent (0.75%) per annum above the prime rate (5.5% as of April 30, 2002) and may be reduced by one-quarter of one percent (0.25%) upon meeting certain fixed charge coverage ratio requirements. The facility also provides a LIBOR based rate at the Company's option. The facility is collateralized by all of the Company's tangible and intangible assets. Borrowings under the credit facility are subject to certain coverage ratios, advance limits and qualifications that are applied to the Company's accounts receivable, inventory and fixed assets. The Company's ability to access the full amount of the credit facility will depend on the future growth of the Company's borrowing base. As of April 30, 2002, the Company had outstanding borrowings under the new credit facility in the amount of \$2,970,236, with \$5,098,264 unused and available. The outstanding balance on the credit facility has been reflected as noncurrent based on the scheduled maturity of the credit facility.

In connection with obtaining the new credit facility described above, the Company incurred various costs totaling \$273,711. These financing costs have been deferred and are included in other assets, net in the accompanying condensed balance sheet as of April 30, 2002. These deferred financing costs are being amortized to interest expense using the straight-line method over the life of the credit facility, which approximates the effective interest method.

7

(5) Stock Option Plan

The Company applies the provisions of Accounting Principles Board (APB) Opinion No. 25, Accounting for Stock Issued to Employees, and related interpretations for employee stock option grants and SFAS No. 123, Accounting for Stock-Based Compensation and EITF Issue No. 96-18, Accounting for Equity Instruments That Are Issued to Other Than Employees for Acquiring, or in Conjunction with Selling, Goods or Services, for nonemployee stock option grants. Stock option activity during the six months ended April 30, 2002 was as follows:

Number of Shares	Weighted-Average Exercise Price
1,013,562 2,743,475	\$6.97 0.93
(32,250)	3.78
3,724,787	\$2.55
	Shares 1,013,562 2,743,475 (32,250)

At April 30, 2002, there were approximately 1,600,000 additional shares available for grant under the Plan.

Effective November 16, 2001, the Board of Directors approved grants of stock options for a total of 250,000 shares with a per share exercise price of \$1.25 equal to the closing price of the Company's common stock on the date of grant. These grants were made to executive officers of the Company and vest in equal monthly installments over four years.

Effective February 12, 2002, the Board of Directors approved grants of stock options to purchase a total of 24,975 shares of the Company's common stock at an exercise price of \$0.89 per share, the closing price at the date of grant. These grants were made to those independent members of the Board of Directors who had not served as an executive officer of the Company during the past year. These options vest monthly over one year.

On March 5, 2002 and April 11, 2002, the Compensation Committee of the Board of Directors approved grants of stock options to purchase a total of 2,388,000 shares and 80,500 shares, respectively, of the Company's common stock at an exercise price of \$0.90 per share and \$0.96 per share, the respective closing prices at the dates of the grants. These grants were made to employees (other than the executive officers) and nonemployee outside sales personnel. These options vest in equal quarterly installments over five years. Included in the 2,388,000 options granted on March 5, 2002 were 135,000 options to nonemployees.

(6) Income Taxes

As of April 30, 2001, the Company had assessed the realizability of its deferred tax assets relating to the capital loss carry forward and unrealized net loss generated by the Company's trading securities during

8

the six months ended April 30, 2001. As a result, the Company determined that it was more likely than not that these deferred tax assets totaling approximately \$3,148,000 as of April 30, 2001, will not be realized. Accordingly, the Company established a valuation allowance for deferred tax assets in the amount of approximately \$3,148,000 as of April 30, 2001, which is included in income tax expense for the six months ended April 30, 2001.

(7) Net Income (Loss) Per Share

Basic net income (loss) per share excludes dilution and is computed by dividing income (loss) available to common stockholders by the weighted-average number of common shares outstanding for the period. Diluted net income (loss) per share reflects the potential dilution that could occur if securities or other contracts to issue common stock were exercised or converted into common stock or resulted in the issuance of common stock that then shared in the net income (loss) of the Company. The following is a reconciliation of the numerators and denominators of the net income (loss) per share computations for the periods presented:

Three Months Ended April 30, 2002		Net Income (Numerator)	Shares (Denominator)		Per Share Amount
Basic net income per share	\$	108,177	55,431,279	\$	
Effect of dilutive stock options			26,484	==	
Diluted net income per share	\$	108,177	55,457,763		
Three Months Ended April 30, 2001	_	Net Loss (Numerator)	Shares (Denominator)		Per Share
Basic net loss per share	Ş		56,277,995	\$	
Effect of dilutive stock options			203,222	==	
Diluted net loss per share	\$	(3,657,928)	56,481,217		· · ·
Six Months Ended April 30, 2002	_	Net Income	Shares (Denominator)		Per Share Amount
Basic net income per share	Ş		55,431,279	\$	
Basic net income per share Effect of dilutive stock options	Ş	712,629	55,431,279 463,147	\$	0.01
-	- \$	712,629 712,629	55,431,279 463,147 55,894,426	\$ == \$	0.01
Effect of dilutive stock options Diluted net income per share Six Months Ended April 30, 2001	- \$	712,629 	55,431,279 463,147 55,894,426 Shares (Denominator)	\$ == ==	0.01 Per Share
Effect of dilutive stock options Diluted net income per share	- \$	712,629 712,629 	55,431,279 463,147 55,894,426 Shares	\$ == \$ == \$	0.01 Per Share Amount (0.07)
Effect of dilutive stock options Diluted net income per share Six Months Ended April 30, 2001	; ;	712,629 712,629 Net Loss (Numerator) (3,772,541) 	55,431,279 463,147 55,894,426 Shares (Denominator) 56,317,635 332,458	\$ == \$ == \$ ==	0.01 Per Share Amount
Effect of dilutive stock options Diluted net income per share Six Months Ended April 30, 2001 Basic net loss per share	; ;	712,629 712,629 Net Loss (Numerator) (3,772,541) 	55,431,279 463,147 55,894,426 Shares (Denominator) 56,317,635 332,458	\$ == \$ == \$ == \$	0.01 Per Share Amount (0.07) (0.07)

9

(8) Shareholders' Equity

On November 2, 2001, the Board of Directors of the Company adopted a new Shareholder Rights Plan (the "Rights Plan") and declared a dividend of one preferred share purchase right (a "Right") on each outstanding share of common stock. Under the terms of the Rights Plan, if a person or group acquires 15% (or other applicable percentage, as provided in the Rights Plan) or more of the outstanding common stock, each Right will entitle its holder (other than such person or members of such group) to purchase, at the Right's then current exercise price, a number of shares of common stock having a market value of twice such price. In addition, if the Company is acquired in a merger or other business transaction after a person or group has acquired such percentage of the outstanding common stock, each Right will entitle its holder (other than such person or members of such group) to purchase, at the Right's then current exercise price, a number of the acquiring company's common shares having a market value of twice such right's then current exercise price, a number of such group) to purchase, at the Right's then current exercise price, a number of such group) to purchase, at the Right's then current exercise price, a number of such group) to purchase, at the Right's then current exercise price, a number of the acquiring company's common shares having a market value of twice such price.

Upon the occurrence of certain events, each Right will entitle its holder to buy one one-thousandth of a Series A preferred share ("Preferred Share"), at an exercise price of \$25, subject to adjustment. Each Preferred Share will entitle its holder to 1,000 votes and will have an aggregate dividend rate of 1,000 times the amount, if any, paid to holders of common stock. The Rights will expire on November 2, 2011, unless the date is extended or unless the Rights are earlier redeemed or exchanged at the option of the Board of Directors for \$0.0001 per Right. Generally, each share of common stock issued after November 5, 2001 will have one Right attached. The adoption of the Rights Plan has no impact on the financial position or results of operations of the Company.

The Company has reserved 100,000 of its authorized preferred stock for issuance upon exercise of the Rights.

(9) Personnel Matters

On December 3, 2001, the Company issued a press release that announced that, upon the recommendation of the independent Special Committee of its Board of Directors, the Board of Directors had removed Robert Kopstein as the Company's Chairman, President and Chief Executive Officer. Also see note 11.

(10) Segment Information and Business and Credit Concentrations

The Company has a single reportable segment for purposes of segment reporting pursuant to SFAS No. 131. In addition, the Company's fiber optic cable products are similar in nature.

10

The Company provides credit, in the normal course of business, to various commercial enterprises, governmental entities and not-for-profit organizations. Concentration of credit risk with respect to trade receivables is limited due to the Company's large number of customers. The Company also manages exposure to credit risk through credit approvals, credit limits and monitoring procedures. Management believes that credit risks as of April 30, 2002 and October 31, 2001 have been adequately provided for in the financial statements.

For the three months ended April 30, 2002, one customer accounted for approximately \$1,096,000, or 10.0% of net sales. For the three months ended April 30, 2001, one major domestic distributor accounted for approximately \$1,822,000, or 10.5% of net sales. During first quarter 2002, this distributor advised the Company that it will no longer stock the Company's products as part of its regular product offering. For the six months ended April 30, 2002 and 2001, no single customer accounted for more than 10% of net sales.

For the six months ended April 30, 2002 and 2001, approximately 73% and 77%, respectively, of net sales were from customers located in the United States, while approximately 27% and 23%, respectively, were from international customers.

(11) Contingencies

On September 27, 2000, the Equal Employment Opportunity Commission ("EEOC") filed a lawsuit under Title VII of the Civil Rights Act against the Company in the United States District Court for the Western District of Virginia. The lawsuit alleged a pattern or practice of discrimination on the bases of gender and race. The lawsuit sought injunctive and other relief and damages in an unspecified amount. On December 13, 2001, the parties reached an agreement as to the amount of a settlement (subject to final documentation and judicial review and approval). On February 20, 2002, the Company reached a final settlement of the case and the court issued a Consent Decree setting forth the terms of the settlement. Pursuant to the settlement and Consent Decree, the Company paid \$500,000 on or about February 20, 2002; and will pay \$175,000 in January 2003 and \$175,000 in January 2004, to satisfy any gender and race class claims; the Company paid \$75,000 on or about February 20, 2002 to one individual specifically named in the complaint; and will spend at least \$75,000 for the Company's planned diversity, recruitment and human resource management programs over the term of the Consent Decree. The Company recorded a charge in the fourth quarter of fiscal year 2001 in the amount of \$901,553 representing \$575,000 (current portion) payable upon entry of the Consent Decree, as well as \$326,553 (noncurrent portion) representing the present value of two equal payments in the amount of \$175,000 payable in January 2003 and 2004. During the six months ended April 30, 2002, the Company recorded accretion of the associated discount in the amount of \$6,621. The \$75,000 used for the Company's planned diversity, recruitment and human resource management programs will be expensed as incurred.

The Company was named as a defendant in two lawsuits filed in the United States District Court for the Southern District of New York seeking to compel the Company to authorize its transfer agent to transfer unregistered, restricted stock on the Company's stock ledger. The first suit was filed on October 22, 2001, by Bear, Stearns & Co. Inc. and Bear, Stearns Securities Corporation (collectively, "Bear Stearns"). The second suit was filed on October 26, 2001, by UBS PaineWebber Inc. ("PaineWebber"). In each case, PaineWebber and Bear Stearns sought injunctive relief with respect to common stock of the Company sold by them in the course of liquidating either repossessed shares or brokerage accounts of the Company's former Chairman, President and Chief Executive Officer, to cover personal margin loans made by the brokerage firms to him. Both suits also contained a claim of

11

monetary damages caused by the alleged wrongful refusal by the Company to authorize the transfers in connection with the liquidations. During the quarter ended April 30, 2002, both suits were dismissed. PaineWebber voluntarily dismissed its lawsuit with prejudice, while the Bear Stearns lawsuit was voluntarily dismissed without prejudice.

The Company, Mr. Robert Kopstein, our former Chairman, President and Chief Executive Officer, two of the Company's officers and directors, Luke J. Huybrechts and Kenneth W. Harber, are named as defendants in a consolidated putative class action lawsuit pending in the United States District Court for the Western District of Virginia (the "Consolidated Suit"). The first class action lawsuit was filed on November 26, 2001, by Charles S. Farrell, Jr., on behalf of himself and others similarly situated. The second class action lawsuit was filed on December 14, 2001, by Lerner Group, on behalf of itself and others similarly situated. The third class action lawsuit was filed on December 27, 2001, by Richard Simone, on behalf of himself and others similarly situated. The fourth class action lawsuit was filed on January 31, 2002, by Charles H. Yeatts, on behalf of himself and others similarly situated. In each of the four suits, the defendants in the actions were the Company, Mr. Kopstein and various John Does (unidentified officers and/or directors of the Company during the class period described below). Pursuant to the Private Securities Litigation Reform Act of 1995, the United States District Court for the Western District of Virginia appointed a group of shareholders as the lead plaintiff for the Consolidated Suit. The four suits were consolidated into a single action and the lead plaintiff filed a consolidated amended class action complaint on May 2, 2002. In the Consolidated Suit, the plaintiffs purport to represent purchasers of the Company's common stock during the period ranging from June 14, $% \left({{{\rm{Company}}} \right)$ 2000, through September 26, 2001, (the putative class period), and allege that the defendants violated Sections 10(b) and 20 of the federal Securities Exchange Act of 1934 in making certain alleged misrepresentations and/or omitting to disclose material facts. The plaintiffs in the Consolidated Suit seek compensatory damages in an unspecified amount, as well as reasonable costs and expenses incurred in the cause of action, including attorneys' fees and expert fees. On June 10, 2002, the Defendants moved to dismiss the Consolidated Suit.

Management intends to vigorously defend the Consolidated Suit. The Company may, however, incur substantial costs in defending the Consolidated Suit, regardless of its merit or outcome. At this early stage in the Consolidated Suit, management cannot make a reasonable estimate of the monetary amount of its resolution, or estimate a range of reasonably possible losses, if any. If the Company is unsuccessful, it could be subject to damages that may be substantial and could have a material adverse effect on the Company's financial position, results of operations and liquidity.

From time to time, the Company is involved in various other claims and legal actions arising in the ordinary course of business. In the opinion of management, the ultimate disposition of these matters will not have a material adverse effect on the Company's financial position, results of operations or liquidity.

(12) Subsequent Events

On June 14, 2002, the Company announced the layoff of 28 employees, representing almost 15 percent of its full-time employees. The reductions included both production and staff positions. The layoffs were part of a company-wide downsizing and will result in a charge of approximately \$120,000 during the third fiscal quarter of 2002.

Subsequent to April 30, 2002, Optical Cable Corporation received notification from Nasdaq regarding the Company's noncompliance with the listing requirements for the Nasdaq National Market, where the Company's common stock is presently traded. As a Nasdaq National Market company, Optical Cable is subject to all the requirements of its listing agreement with Nasdaq. The Company received notification from Nasdaq that its common stock failed to meet the minimum \$1.00 per share trading price requirement for continued inclusion under Marketplace Rule 4450(a)(5). The Company will be provided until August 13, 2002, to regain compliance or be delisted. In order to regain compliance the Company's common stock must close at \$1.00 per share or more for a minimum of 10 consecutive trading days prior to August 13. In an effort to regain compliance with the Nasdaq National Market's minimum trading price requirement, the Company is considering several options, including a reverse stock split. Under Virginia law, such an action would require an amendment to the Company's articles of incorporation that must be approved by both the Board of Directors and the shareholders. In the event a reverse stock split were effected, the number of shares held by each shareholder and the total number of outstanding shares would be reduced proportionately.

Item 2. Management's Discussion and Analysis Results of Operations and Financial Condition $% \left({{{\left({{{{\rm{A}}}} \right)}_{\rm{cl}}}} \right)$

Forward Looking Information

This Form 10-Q may contain certain forward-looking information within the meaning of the federal securities laws. The forward-looking information may include, among other information, statements concerning our outlook for the future; statements of belief; future plans, strategies or anticipated events; and similar information and statements concerning matters that are not historical facts. Such forward-looking information is subject to risks and uncertainties that may cause actual events to differ materially from our expectations. Factors that could cause or contribute to such differences include, but are not limited to, the level of sales to key customers or distributors; the economic conditions affecting network service providers; the slowdown in corporate spending on information technology; actions by competitors; fluctuations in the price of raw materials (including optical fiber); our dependence on a single manufacturing facility; our ability to protect our proprietary manufacturing technology; market conditions influencing prices or pricing; our dependence on a limited number of suppliers; an adverse outcome in litigation, claims and other actions against us, including, but not limited to, the shareholder litigation that has been filed; the effect of sales of our common stock by the various brokerage firms alleging that our former President and Chief Executive Officer pledged substantially all of his personally-held unregistered shares of our common stock to cover personal margin loans; technological changes and introductions of new competing products; the current recession; terrorist attacks or acts of war, particularly given the acts of terrorism against the United States on September 11, 2001 and subsequent military responses by the United States; ability to retain key personnel; changes in market demand, exchange rates, productivity, weather and market and economic conditions in the areas of the world in which we operate and market our products.

Amounts presented in the following discussion have been rounded to the nearest hundred thousand, unless the amounts are less than one million, in which case the amounts have been rounded to the nearest thousand.

Overview

We are a leading manufacturer of a broad range of tight-buffered fiber optic cables primarily for the local area network and premise markets, often referred to as the enterprise market. Our tight-buffered fiber optic cables are well-suited for use in short to moderate distance applications to connect metropolitan, access and enterprise networks. Our tight-buffered fiber optic cables are derived from technology originally developed for military applications requiring rugged, flexible and compact fiber optic cables. Our tight-buffered fiber optic cables can be used both indoors and outdoors, are easy and economical to install, provide a high degree of reliability and offer industry leading performance characteristics. We have designed and implemented an efficient and highly automated manufacturing process based on proprietary technologies. This enables us to produce high quality indoor/outdoor tight-buffered fiber optic cable rapidly and cost efficiently.

We sell our products through our sales force to original equipment manufacturers, and to major distributors, regional distributors and various smaller distributors. For the three months ended April 30, 2002 and 2001, approximately 48.4% and 41.5% of our net sales were from sales to distributors. For the six months ended April 30, 2002 and 2001, approximately 44.2% and 43.3% of our net sales were from sales to distributors. International net sales were 20.5% and 24.2% of total net sales for the three months ended April 30, 2002 and 2001. International net sales were 27.5% and 23.5% of total net sales for the six months ended April 30, 2002 and 2001. Substantially all of our international sales are denominated in U.S. dollars.

Net sales consist of gross sales of products less discounts, refunds and returns. Revenue is recognized at the time of product shipment or delivery to the customer and at the time the customer takes ownership and assumes risk of loss, based on shipping terms. During the second quarter of 2002, one customer accounted

13

for approximately \$1.1 million, or 10% of net sales. During the second quarter of 2001, one major domestic distributor accounted for approximately \$1.8 million, or 10.5% of net sales. During first quarter 2002, this distributor advised us that it will no longer stock our products as part of its regular product offering. No single customer accounted for more than 10% of our net sales during the six months ended April 30, 2002 or 2001.

A significant percentage of the selling price of our fiber optic cable is based on the cost of raw materials used. Because single-mode fiber is less expensive than multimode fiber, single-mode fiber optic cables have a lower per unit selling price than comparable multimode fiber optic cables. We believe that the metropolitan and access markets are predominantly the users of single-mode fiber optic cable, and that increasingly, single-mode fiber is also being used for other short to moderate distance installations where higher bandwidth is required. To the extent that our sales mix shifts toward single-mode cables, we will have to increase the volume of our sales to maintain our current level of net sales.

Cost of goods sold consists of the cost of materials, compensation costs, product warranty costs and overhead related to our manufacturing operations. The largest percentage of costs included in cost of goods sold is attributable to costs of materials that are variable as opposed to fixed costs.

Selling, general and administrative expenses consist of the compensation costs for sales and marketing personnel, shipping costs, travel expenses, customer support expenses, trade show expenses, advertising, bad debt expense, the compensation cost for administration, finance and general management personnel, as well as legal and accounting fees.

Other expense, net consists primarily of realized and unrealized net losses on trading securities, interest income and interest expense. In January 2000, we began actively buying and selling shares in the Nasdaq 100 Trust, which is designed to closely track the price and yield performance of the Nasdaq 100 stock index. We utilized short-term margin borrowings payable to an investment broker to finance our position in these trading securities. Our margin borrowings were collateralized by the trading securities and were subject to margin provisions, which could have resulted in the sale of some or all of, and on certain occasions did result in the sale of some of, the trading securities to meet margin calls. Our active trading in the Nasdaq 100 Trust continued through May 14, 2001, the date of the last purchase of these shares. On October 3, 2001, as part of a policy to invest future excess funds only in short-term interest-bearing investments, we sold all of our remaining investment in the Nasdaq 100 Trust and paid off the outstanding margin borrowings. As a result, there were no gains or losses resulting from the trading of securities for the six months ended April 30, 2002. Our Board of Directors has adopted an Investment Objectives and Guidelines policy, in which we state that we will make no additional cash investments in the above-mentioned Nasdaq 100 Trust or in stocks of other companies. In addition, our Investment Objectives and Guidelines policy states that any future investments will be in U.S. dollar denominated short-term, interest-bearing, investment-grade securities.

For accounting purposes during first quarter 2001, we categorized our investment in the Nasdaq 100 Trust as trading securities, and we recorded the investment on our balance sheet at fair value, which was based on quoted market prices. Purchases and sales of trading securities were recognized on a trade-date basis, the date the order to buy or sell is executed. Net realized gains or losses were determined on the first-in, first-out cost method. We marked our investment to market on each balance sheet date. Any decline in fair value was recorded as an unrealized loss, while any increase in fair value was recorded as an unrealized gain. Realized gains and losses and unrealized holding gains and losses from trading securities were included in other expense, net.

14

In the second quarter of 2001, we recognized net losses of \$5.2 million in connection with our securities trading activities in other expense, net. Likewise, during the six months ended April 30, 2001, we recognized net losses of \$9.2 million in connection with our securities trading activities in other expense, net. In the second quarter of 2001 and for the six months ended April 30, 2001, we incurred interest expense of \$98,000 and \$248,000, respectively, on the margin borrowings related to our securities trading activities. Since October 2001, we have held no trading securities in accordance with our current investment policy and had no outstanding margin borrowings.

Results of Operations

The following table sets forth selected line items from our condensed statements of operations as a percentage of net sales for the periods indicated:

	Three Months April 30		Six Months Ended April 30,		
	2002	2001	2002	2001	
Net sales Cost of goods sold	100.0 % 67.0	100.0 % 54.9	100.0 % 63.7	100.0 % 54.3	
Gross profit Selling, general and administrative expenses	33.0 31.1	45.1 21.6	36.3 30.9	45.7 22.3	
Income from operations Other expense, net	1.9 (0.4)	23.5 (30.1)	5.4 (0.5)	23.4 (27.2)	
Income (loss) before income tax expense Income tax expense	1.5 0.5	(6.6) 14.4	4.9 1.7	(3.8) 7.1	
Net income (loss)	1.0 %	(21.0)%	3.2 %	(10.9) %	

Three Months Ended April 30, 2002 and 2001

Net Sales

Net sales decreased 37.2% to \$10.9 million for the three months ended April 30, 2002 from \$17.4 million for the three months ended April 30, 2001. The decrease in net sales during the second quarter of 2002, when compared to the same period last year, was a result of weak economic conditions on market demand, a trend that significantly affected our net sales beginning in the second half of fiscal year 2001. Total fiber meters shipped during the second quarter of 2002 decreased 39.7% to 36.5 million fiber meters shipped from 60.5 million fiber meters shipped for the same period in 2001.

During the second quarter of 2002, we experienced an increase in product mix for cable containing multimode fiber (which typically has a higher relative sales price), compared to cable containing single-mode fiber (which typically has a lower relative sales price), when compared to the second quarter of 2001. The percentage of multimode fiber meters shipped to total fiber meters shipped during the second quarter of 2002 was 76.1% compared to 58.1% during the second quarter of 2001; however, the impact of the decreased market demand and pricing outweighed the impact on sales resulting from the change in product mix.

Cable containing multimode fiber is generally used for communications over shorter distances where the higher bandwidth capacity and the higher transmission equipment cost of single-mode fiber is not required. Multimode fiber cable is often used in datacom applications. Cable containing single-mode fiber is generally used for communications over longer distances and where higher bandwidth capacity is required. Single-mode fiber cable is often used in telecom, CATV and various Internet applications.

15

Gross Profit

Gross profit margin, or gross profit as a percentage of net sales, decreased to 33.0% in the second quarter of 2002 from 45.1% in the second quarter of 2001, as production costs did not decrease at the same relative rate as the decrease in net sales. By comparison, the gross profit margin for fourth quarter 2001 was 21.8%, a quarter in which we wrote-off slow-moving and impaired inventory to net realizable value, disposed of certain impaired finished goods inventory, and adjusted inventory for book to physical variances resulting from year-end physical inventory counts. Although raw material cost per fiber meter shipped decreased slightly during the second quarter of 2002 compared to the second quarter of 2001, an increase in production cost resulted from increases in labor and overhead cost per fiber meter shipped.

Selling, General and Administrative Expenses

Selling, general and administrative expenses as a percentage of net sales were 31.1% in the second quarter of 2002, compared to 21.6% in the second quarter of 2001. By comparison, selling, general and administrative expenses were 47.2% of net sales in the fourth quarter of 2001, mainly due to the \$902,000 charge recorded in anticipation of settlement of the lawsuit with the Equal Employment Opportunity Commission for alleged prior discriminatory practices and related complaints. The higher percentage in the second quarter of 2002 reflects the fact that net sales for the quarter decreased 37.2% compared to the same period last year, while selling, general and administrative expenses only decreased 9.4% compared to the second quarter of 2001. The decrease in selling, general and administrative expenses during the second quarter of 2002 compared to the same period last year is explained by a decrease in sales commissions and shipping costs, and a decrease of \$201,000 in bad debt expense, that were partially offset by an increase in legal and other professional fees.

Other Income (Expense)

Other expense, net decreased to \$48,000 in the second quarter of 2002 from \$5.2 million in the second quarter of 2001. The decrease was primarily due to cessation of securities trading activity, the liquidation of our remaining investments in the Nasdaq 100 Trust and the pay off of the outstanding margin borrowings prior to the end of fiscal year 2001. In the second quarter of 2001, we recognized net losses of \$5.2 million in connection with our securities trading activities, and incurred interest expense of \$98,000 on the margin borrowings related to such activities. Since October 2001, we have held no trading securities in accordance with our current investment policy and had no outstanding margin borrowings. Please see our discussion of trading securities in "Overview" above.

Income (Loss) Before Income Tax Expense

Income (loss) before income tax expense increased to pretax income of \$167,000 in the second quarter of 2002 from a pretax loss of \$1.2 million in the second quarter of 2001. This increase was primarily due to the \$5.2 million decrease in losses on trading securities, net and the \$353,000 decrease in selling, general and administrative expenses, partially offset by the \$4.2 million decrease in gross profit largely resulting from the decrease in net sales.

Income Tax Expense

Income tax expense decreased to \$59,000 in the second quarter of 2002 from \$2.5 million in the second quarter of 2001. Although we incurred a pretax loss for the three months ended April 30, 2001, our reported income tax expense differed from the expected income tax benefit, computed based on an expected effective tax rate of 35.0%, due primarily to the establishment of a valuation allowance for deferred tax assets in the

16

amount of \$3.1 million. As of April 30, 2001, we assessed the realizability of our deferred tax assets relating to the capital loss carryforward and unrealized net losses from our trading securities and determined that it was more likely than not that these deferred tax assets will not be realized. For the three months ended April 30, 2002, our effective tax rate was 35.1%. Fluctuations in our effective tax rates typically are due primarily to the amount and timing of the tax benefits related to our Extraterritorial Income Exclusion and our foreign sales corporation, both of which exempt from federal income taxation a portion of the net profit realized from sales outside of the United States of products manufactured in the United States.

Net Income (Loss)

Net income for the second quarter of 2002 was \$108,000, compared to a net loss of \$3.7 million for the second quarter of 2001. This \$3.8 million increase in net income was primarily due to the \$5.2 million decrease in losses on trading securities, net, the \$353,000 decrease in selling, general and administrative expenses, and the \$2.4 million decrease in income tax expense, partially offset by the \$4.2 million decrease in gross profit largely resulting from the decrease in net sales.

Six Months Ended April 30, 2002 and 2001

Net Sales

Net sales decreased 35.1% to \$22.3 million for the six months ended April 30, 2002 from \$34.4 million for the six months ended April 30, 2001. The decrease in net sales during the first six months of 2002, when compared to the same period last year, was a result of weak economic conditions on market demand, discussed above. Total fiber meters shipped during the first six months of 2002 decreased 37.8% to 74.2 million fiber meters shipped from 119.2 million fiber meters shipped for the same period in 2001.

During the first six months of 2002, we experienced an increase in product mix for cable containing multimode fiber (which typically has a higher relative sales price), compared to cable containing single-mode fiber (which typically has a lower relative sales price), when compared to the same period in 2001. The percentage of multimode fiber meters shipped to total fiber meters shipped during the first six months of 2002 was 69.7% compared to 59.9% during the same period in 2001; however, the impact of the decreased market demand and pricing outweighed the impact on sales resulting from the change in product mix.

Gross Profit

Gross profit margin decreased to 36.3% in the first six months of 2002 from 45.7% for the same period in 2001, as production costs did not decrease at the same relative rate as the decrease in net sales. Raw material cost per fiber meter shipped was relatively constant when comparing the first six months of 2002 to the same period in 2001. Therefore, most of the production cost increase was a result of increases in labor and overhead cost per fiber meter shipped.

Selling, General and Administrative Expenses

Selling, general and administrative expenses as a percentage of net sales were 30.9% in the first six months of 2002, compared to 22.3% for the same period in 2001. The higher percentage in the first six months of 2002 reflects the fact that net sales for the period decreased 35.1% compared to the same period last year, while selling, general and administrative expenses only decreased 10.1% compared to the same period in 2001. The decrease in selling, general and administrative expenses of 2002 compared to the same period six year is explained by a decrease in sales commissions and shipping costs,

17

and a decrease of \$361,000 in bad debt expense, that were partially offset by an increase in legal and other professional fees.

Other Income (Expense)

Other expense, net decreased to \$105,000 for the first six months of 2002 from \$9.4 million for the same period in 2001. The decrease was primarily due to cessation of securities trading activity, the liquidation of our remaining investments in the Nasdaq 100 Trust and the pay off of the outstanding margin borrowings prior to the end of fiscal year 2001. In the first six months of 2001, we recognized net losses of \$9.2 million in connection with our securities trading activities, and incurred interest expense of \$248,000 on the margin borrowings related to such activities,. Since October 2001, we have held no trading securities in accordance with our current investment policy and had no outstanding margin borrowings. Please see our discussion of trading securities in "Overview" above.

Income (Loss) Before Income Tax Expense

Income (loss) before income tax expense increased to pretax income of \$1.1 million in the first six months of 2002 from a pretax loss of \$1.3 million for the same period in 2001. This increase was primarily due to the \$9.2 million decrease in losses on trading securities, net and the \$774,000 decrease in selling, general and administrative expenses, partially offset by the \$7.6 million decrease in gross profit largely resulting from the decrease in net sales.

Income Tax Expense

Income tax expense decreased to \$387,000 in the first six months of 2002 from \$2.4 million for the same period in 2001. Although we incurred a pretax loss for the six months ended April 30, 2001, our reported income tax expense differed from the expected income tax benefit, computed based on an expected effective tax rate of 35.0%, due primarily to the establishment of a valuation allowance for deferred tax assets discussed above. For the six months ended April 30, 2002, our effective tax rate was 35.2%.

Net Income (Loss)

Net income for the first six months of 2002 was \$713,000, compared to a net loss of \$3.8 million for the same period in 2001. This \$4.5 million increase in net income was primarily due to the \$9.2 million decrease in losses on trading securities, net, the \$774,000 decrease in selling, general and administrative expenses, and the \$2.1 million decrease in income tax expense, partially offset by the \$7.6 million decrease in gross profit largely resulting from the decrease in net sales.

Financial Condition

Total assets decreased \$7.5 million, or 17.5%, to \$35.3 million at April 30, 2002, from \$42.8 million at October 31, 2001. This decrease was primarily due to a \$2.7 million decrease in trade accounts receivable, net, resulting from the decreased sales volume during the first six months of 2002 as compared to the last six months of 2001, a \$2.2 million decrease in inventories, and a \$2.0 million decrease in cash and cash equivalents.

Total liabilities decreased \$8.2 million, or 54.9%, to \$6.7 million at April 30, 2002, from \$14.9 million at October 31, 2001. This decrease was primarily due to a \$8.3 million decrease in notes payable to our bank under our previous lines of credit, and a \$2.7 million decrease in accounts payable and accrued expenses and

18

accrued compensation and payroll taxes, partially offset by an increase in long-term debt of \$3.0 million under our new revolving line of credit with a three year term.

Total stockholders' equity at April 30, 2002 increased \$718,000, or 2.3% during the first two quarters of 2002. The increase resulted primarily from net income retained for the six months ended April 30, 2002.

Liquidity and Capital Resources

Our primary capital needs have been to fund working capital requirements and capital expenditures. Our primary source of capital for these purposes has been cash provided from operations and borrowings under our bank lines of credit described below. The outstanding balance under our new line of credit totaled \$3.0 million as of April 30, 2002 and has been reflected as long-term debt based on the scheduled maturity of the credit facility. The outstanding balance of \$3.0 million as of April 30, 2002 reflects a decrease of \$5.3 million from the \$8.3 million balance outstanding under our previous lines of credit at October 31, 2001.

Our cash and cash equivalents totaled \$133,000 as of April 30, 2002, a decrease of \$2.0 million, compared to \$2.1 million as of October 31, 2001. The cash and cash equivalents decrease for the six months ended April 30, 2002, was primarily due to the repayment of notes payable to bank, net, totaling \$5.3 million, partially offset by net cash provided by operating activities of \$3.4 million.

On April 30, 2002, we had working capital of \$18.4 million, compared to \$14.2 million as of October 31, 2001, an increase of \$4.2 million. The ratio of current assets to current liabilities as of April 30, 2002, was 6.1 to 1, compared to 2.0 to 1 as of October 31, 2001. The increase in working capital during the six months ended April 30, 2002 was primarily caused by a decrease in notes payable to bank of \$8.3 million, reflected entirely as a current liability at October 31, 2001, and a \$2.7 million decrease in accounts payable and accrued expenses and accrued compensation and payroll taxes, partially offset by a \$2.0 million decrease in trade accounts receivable, net and a \$2.2 million decrease in inventories.

Net cash provided by operating activities was approximately \$3.4 million for the six months ended April 30, 2002, compared to \$2.4 million for the six months ended April 30, 2001. Net cash provided by operating activities during the first six months of 2002 primarily resulted from cash provided by operating income, a \$2.6 million decrease in trade accounts receivable and a \$2.2 million decrease in inventories, partially offset by a \$2.9 million decrease in accounts payable and accrued expenses (including accrued compensation and payroll taxes). Net cash provided by operating activities for the six months ended April 30, 2001 primarily resulted from cash provided by operating income, a decrease in trading securities of \$13.1 million, a decrease in income taxes refundable of \$1.2 million and an increase in accounts payable and accrued expenses of \$878,000. partially offset by an increase in trade accounts receivable of \$5.1 million, an increase in inventories of \$1.2 million and a decrease in payable to investment broker of \$4.6 million. We have entered into written agreements to purchase raw optical fiber. These commitments total \$7.6 million, \$10.7 million, \$6.9 million and \$1.0 million in fiscal years 2002, 2003, 2004 and 2005, respectively.

Net cash provided by investing activities totaled \$215,000 for the first six months in 2002, compared to net cash used in investing activities of \$1.2 million for the first six months of 2001. Net cash generated in investing activities during the first six months of 2002 primarily resulted from \$367,000 provided by the receipt of the cash surrender value of a life insurance policy on a former officer, partially offset by \$153,000 in purchases of property and equipment. Net cash used in investing activities during the first six months of 2001 primarily resulted from expenditures related to property and equipment and a \$54,000 increase in the cash surrender value of the life insurance policy of a former officer.

19

Net cash used in financing activities was \$5.6 million for the six months ended April 30, 2002, compared to \$1.3 million for the six months ended April 30, 2001. Net cash used in financing activities for the six months ended April 30, 2002 was the result of repayments on notes payable to our bank under our lines of credit and payments for financing costs related to our new credit facility. Net cash used in financing activities for the six months ended April 30, 2001 was primarily the result of repurchases of common stock in the amount of \$1.6 million.

On April 25, 2002, we closed our new credit facility with Wachovia Bank, National Association (formerly First Union National Bank). The new three-year credit facility provides up to a maximum of \$25.0 million and replaces our previous \$9.5 million credit facility.

The new credit facility bears interest at three-quarters of one percent (0.75%) per annum above the prime rate (5.5% as of April 30, 2002) and may be reduced by one-quarter of one percent (0.25%) upon meeting certain fixed charge coverage ratio requirements. The facility also provides a LIBOR based rate at our option. The facility is collateralized by all of our tangible and intangible assets. Borrowings under the credit facility are subject to certain coverage ratios, advance limits and qualifications that are applied to our accounts receivable, inventory and fixed assets. Our ability to access the full amount of the credit facility will depend on the future growth of our borrowing base. As of April 30, 2002, we had outstanding borrowings under the new credit facility in the amount of approximately \$3.0 million, with approximately \$5.1 million unused and available. The outstanding balance on the credit facility has been reflected as noncurrent based on the scheduled maturity of the credit facility. We believe that our cash flow from operations and our credit facility will be adequate to fund our operations for at least the next twelve months.

On September 27, 2000, the Equal Employment Opportunity Commission ("EEOC") filed a lawsuit under Title VII of the Civil Rights Act against us in the United States District Court for the Western District of Virginia. The lawsuit alleged a pattern or practice of discrimination on the bases of gender and race. The lawsuit sought injunctive and other relief and damages in an unspecified amount. On December 13, 2001, the parties reached an agreement as to the amount of a settlement (subject to final documentation and judicial review and approval). On February 20, 2002, we reached a final settlement of the case with the EEOC and the court issued a Consent Decree setting forth the terms of the settlement. Pursuant to the settlement and Consent Decree, we paid \$500,000 on or about February 20, 2002; and will pay \$175,000 in January 2003 and \$175,000 in January 2004, to satisfy any gender and race class claims; we paid \$75,000 on or about February 20, 2002 to one individual specifically named in the complaint; and will spend at least \$75,000 for our planned diversity, recruitment and human resource management programs over the term of the Consent Decree. We recorded a charge in the fourth quarter of fiscal year 2001 in the amount of \$902,000 representing \$575,000 payable upon entry of the Consent Decree, as well as \$327,000 representing the present value of two equal payments in the amount of \$175,000 payable in January 2003 and 2004. As of April 30, 2002, we have paid \$575,000 and accreted \$7,000 of the discount associated with recording the liability at its present value. Of the remaining liability totaling \$333,000 as of April 30, 2002, \$170,000 has been reflected as current. The \$75,000 to be used for our planned diversity, recruitment and human resource management programs will be expensed as incurred.

We were named as a defendant in two lawsuits filed in the United States District Court for the Southern District of New York seeking to compel us to authorize our transfer agent to transfer unregistered, restricted stock on our stock ledger. The first suit was filed on October 22, 2001, by Bear, Stearns & Co. Inc. and Bear, Stearns Securities Corporation (collectively, "Bear Stearns"). The second suit was filed on October 26, 2001, by UBS PaineWebber Inc. ("PaineWebber"). In each case, PaineWebber and Bear Stearns sought injunctive relief with respect to our common stock sold by them in the course of liquidating either repossessed shares or brokerage accounts of Mr. Robert Kopstein, our former Chairman, President and Chief Executive Officer, to

20

cover personal margin loans made by the brokerage firms to Mr. Kopstein. Both suits also contained a claim of monetary damages caused by the alleged wrongful refusal by us to authorize the transfers in connection with the liquidations. During the quarter ended April 30, 2002, both suits were dismissed. PaineWebber voluntarily dismissed its lawsuit with prejudice, while the Bear Stearns lawsuit was voluntarily dismissed without prejudice.

The Company, Mr. Robert Kopstein, our former Chairman, President and Chief Executive Officer, two of the Company's officers and directors, Luke J. Huybrechts and Kenneth W. Harber, are named as defendants in a consolidated putative class action lawsuit pending in the United States District Court for the Western District of Virginia (the "Consolidated Suit"). The first class action lawsuit was filed on November 26, 2001, by Charles S. Farrell, Jr., on behalf of himself and others similarly situated. The second class action lawsuit was filed on December 14, 2001, by Lerner Group, on behalf of itself and others similarly situated. The third class action lawsuit was filed on December 27, 2001, by Richard Simone, on behalf of himself and others similarly situated. The fourth class action lawsuit was filed on January 31, 2002, by Charles H. Yeatts, on behalf of himself and others similarly situated. In each of the four suits, the defendants in the actions were the Company, Mr. Kopstein and various John Does (unidentified officers and/or directors of the Company during the class period described below). Pursuant to the Private Securities Litigation Reform Act of 1995, the United States District Court for the Western District of Virginia appointed a group of shareholders as the lead plaintiff for the Consolidated Suit. The four suits were consolidated into a single action and the lead plaintiff filed a consolidated amended class action complaint on May 2, 2002. In the Consolidated Suit, the plaintiffs purport to represent purchasers of the Company's common stock during the period ranging from June 14, 2000, through September 26, 2001, (the putative class period), and allege that the defendants violated Sections 10(b) and 20 of the federal Securities Exchange Act of 1934 in making certain alleged misrepresentations and/or omitting to disclose material facts. The plaintiffs in the Consolidated Suit seek compensatory damages in an unspecified amount, as well as reasonable costs and expenses incurred in the cause of action, including attorneys' fees and expert fees. On June 10, 2002, the Defendants moved to dismiss the Consolidated Suit.

Management intends to vigorously defend the Consolidated Suit. We may, however, incur substantial costs in defending ourselves against the Consolidated Suit, regardless of its merit or outcome. At this early stage in the Consolidated Suit, management cannot make a reasonable estimate of the monetary amount of its resolution, or estimate a range of reasonably possible losses, if any. If we are unsuccessful, we could be subject to damages that may be substantial and could have a material adverse effect on our financial position, results of operations and liquidity.

From time to time, we are involved in various other claims and legal actions arising in the ordinary course of business. In the opinion of management, the ultimate disposition of these matters will not have a material adverse effect on our financial position, results of operations or liquidity.

On June 14, 2002, we announced the layoff of 28 employees, representing almost 15% of our full-time employees. The reductions included both production and staff positions. The layoffs were part of a company-wide downsizing and will result in a charge of approximately \$120,000 during the third fiscal quarter of 2002.

21

The table below sets forth a summary of our contractual obligations and commitments that will impact on our future liquidity:

Contractual Obligations	Fiscal Years Ending October 31,						
and Commitments		2002	2003	2004	2005	Totals	
Bank lines of credit Long-term Optical Fiber	\$				2,970,000	2,970,000	
Supply Agreements EEOC Settlement		7,576,000	10,717,000 175,000	6,931,000 175,000	1,016,000	26,240,000 350,000	
Total	\$ =	7,576,000	10,892,000	7,106,000	3,986,000	29,560,000	

Bank Lines of Credit

On April 25, 2002, we closed our new credit facility with Wachovia Bank, National Association (formerly First Union National Bank). The new credit facility has a term of three-years and replaces our previous \$9.5 million credit facility. See further discussion under "Liquidity and Capital Resources."

Long-Term Optical Fiber Supply Agreements

During fiscal year 2001, we entered into separate long-term supply agreements with two optical fiber suppliers. One agreement expires on December 31, 2003 and the other on December 31, 2004. The aggregate required purchases related to these agreements (subject to certain annual price adjustments) during the terms of the agreements are set forth above. Additionally, one of the supply agreements requires that one-half of all single-mode fiber purchases through December 31, 2004 above the committed amounts be purchased from that supplier at market prices. We believe that the fiber purchase commitments are consistent with our expected requirements.

Equal Employment Opportunity Commission Settlement

On September 27, 2000, the Equal Employment Opportunity Commission ("EEOC") filed a lawsuit under Title VII of the Civil Rights Act against us in the United States District Court for the Western District of Virginia. The lawsuit alleged a pattern or practice of discrimination on the bases of gender and race. The lawsuit sought injunctive and other relief and damages in an unspecified amount. On December 13, 2001, the parties reached an agreement as to the amount of a settlement (subject to final documentation and judicial review and approval). On February 20, 2002, we reached a final settlement of the case with EEOC and the court issued a Consent Decree setting forth the terms of the settlement. Pursuant to the settlement and Consent Decree, we paid \$500,000 on or about February 20, 2002; and will pay \$175,000 in January 2003 and \$175,000 in January 2004, to satisfy any gender and race class claims; we paid \$75,000 on or about February 20, 2002 to one individual specifically named in the complaint; and will spend at least \$75,000 for our planned diversity, recruitment and human resource management programs over the term of the Consent Decree. We recorded a charge in the fourth quarter of fiscal year 2001 in the amount of \$902,000representing \$575,000 payable upon entry of the Consent Decree, as well as \$327,000 representing the present value of two equal payments in the amount of \$175,000 payable in January 2003 and 2004. As of April 30, 2002, we have paid \$575,000 and accreted \$7,000 of the discount associated with recording the liability at its present value. Of the remaining liability totaling \$333,000 as of April 30, 2002, \$170,000 has been reflected as current. The \$75,000 to be used for our planned diversity, recruitment and human resource management programs will be expensed as incurred, and is not reflected in the Contractual Obligations and Commitments table above.

22

Critical Accounting Policies

Revenue Recognition

Revenue is recognized at the time of product shipment or delivery to the customer, provided that the customer takes ownership and assumes risk of loss based on shipping terms. Net sales consists of gross sales of products, less discounts, refunds and returns. We estimate sales returns based on our analysis and judgment of historical trends, identified returns and the potential for additional returns. We also provide certain volume incentives, discounts and rebates to certain of our distributors. Payments of any such volume incentives, discounts and rebates are reflected in net sales.

Trade Accounts Receivable and Allowance for Doubtful Accounts

In connection with the sale of our products, we have trade accounts receivable outstanding from our customers at any given time. We review outstanding trade accounts receivable at the end of each quarter and record allowances for doubtful accounts as deemed appropriate for (i) certain individual customers and (ii) for all other trade accounts receivable in total not specifically reviewed. In determining the amount of allowance for doubtful accounts to be recorded for individual customers, we consider the age of the receivable, the financial stability of the customer, discussions that may have been had with the customer and our judgment as to the overall collectibility of the receivable from that customer. In addition, we establish an allowance for all receivables that have not been individually reviewed. This general allowance for doubtful accounts is based on a percentage of total trade accounts receivable with different percentages used based on the different age of the receivables. The percentages used are based on our historical experience and our current judgment regarding the state of the economy.

Inventories

Inventories of raw materials and production supplies are stated at the lower of cost (specific identification for optical fibers and first-in, first-out for other raw materials and production supplies) or market. Inventories of work in process and finished goods are stated at average cost, which includes raw materials, direct labor and manufacturing overhead. At the end of each quarter, we review our inventories to ensure they are carried at net realizable value. Individual inventory items are viewed and adjustments are made based on the age of the inventory and our judgment as to the salability of that inventory.

Long-Lived Assets

Our property and equipment are stated at cost. Depreciation and amortization are provided for using both straight-line and declining balance methods over the estimated useful lives of the assets. We review long-lived assets for impairment whenever events or changes in circumstances indicate that the carrying amount of an asset may not be recoverable. We have no significant intangible assets recorded on our balance sheet.

Commitments and Contingencies

Liabilities for loss contingencies arising from product warranties and defects, claims, assessments, litigation, fines and penalties and other sources are recorded when it is probable that a liability has been incurred and the amount of the assessment can be reasonably estimated. Actual results could differ from these estimates.

(Continued)

Future Accounting Considerations

In July 2001, the Financial Accounting Standards Board, also known as the FASB, issued Statement of Financial Accounting Standards, referred to as SFAS, No. 141, Business Combinations, and SFAS No. 142, Goodwill and Other Intangible Assets. SFAS No. 141 requires the use of the purchase method of accounting for all business combinations. The use of the pooling-of-interests method is prohibited for business combinations initiated after June 30, 2001. SFAS No. 142 requires that goodwill and certain intangible assets would no longer be amortized, but rather be tested for impairment annually or whenever an event occurs indicating that the assets may be impaired. SFAS No. 142 is effective for fiscal years beginning after December 15, 2001. Neither standard is expected to have a material effect on our financial position, results of operations or liquidity.

In August 2001, the FASB issued SFAS No. 143, Accounting for Asset Retirement Obligations, which addresses financial accounting and reporting for obligations associated with the retirement of tangible long-lived assets and the associated asset retirement costs. The standard applies to legal obligations associated with the retirement of long-lived assets that result from the acquisition, construction, development and (or) normal use of the asset.

SFAS No. 143 requires that the fair value of a liability for an assets retirement obligation be recognized in the period in which it is incurred if a reasonable estimate of fair value can be made. The fair value of the liability is added to the carrying amount of the associated asset and this additional carrying amount is depreciated over the life of the asset. The liability is accreted at the end of each period through charges to operating expense. If the obligation is settled for other than the carrying amount of the liability, an entity would recognize a gain or loss on settlement.

SFAS No. 143 is effective for fiscal years beginning after June 15, 2002. We are currently evaluating the impact of SFAS No. 143 on our financial position, results of operations and liquidity.

In October 2001, the FASB issued SFAS No. 144, Accounting for the Impairment or Disposal of Long-Lived Assets, which addresses financial accounting and reporting for the impairment or disposal of long-lived assets. SFAS No. 144 supersedes SFAS No. 121, Accounting for the Impairment of Long-Lived Assets and for Long-Lived Assets to Be Disposed Of; however, it retains many of the fundamental provisions of that Statement.

SFAS No. 144 also supersedes the accounting and reporting provisions of APB Opinion No. 30, Reporting the Results of Operations--Reporting the Effects of Disposal of a Segment of a Business, and Extraordinary, Unusual and Infrequently Occurring Events and Transactions, for the disposal of a segment of a business. However, it retains the requirement in APB No. 30 to report separately discontinued operations and extends that reporting to a component of an entity that either has been disposed of (by sale, abandonment, or in a distribution to owners) or is classified as held for sale. By broadening the presentation of discontinued operations to include more disposal transactions, the FASB has enhanced management's ability to provide information that helps financial statement users to assess the effects of a disposal transaction on the ongoing operations of an entity.

SFAS No. 144 is effective for fiscal years beginning after December 15, 2001 and interim periods within those fiscal years. Early application is encouraged. The provisions of SFAS No. 144 generally are to be applied prospectively. We are currently evaluating the impact of SFAS No. 144 on our financial position, results of operations and liquidity.

In April 2002, the FASB issued SFAS No. 145, Rescission of FASB Statements No. 4, 44 and 64, Amendment of FASB Statement No. 13, and Technical Corrections, which rescinds SFAS No. 4, Reporting Gains and Losses from Extinguishment of Debt, and an amendment of that Statement, SFAS No. 64, Extinguishments of Debt Made to Satisfy Sinking-Fund Requirements. SFAS No. 145 also rescinds SFAS No. 44, Accounting for Intangible Assets of Motor Carriers. SFAS No. 145 amends SFAS No. 13, Accounting for Leases, to eliminate an inconsistency between the required accounting for sale-leaseback transactions and the required accounting for certain lease modifications that have economic effects that are similar to sale-leaseback transactions. SFAS No. 145 also amends other existing authoritative pronouncements to make various technical corrections, clarify meanings, or describe their applicability under changed conditions.

The provisions of SFAS No. 145 related to the rescission of SFAS No. 4 shall be applied in fiscal years beginning after May 15, 2002. The provisions of SFAS No. 145 related to SFAS No. 13 shall be effective for transactions occurring after May 15, 2002. All other provisions of SFAS No. 145 shall be effective for financial statements issued on or after May 15, 2001. We are currently evaluating the impact of SFAS No. 145 on our financial position, results of operations and liquidity.

As of April 30, 2002, there are no other new accounting standards issued, but not yet adopted by us, which are expected to be applicable to our financial position, operating results or financial statement disclosures.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

We do not engage in derivative financial instruments or derivative commodity instruments. As of April 30, 2002, our financial instruments are not exposed to significant market risk due to interest rate risk, foreign currency exchange risk, commodity price risk or equity price risk.

Item 1. Legal Proceedings

On September 27, 2000, the Equal Employment Opportunity Commission ("EEOC") filed a lawsuit under Title VII of the Civil Rights Act against the Company in the United States District Court for the Western District of Virginia. The lawsuit alleged a pattern or practice of discrimination on the bases of gender and race. The lawsuit sought injunctive and other relief and damages in an unspecified amount. On December 13, 2001, the parties reached an agreement as to the amount of a settlement (subject to final documentation and judicial review and approval). On February 20, 2002, the Company reached a final settlement of the case and the court issued a Consent Decree setting forth the terms of the settlement. Pursuant to the settlement and Consent Decree, the Company paid \$500,000 on or about February 20, 2002; and will pay \$175,000 in January 2003 and \$175,000 in January 2004, to satisfy any gender and race class claims; the Company paid \$75,000 on or about February 20, 2002 to one individual specifically named in the complaint; and will spend at least \$75,000 for the Company's planned diversity, recruitment and human resource management programs over the term of the Consent Decree. We recorded a charge in the fourth quarter of fiscal year 2001 in the amount of \$901,553 representing \$575,000 payable upon entry of the Consent Decree, as well as \$326,553 representing the present value of two equal payments in the amount of \$175,000 payable in January 2003 and 2004. As of April 30, 2002, we have paid \$575,000 and accreted \$6,621 of the discount associated with recording the liability at its present value. Of the remaining liability totaling \$333,174 as of April 30, 2002, \$170,238 has been reflected as current. The \$75,000 to be used for the Company's planned diversity, recruitment and human resource management programs will be expensed as incurred.

The Company was named as a defendant in two lawsuits filed in the United States District Court for the Southern District of New York seeking to compel the Company to authorize its transfer agent to transfer unregistered, restricted stock on the Company's stock ledger. The first suit was filed on October 22, 2001, by Bear, Stearns & Co. Inc. and Bear, Stearns Securities Corporation (collectively, "Bear Stearns"). The second suit was filed on October 26, 2001, by UBS PaineWebber Inc. ("PaineWebber"). In each case, PaineWebber and Bear Stearns sought injunctive relief with respect to common stock of the Company sold by them in the course of liquidating either repossessed shares or brokerage accounts of the Company's former Chairman, President and Chief Executive Officer, to cover personal margin loans made by the brokerage firms to him. Both suits also contained a claim of monetary damages caused by the alleged wrongful refusal by the Company to authorize the transfers in connection with the liquidations. During the quarter ended April 30, 2002, both suits were dismissed. PaineWebber voluntarily dismissed its lawsuit with prejudice, while the Bear Stearns lawsuit was voluntarily dismissed without prejudice.

The Company, Mr. Robert Kopstein, our former Chairman, President and Chief Executive Officer, two of the Company's officers and directors, Luke J. Huybrechts and Kenneth W. Harber, are named as defendants in a consolidated putative class action lawsuit pending in the United States District Court for the Western District of Virginia (the "Consolidated Suit"). The first class action lawsuit was filed on November 26, 2001, by Charles S. Farrell, Jr., on behalf of himself and others similarly situated. The second class action lawsuit was filed on December 14, 2001, by Lerner Group, on behalf of itself and others similarly situated. The third class action lawsuit was filed on December 27, 2001, by Richard Simone, on behalf of himself and others similarly situated. The fourth class action lawsuit was filed on January 31, 2002, by Charles H. Yeatts, on behalf of himself and others similarly situated. In each of the four suits, the defendants in the actions were the Company, Mr. Kopstein and various John Does (unidentified officers and/or directors of the Company during the class period described below). Pursuant to the Private Securities Litigation Reform Act of 1995, the United States District Court for

the Western District of Virginia appointed a group of shareholders as the lead plaintiff for the Consolidated Suit. The four suits were consolidated into a single action and the lead plaintiff filed a consolidated amended class action complaint on May 2, 2002. In the Consolidated Suit, the plaintiffs purport to represent purchasers of the Company's common stock during the period ranging from June 14, 2000, through September 26, 2001, (the putative class period), and allege that the defendants violated Sections 10(b) and 20 of the federal Securities Exchange Act of 1934 in making certain alleged misrepresentations and/or omitting to disclose material facts. The plaintiffs in the Consolidated Suit seek compensatory damages in an unspecified amount, as well as reasonable costs and expenses incurred in the cause of action, including attorneys' fees and expert fees. On June 10, 2002, the Defendants moved to dismiss the Consolidated Suit.

Management intends to vigorously defend the Consolidated Suit. The Company may, however, incur substantial costs in defending the Consolidated Suit, regardless of its merit or outcome. At this early stage in the Consolidated Suit, management cannot make a reasonable estimate of the monetary amount of its resolution, or estimate a range of reasonably possible losses, if any. If we are unsuccessful, we could be subject to damages that may be substantial and could have a material adverse effect on our financial position, results of operations and liquidity.

From time to time, the Company is involved in various other claims and legal actions arising in the ordinary course of business. In the opinion of management, the ultimate disposition of these matters will not have a material adverse effect on the Company's financial position, results of operations or liquidity.

Item 4. Submission of Matters to a Vote of Security Holders

The following information is furnished for matters submitted to a vote of security holders during the three months ended April 30, 2002:

- (a) The Annual Meeting of Shareholders of Optical Cable Corporation was held on March 12, 2002.
- (b) The name of each director elected at the meeting follows:

Neil D. Wilkin, Jr. Craig H. Weber Luke J. Huybrechts Kenneth W. Harber Randall H. Frazier John M. Holland

(c) A brief description of each matter voted upon at the meeting and the number of votes cast for, against or withheld, as well as the number of abstentions and broker non-votes, as to each such matter, including a separate tabulation with respect to each nominee for office follows:

 To elect the following six directors to serve for the terms of office specified in the proxy statement and until their successors are duly elected and qualified.

Director	Votes for	Votes against 	Votes abstaining	10 non-votes
Neil D. Wilkin, Jr.	33,286,956		1,187,321	
Craig H. Weber	33,824,940		649,337	
Luke J. Huybrechts	33,297,549		1,176,728	
Kenneth W. Harber	33,283,874		1,190,403	
Randall H. Frazier	33,837,822		636,455	
John M. Holland	33,844,397		629,880	

2. To ratify the selection of KPMG LLP as independent accountants for the Company for fiscal year 2002.

	Votes	Votes	Broker
Votes for	against	abstaining	non-votes
34,012,174	409,258	52,845	

Item 5. Other Information.

In February 2002, our Board appointed Craig H. Weber to serve as a new member of the Board.

On June 14, 2002, the Company announced the layoff of 28 employees, representing almost 15% of its full-time employees. The reductions included both production and staff positions. The layoffs were part of a company-wide downsizing and will result in a charge of approximately \$120,000 during the third fiscal quarter of 2002.

Also on June 14, 2002, the Company announced the receipt of notification from Nasdaq regarding the Company's noncompliance with the listing requirements for the Nasdaq National Market, where the Company's common stock is presently traded. As a Nasdaq National Market company, the Company is subject to all the requirements of its listing agreement with Nasdaq. The Company received notification from Nasdaq that its common stock failed to meet the minimum \$1.00 per share trading price requirement for continued inclusion on the Nasdaq National Market. The Company will be provided until August 13, 2002 to regain compliance. The Company is considering several options to regain compliance with the minimum price requirement. If compliance is not demonstrated by August 13, 2002, Nasdaq may decide to delist the Company. The Company would have the right to appeal any delisting decision.

- Item 6. Exhibits and Reports on Form 8-K.
 - (a) Exhibits required by Item 601 of Regulation S-K for the six months ended April 30, 2002:

Exhibit 3.1 - Amended and Restated Articles of Incorporation (incorporated by reference to Exhibit 1 to the Company's Form 8-A filed with the Commission on November 5, 2001).

Exhibit 3.2 - Bylaws of Optical Cable Corporation, as amended (filed as Exhibit 3.2 to the Registrant's Annual Report on Form 10-K for the fiscal year ended October 31, 1997 (file number 0-37022), and incorporated herein by reference)).

Exhibit 4.1 - Form of certificate representing preferred share purchase right (filed as Exhibit 5 to the Company's Form 8-A filed with the Commission on November 5, 2001 and incorporated by reference herein).

Exhibit 10.1 - Loan and Security Agreement by and among Congress Financial Corporation (as agent), Wachovia Bank, National Association (as lender) and Optical Cable Corporation dated April 18, 2002.

(b) Reports on Form 8-K filed during the three months ended April 30, 2002:

None.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

OPTICAL CABLE CORPORATION (Registrant)

Date: June 14, 2002

/s/Neil D. Wilkin, Jr. Neil D. Wilkin, Jr. President and Chief Financial Officer (principal executive officer and principal financial and accounting officer)

LOAN AND SECURITY AGREEMENT

By and among

Congress Financial Corporation

"Agent"

Wachovia Bank, National Association

"Lender"

and

Optical Cable Corporation

"Borrower"

Dated: April 18, 2002

____ BACKGROUND 1 SECTION 1. DEFINITIONS SECTION 2. CREDIT FACILITIES 10 2.1 Revolving Loans 10 2.2 Fixed Asset Subline 11 Availability Reserves 2.3 11 2.4 Equipment Line of Credit 12 SECTION 3. INTEREST AND FEES 12 3.1 Interest 12 3.2 Closing Fee 14 3.3 Servicing Fee 14 3.4 Unused Line Fee 14 Changes in Laws and Increased Costs of Loans 3.5 14 3.6 Adjustments to Interest Rate 15 SECTION 4. CONDITIONS PRECEDENT 15 4.1 Conditions Precedent to Initial Loans: 15 Conditions Precedent to All Loans 4.2 16 4.3 Lien on Real Property SECTION 5. GRANT OF SECURITY INTEREST 17 17 SECTION 6. COLLECTION AND ADMINISTRATION 18 6.1 Borrower's Loan Account 18 6.2 Statements 18 6.3 Collection of Accounts 18 6.4 Payments 19 6.5 Authorization to Make Loans 2.0 6.6 Use of Proceeds 2.0 SECTION 7. COLLATERAL REPORTING AND COVENANTS 2.0 7.1 Collateral Reporting 2.0 7.2 Accounts Covenants 21 7.3 Inventory Covenants 2.2 7.4 Equipment Covenants 23 7.5 Power of Attorney 23 7.6 Right to Cure 24 Access to Premises 7.7 24 SECTION 8. REPRESENTATIONS AND WARRANTIES 24 8.1 Corporate Existence, Power and Authority; Subsidiaries 24 8.2 Financial Statements; No Material Adverse 25 8.3 Chief Executive Office; Collateral Locations 25 8.4 Priority of Liens; Title to Properties 25 8.5 25 Tax Returns 8.6 25 Litigation 8.7 Compliance with Other Agreements and Applicable Laws 26 8.8 Bank Accounts 26 8.9 Accuracy and Completeness of Information 26

Page

8.10.	Environmental Compliance	26
8.11.	Solvency	27
8.12.	Employee Benefits	27
8.13.	Survival of Warranties; Cumulative	28
SECTION	9. AFFIRMATIVE AND NEGATIVE COVENANTS	28
9.1.	Maintenance of Existence	28
9.2.	New Collateral Locations	29
9.3.	Compliance with Laws, Regulations, etc	29
9.4.	Payment of Taxes and Claims	30
9.5.	Insurance	30
9.6.	Financial Statements and Other Information	30
9.7.	Sale of Assets, Consolidation, Merger, Dissolution, Etc	32
9.8.	Encumbrances	32
9.9.	Indebtedness	32
9.10.	Loans, Investments, Guarantees, Etc	33
9.11.	Dividends and Redemptions	33
9.12.	Transactions with Affiliates	34
9.13.	Additional Bank Accounts	34
9.14.	Adjusted Net Worth	34
9.15.	Costs and Expenses	34
9.16.	Compliance with ERISA	35
9.17.	Further Assurances	35
	10. EVENTS OF DEFAULT AND REMEDIES	36
10.1.	Events of Default	36
10.2.	Remedies	38
SECTION	11. JURY TRIAL WAIVER; OTHER WAIVERS AND CONSENTS; GOVERNING LAW	39
11.1.	Governing Law; Choice of Forum; Service of Process; Jury Trial Waiver	39
11.2.	Waiver of Notices	40
11.3.	Amendments and Waivers	40
11.4.	Waiver of Counterclaims	40
11.5.	Indemnification	40
	Waiver and Consent	41
SECTION	12. TERM OF AGREEMENT; MISCELLANEOUS	41
	Term	41
	Notices	42
12.3.	Partial Invalidity	43
12.4.	Successors	43
12.5.	Entire Agreement	43

ii

INDEX TO

EXHIBITS AND SCHEDULES

Exhibit A	Information Certificate
Schedule 6.5	Borrower's Authorized Representatives
Schedule 8.4	Permitted Liens
Schedule 8.6	Litigation
Schedule 8.8	Bank Accounts
Schedule 8.10	Environmental Matters
Schedule 8.13	Commercial Tort Claims
Schedule 8.14	Letter of Credit Right
Schedule 9.9	Permitted Indebtedness
Schedule 9.10	Permitted Loans, Advances and Guarantees
Schedule 9.11	Dividends and Redemptions

iii

This Loan and Security Agreement ("Agreement") dated April 18, 2002 is entered into by and among Congress Financial Corporation, as agent ("Agent"), Wachovia Bank, National Association, as lender ("Lender"), and Optical Cable Corporation, a Virginia corporation, as borrower ("Borrower").

BACKGROUND

A. Borrower has requested that Agent and Lender enter into certain financing arrangements pursuant to which Agent and Lender may make loans and provide other financial accommodation to Borrower to provide additional working capital to Borrower.

B. Agent and Lender are willing to make such loans and provide such financial accommodations on the terms and conditions set forth herein.

NOW, THEREFORE, in consideration of the mutual conditions and agreements set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged and intending to be legally bound hereby, the parties hereto agree as follows:

SECTION 1. DEFINITIONS.

All terms used herein which are defined in Article 1 or Article 9 of the Uniform Commercial Code ("UCC"), as revised and in effect from time to time in the State of New York, shall have the meanings given therein unless otherwise defined in this Agreement. All references to the plural herein shall also mean the singular and to the singular shall also mean the plural unless the context otherwise requires. All references to Borrower, Agent and Lender pursuant to the definitions set forth in the recitals hereto, or to any other person herein, shall include their respective successors and assigns. The words "hereof", "herein", "hereunder", "this Agreement" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not any particular provision of this Agreement and as this Agreement now exists or may hereafter be amended, modified, supplemented, extended, renewed, restated or replaced. An Event of Default shall exist or continue or be continuing until such Event of Default is waived in accordance with Section 11.3 or is cured in a manner satisfactory to Agent, if such Event of Default is capable of being cured as determined by Agent. Any accounting term used herein unless otherwise defined in this Agreement shall have the meanings customarily given to such term in accordance with GAAP. For purposes of this Agreement, the following terms shall have the respective meanings given to them below:

1.1. "Account Debtor" shall mean any person obligated on an Account.

1.2. "Account" shall have the meaning ascribed thereto in the UCC.

1.3. "Adjusted Eurodollar Rate" shall mean, with respect to each Interest Period for any Eurodollar Rate Loan, the rate per annum (rounded upwards, if necessary, to the next one-sixteenth (1/16) of one (1.0%) percent) determined by dividing (a) the Eurodollar Rate for such Interest Period by (b) a percentage equal to: (i) one (1) minus (ii) the Reserve Percentage. For purposes hereof, "Reserve Percentage" shall mean the reserve percentage, expressed as a decimal, prescribed by any United States or foreign banking authority for determining the reserve requirement which is or would be applicable to deposits of United States dollars in a non-United States or an international banking office of Reference Bank used to fund a Eurodollar Rate Loan or any Eurodollar Rate Loan made with the proceeds of such deposit, whether or not the Reference Bank actually holds or has made any such deposits or loans. The Adjusted Eurodollar Rate shall be adjusted on and as of the effective day of any change in the Reserve Percentage.

1.4. "Adjusted Net Worth" shall mean as to Borrower, at any time, in accordance with GAAP (except as otherwise specifically set forth below), on a consolidated basis for such Person and its subsidiaries (if any), the amount equal to (a) the difference between: (i) the aggregate net book value of all tangible assets of such Person and its subsidiaries, calculating the book value of inventory for this purpose on a first-in-first-out basis, after deducting from such book values all appropriate reserves in accordance with GAAP (including all reserves for doubtful receivables, obsolescence, depreciation and amortization and excluding gains or losses from any of the retirement plans of Borrower) and (ii) the aggregate amount of the indebtedness and other liabilities of such Person and its subsidiaries (including tax and other proper accruals but excluding deferred taxes and excluding non-cash charges as to stock options); plus (b) indebtedness of such Person or its subsidiaries which is subordinated in right of payment to the full and final payment of all of the Obligations or terms and conditions acceptable to Agent.

1.5. "Applicable Eurodollar Margin" shall mean a marginal rate of interest which is added to the Adjusted Eurodollar Rate to determine the effective Interest Rate on Eurodollar Rate Loans. The Applicable Eurodollar Margin on Eurodollar Rate Loans shall be three and one-quarter percent (3.25%), as may be adjusted pursuant to Section 3.6 of this Agreement.

1.6. "Applicable Margin" shall mean a marginal rate of interest which is added to the Base Rate to determine the effective Interest Rate on Base Rate Loans. The Applicable Margin shall be three quarters of one percent (.75%), as may be adjusted pursuant to Section 3.6 of this Agreement.

1.7. "Availability Reserves" shall mean, as of any date of determination, such amounts as Agent may from time to time establish and revise in good faith reducing the amount of Revolving Loans which would otherwise be available to Borrower under the lending formula(s) provided for herein: (a) to reflect events, conditions, contingencies or risks which, as determined by Agent in good faith, do or may affect either (i) the Collateral or any other property which is security for the Obligations or its value, (ii) the assets, business or prospects of Borrower or any Obligor or (iii) the security interests and other rights of Agent in the Collateral (including the enforceability, perfection and priority thereof) or (b) to reflect Agent's good faith belief that any collateral report or financial information furnished by or on behalf of Borrower or any Obligor to Agent is or may have been incomplete, inaccurate or misleading in any material respect or (c) in respect of any state of facts which Agent determines in good faith constitutes an Event of Default or may, with passage of time or giving of notice, or both, constitute an Event of Default.

1.8. "Base Rate" shall mean the rate announced by the Reference Bank or its successors, from time to time, as its prime rate, whether or not such announced rate is the best rate available at such bank.

1.9. "Base Rate Loans" shall mean loans under the Revolving Credit, or any portion thereof, on which interest is calculated under the terms hereof based on the Base Rate plus the Applicable Margin.

1.10. "Blocked Accounts" shall have the meaning set forth in Section 6.3 hereof.

1.11. "Business Day" shall mean any day other than a Saturday, Sunday, or other day on which commercial banks are authorized or required to close under the laws of the State of New York.

1.12. "Capital Expenditures" shall mean, as applied to any Person for any period, the aggregate of all expenditures (including that portion attributable to any lease of any property, whether real, personal or mixed, by that Person or lessee which is accounted for as a capital lease on the balance sheet of such Person prepared in accordance with GAAP, incurred during that period), made by that Person during such period in respect of the purchase, construction or other acquisition of fixed or capital assets, determined in accordance with GAAP.

1.13. "Change of Control" shall mean, after the closing date of this Agreement, the acquisition by any Person, either individually or acting in concert with one or more other Persons, of beneficial ownership, directly or indirectly, of capital stock of Borrower (or other securities convertible into such capital stock) representing 20% or more of the combined voting power of all capital stock of Borrower entitled to vote in the election of members of the Board of Directors of Borrower. As used herein, the term "beneficially own" or "beneficial ownership" shall have the meaning set forth in the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

1.14. "Code" shall mean the Internal Revenue Code of 1986, as the same now exists or may from time to time hereafter be amended, modified, re-codified or supplemented, together with all rules, regulations and interpretations thereunder or related thereto.

1.15. "Collateral" shall have the meaning set forth in Section 5 hereof.

1.16. "Commercial Tort Claims" shall have the meaning set forth in the UCC.

1.17. "Deed of Trust" shall mean that certain mortgage or deed of trust, as the same may be amended, modified, supplemented, renewed, extended, replaced or restated from time to time, to be executed by Borrower on or prior to the date of this Agreement under which Borrower shall grant and convey to Agent, or to trustees for the benefit of Agent, as security for the Obligations, a lien upon the Real Property.

1.18. "Default Rate" shall mean with respect to any amounts payable hereunder or under the Financing Agreements, a rate equal to (a) the sum of (i) two percent (2.0%) per annum and (ii) the Interest Rate otherwise in effect with respect to such amounts, or, (b) if no such Interest Rate is otherwise in effect with respect to such amounts, a rate equal to the sum of (i) the Base Rate plus (ii) the highest Applicable Margin thereon plus (iii) two percent (2.0%).

1.19. "EBITDA" shall mean for any period, the sum of Borrower's (a) net income (or deficit), plus (b) Interest Expense, plus (c) depreciation, amortization and income tax expense, plus (d) extraordinary non-cash losses, plus (e) accruals for stock options to the extent the same is a charge against Borrower's net income, minus (f) extraordinary gains, all as determined in accordance with GAAP.

1.20. "Eligible Accounts" shall mean Accounts created by Borrower which are and

continue to be acceptable to Agent (in its sole discretion) based on the criteria set forth below. In general, Accounts shall be Eligible Accounts if:

 (a) Such Accounts arise from the actual and bona fide sale and delivery of goods by Borrower or rendition of services by Borrower in the ordinary course of its business which transactions are completed in accordance with the terms and provisions contained in any documents related thereto;

(b) Such Accounts are not unpaid more than the earlier of sixty (60) days from their due date or ninety (90) days after the date of the original invoice for them;

(c) Such Accounts comply with the terms and conditions contained in Section 7.2(c) of this Agreement;

(d) Such Accounts do not arise from sales on consignment, guaranteed sale, sale and return, sale on approval, or other terms under which payment by the Account Debtor may be conditional or contingent;

(e) The chief executive office of the Account Debtor with respect to such Accounts is located in the United States of America, or, at Agent's option, if either: (i) the Account Debtor has delivered to Borrower an irrevocable letter of credit issued or confirmed by a bank satisfactory to Agent sufficient to cover such Account, in form and substance satisfactory to Agent and, if required by Agent, the original of such letter of credit has been delivered to Agent or Agent's agent and the issuer thereof notified of the assignment of the proceeds of such letter of credit to Agent, or (ii) such Account is subject to credit insurance payable to Agent issued by an insurer and on terms and in an amount acceptable to Agent, or (iii) such Account is otherwise acceptable in all respects to Agent in its sole and absolute discretion (subject to such lending formula with respect thereto as Agent may determine);

(f) Such Accounts do not consist of progress billings, bill and hold invoices or retainage invoices; except as to bill and hold invoices, if Agent has received an agreement in writing from the Account Debtor, in form and substance satisfactory to Agent, confirming the unconditional obligation of the Account Debtor to take the goods related thereto and pay such invoice;

(g) The Account Debtor with respect to such Accounts has not asserted a counterclaim, defense or dispute and does not have, and does not engage in transactions which may give rise to, any right of setoff against such Accounts (but the portion of the Accounts of such Account Debtor in excess of the amount at any time and from time to time owed by Borrower to such Account Debtor or claimed owed by such Account Debtor may be deemed Eligible Accounts);

(h) There are no facts, events or occurrences which would impair the validity, enforceability or collectibility of such Accounts or reduce the amount payable or delay payment thereunder;

Such Accounts are subject to the first priority, valid and perfected security interest of Agent and any goods giving rise thereto are not, and were not at the time of the sale thereof, subject to any liens except those permitted in this Agreement;

(j) Neither the Account Debtor nor any officer or employee of the Account Debtor with respect to such Accounts is an officer, employee or agent of or affiliated with Borrower directly or indirectly by virtue of family membership, ownership, control, management or otherwise;

(k) Unless approved by Agent in writing (such approval to be in the sole and absolute discretion of Agent), the Account Debtors with respect to such Accounts are not any foreign government, the United States of America, any State, political subdivision, department, agency or instrumentality thereof, unless, if the Account Debtor is the United States of America, any State, political subdivision, department, agency or instrumentality thereof, upon Agent's request, the Federal Assignment of Claims Act of 1940, as amended or any similar State or local law, if applicable, has been complied with in a manner satisfactory to Agent;

(1) There are no proceedings or actions which are threatened or pending against the Account Debtors with respect to such Accounts which might result in any material adverse change in any such Account Debtor's financial condition;

(m) Such Accounts of a single Account Debtor or its affiliates
 (other than governmental agencies) do not constitute more than twenty percent
 (20%) of all otherwise Eligible Accounts (but the portion of the Accounts not in
 excess of such percentage may be deemed Eligible Accounts);

 (n) Such Accounts are not owed by an Account Debtor who has Accounts unpaid more than ninety (90) days after the date of the original invoice for them or sixty (60) days past the due date, whichever is sooner, which constitute more than fifty (50%) percent of the total Accounts of such Account Debtor;

(o) Such Accounts are owed by Account Debtors whose total indebtedness to Borrower does not exceed the credit limit with respect to such Account Debtors as may be reasonably determined by Agent from time to time (but the portion of the Accounts not in excess of such credit limit may be deemed Eligible Accounts); and

(p) Such Accounts are owed by Account Debtors deemed creditworthy at all times by Agent, as reasonably determined by Agent.

General criteria for Eligible Accounts may be established and revised from time to time by Agent in good faith. Any Accounts which are not Eligible Accounts shall nevertheless be part of the Collateral.

1.21. "Eligible Equipment" shall mean Equipment which is acceptable to Agent based on the criteria set forth below. In general, Eligible Equipment shall not include: (a) obsolete or worn out Equipment; (b) spare parts for Equipment; (c) Equipment at premises other than those owned and controlled by Borrower, unless Agent shall have received a Waiver from the owner or operator of such premises in form and substance satisfactory to Agent; (d) Equipment located at any affiliate of Borrower, unless Agent has received a Waiver from such affiliate; (e) Equipment subject to a security interest or lien in favor of any person other than Agent except those permitted in this Agreement; (f) Equipment which is not subject to the first priority, valid and perfected security interest of Agent; and (g) damaged and/or defective Equipment. General criteria for Eligible Equipment may be established and revised from time to time by Agent in good faith. Any

Equipment which is not Eligible Equipment shall nevertheless be part of the Collateral.

1.22. "Eligible Inventory" shall mean Inventory consisting of finished goods held for resale in the ordinary course of the business of Borrower and raw materials for such finished goods which are acceptable to Agent based on the criteria set forth below. In general, Eligible Inventory shall not include: (a) work-in-process; (b) spare parts for Equipment; (c) packaging and shipping materials; (d) supplies used or consumed in Borrower's business; (e) Inventory at premises other than those owned and controlled by Borrower, unless Agent shall have received a Waiver from the person in possession of such Inventory and/or the owner or operator of such premises in form and substance satisfactory to Agent; (f) Inventory located at any affiliate of the Borrower, unless Agent has received a Waiver from such affiliate; (g) Inventory subject to a security interest or lien in favor of any person other than Agent except those permitted in this Agreement; (h) bill and hold goods; (i) stale, spoiled, obsolete or slow moving Inventory; (j) Inventory which is not subject to the first priority, valid and perfected security interest of Agent; (k) returned, damaged and/or defective Inventory; (1) Inventory purchased or sold on consignment; and (m) Inventory commingled with material or inventory of any of Borrower's customers or any other third party. General criteria for Eligible Inventory may be established and revised from time to time by Agent in good faith. Any Inventory which is not Eligible Inventory shall nevertheless be part of the Collateral.

1.23. "Environmental Laws" shall mean any federal or state statute, ordinance, law, rule or regulation at any time enacted or adopted by any federal, state or local government, governmental agency or other governmental or quasi-governmental entity pertaining to environmental matters, including, without limitation, the Federal Comprehensive Environmental Response Compensation Liability Act ("CERCLA"), Environmental Cleanup Responsibility Act ("ECRA"), and the Federal Resource Conservation Recovery Act ("RCRA"), or to the releasing, spilling, leaking, pumping, pouring, emitting, emptying, dumping or otherwise disposing of Hazardous Waste (as defined in 12 U.S.C. (S)6903(5)) or Hazardous Materials.

1.24. "Equipment" shall have the meaning ascribed thereto in the UCC and shall include all of Borrower's now owned and hereafter acquired equipment, machinery, computers, computer hardware and software (whether owned or licensed), vehicles, tools, furniture, fixtures, all attachments, accessions and property now or hereafter affixed thereto or used in connection therewith, and substitutions and replacements thereof, wherever located.

1.25. "ERISA" shall mean the United States Employee Retirement Income Security Act of 1974, as the same now exists or may hereafter from time to time be amended, modified, re-codified or supplemented, together with all rules, regulations and interpretations thereunder or related thereto.

1.26. "ERISA Affiliate" shall mean any person required to be aggregated with Borrower or any of its Subsidiaries under Sections 414(b), 414(c), 414(m) or 414(o) of the Code.

1.27. "Eurodollar Rate" shall mean with respect to the Interest Period for a Eurodollar Rate Loan, the interest rate per annum equal to the arithmetic average of the rates of interest per annum (rounded upwards, if necessary, to the next one-sixteenth (1/16) of one (1.0%) percent) at which Reference Bank is offered deposits of United States dollars in the London interbank market (or other Eurodollar Rate market selected by Borrower and approved by Agent) on or about 9:00 a.m. (New York time) two (2) Business Days prior to the commencement of such Interest Period in

amounts substantially equal to the principal amount of the Eurodollar Rate Loans requested by and available to Borrower in accordance with this Agreement, with a maturity of comparable duration to the Interest Period selected by Borrower.

1.28. "Eurodollar Rate Loans" shall mean loans under the Revolving Credit, or any portion thereof, on which interest is calculated under the terms hereof based on the Adjusted Eurodollar Rate plus the Applicable Eurodollar Margin.

1.29. "Event of Default" shall have the meaning set forth in Section 10.1 hereof.

1.30. "Excess Availability" shall mean the amount, as determined by Agent, calculated at any time, equal to: (a) the lesser of (i) the amount of the Revolving Loans available to Borrower as of such time based on the applicable lending formula, and subject to the sublimits and Availability Reserves from time to time established by Agent hereunder or (ii) the Maximum Credit, minus (b) the sum of (i) the amount of all then outstanding and unpaid Revolving Loans and (ii) the aggregate amount of all trade payables of Borrower which are more than sixty (60) days past due as of such time.

1.31. "Financing Agreements" shall mean, collectively, this Agreement and all notes, guarantees, security agreements and other agreements, documents and instruments now or at any time hereafter executed and/or delivered by Borrower or any Obligor in connection with this Agreement, as the same now exist or may hereafter be amended, modified, supplemented, extended, renewed, restated or replaced.

1.32. "Fixed Charge Coverage Ratio" shall mean, as of the last day of each fiscal quarter of Borrower, the ratio of (a) EBITDA to (b) Fixed Charges, all as calculated for the four fiscal quarterly periods then ended and all determined on a consolidated basis in accordance with GAAP.

1.33. "Fixed Charges" shall mean, for any applicable period of computation, without duplication, the sum of (a) all Interest Expense for the applicable period, plus (b) all income tax expense for the applicable period, plus (c) scheduled principal payments on account of long-term Indebtedness made during the applicable period, plus (d) cash dividends or other cash distributions paid by Borrower during the applicable period, plus (e) Capital Expenditures made during the applicable period, all as determined in accordance with GAAP.

1.34. "GAAP" shall mean generally accepted accounting principles in the United States of America as in effect from time to time as set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and the statements and pronouncements of the Financial Accounting Standards Board which are applicable to the circumstances as of the date of determination consistently applied.

1.35. "Free Cash Flow" shall mean, for the prior annual period, the greater of (a) Zero Dollars (\$0) or (b) without duplication, net income, plus depreciation, minus Capital Expenditures.

1.36. "Hazardous Materials" shall mean any hazardous, toxic or dangerous substances, materials and wastes, including hydrocarbons (including naturally occurring or man-made petroleum and hydrocarbons), flammable explosives, asbestos, urea formaldehyde insulation, radioactive materials, biological substances, polychlorinated biphenyls, pesticides, herbicides and any other kind and/or type of pollutants or contaminants (including materials which include hazardous

constituents), sewage, sludge, industrial slag, solvents and/or any other similar substances, materials, or wastes and including any other substances, materials or wastes that are or become regulated under any Environmental Law (including any that are or become classified as hazardous or toxic under any Environmental Law).

1.37. "Information Certificate" shall mean the Information Certificate of Borrower attached hereto as Exhibit A containing material information with respect to Borrower, its business and assets, provided by or on behalf of Borrower to Agent in connection with the preparation of this Agreement and the other Financing Agreements and the financing arrangements provided for herein.

1.38. "Interest Expense" shall mean for any period, the aggregate amount of interest expense required to be paid or accrued during such period for all indebtedness of Borrower outstanding during all or any part of such period, as determined in accordance with GAAP.

1.39. "Interest Period" shall mean for any Eurodollar Rate Loan, a period of approximately one (1), two (2) or three (3) months duration as Borrower may elect, the exact duration to be determined in accordance with the customary practice in the applicable Eurodollar Rate market; provided, that, Borrower may not elect an Interest Period which will end after the last day of the then-current term of this Agreement.

1.40. "Interest Rate" shall mean (a) as to Base Rate Loans, a rate per annum equal to the Base Rate plus the Applicable Margin, and (b) as to Eurodollar Rate Loans, a rate per annum equal to the Applicable Eurodollar Margin plus the Adjusted Eurodollar Rate (based on the Eurodollar Rate applicable for the Interest Period selected by Borrower as in effect three (3) Business Days after the date of receipt by Agent of the request of Borrower for such Eurodollar Rate Loans in accordance with the terms hereof, whether such rate is higher or lower than any rate previously quoted to Borrower).

1.41. "Inventory" shall have the meaning ascribed thereto in the UCC and shall include all of Borrower's now owned and hereafter existing or acquired raw materials, work in process, finished goods and all other inventory of whatsoever kind or nature, wherever located.

1.42. "Investment Property" shall have the meaning ascribed thereto in the UCC.

1.43. "Letter of Credit Rights" shall have the meaning ascribed thereto in the UCC.

1.44. "Loans" shall mean the Revolving Loans.

1.45. "Material Adverse Effect" shall mean a material adverse effect on (i) the business, assets or prospects of Borrower or Borrower and its subsidiaries, considered as a whole, (ii) the ability of Borrower to perform its obligations under the Financing Agreements or (iii) the ability of Agent to enforce any of the Obligations or to realize upon any Collateral.

1.46. "Maximum Credit" shall mean the amount of Twenty Five Million Dollars ($25,000,000)\,.$

1.47. "MERE Adjustment Date" shall mean for the purposes of Section 2.3, the date that Agent has adjusted the MERE Sublimit pursuant to Section 2.3(b).

1.48. "MERE Adjustment Period" shall mean the period commencing on a MERE Adjustment Date and ending on the next MERE Adjustment Date, but in no event shall a MERE Adjustment Period be less than one hundred eighty days (180).

1.49. "MERE Sublimit" shall mean an amount equal to the lesser of (a) the sum of eighty percent (80%) of the net orderly liquidation value (as reasonably determined by Agent) of Eligible Equipment plus sixty percent (60%) of the fair market value (as reasonably determined by Agent) of the Real Property and (b) Five Million Dollars (\$5,000,000), subject to the adjustments as provided in Section 2.3.

1.50. "Net Amount of Eligible Accounts" shall mean the gross amount of Eligible Accounts less (a) sales, excise or similar taxes included in the amount thereof and (b) returns, discounts, claims, credits and allowances of any nature at any time issued, owing, granted, outstanding, available or claimed with respect thereto.

1.51. "Obligations" shall mean any and all Revolving Loans, and all other obligations, liabilities and indebtedness of every kind, nature and description owing by Borrower to Agent and/or Lender and/or its affiliates, including principal, interest, charges, fees, costs and expenses, however evidenced, whether as principal, surety, endorser, guarantor or otherwise, whether arising under this Agreement or otherwise, whether now existing or hereafter arising, whether arising before, during or after the initial or any renewal term of this Agreement or after the commencement of any case with respect to Borrower under the United States Bankruptcy Code or any similar statute (including the payment of interest and other amounts which would accrue and become due but for the commencement of such case, whether or not such amounts are allowed or allowable in whole or in part in such case), whether direct or indirect, absolute or contingent, joint or several, due or not due, primary or secondary, liquidated or unliquidated, secured or unsecured, and however acquired by Agent and/or Lender.

1.52. "Obligor" shall mean any guarantor, endorser, acceptor, surety or other person liable on or with respect to the Obligations or who is the owner of any property which is security for the Obligations, other than Borrower.

1.53. "Payment Account" shall have the meaning set forth in Section 6.3 hereof.

1.54. "Person" or "person" shall mean any individual, sole proprietorship, partnership, limited liability company or partnership, corporation (including any corporation which elects subchapter S status under the Internal Revenue Code of 1986, as amended), business trust, unincorporated association, joint stock corporation, trust, joint venture or other entity or any government or any agency or instrumentality or political subdivision thereof.

1.55. "Real Property" shall mean that certain real property of Borrower located at 5290 Concourse Drive, Roanoke, Virginia 24019.

1.56. "Records" shall mean all of Borrower's present and future books of account of every kind or nature, purchase and sale agreements, invoices, ledger cards, bills of lading and other shipping evidence, statements, correspondence, memoranda, credit files and other data relating to the Collateral or any Account Debtor, together with the tapes, disks, diskettes and other data and software storage media and devices, file cabinets or containers in or on which the foregoing are

stored (including any rights of Borrower with respect to the foregoing maintained with or by any other person).

1.57. "Reference Bank" shall mean Wachovia Bank, National Association or such other bank as Agent may from time to time designate in its sole discretion.

1.58. "Revolving Loans" shall mean the loans now or hereafter made by Lender to or for the benefit of Borrower on a revolving basis (including advances, repayments and readvances) as set forth in Section 2.1 hereof.

1.59. "Settlements" shall mean the settlements of litigation listed on Schedule 8.6 attached hereto.

1.60. "Value" shall mean, as determined by Agent its sole discretion, the lower of cost computed on a first in-first out basis in accordance with GAAP or market value.

1.61. "Waiver" shall mean an agreement in writing from a Person in possession of Collateral and/or the owner or operator of premises other than those owned and controlled by Borrower on or in which Collateral is located, in form and substance satisfactory to Agent acknowledging Agent's first priority security interest in the Collateral, waiving security interests and claims by such Person against the Collateral and permitting Agent access to, and the rights to remain on, the premises so as to exercise Agent's rights and remedies and otherwise deal with the Collateral including removal of the Collateral from the premises.

SECTION 2. CREDIT FACILITIES.

2.1. Revolving Loans.

(a) Subject to and upon the terms and conditions contained herein, Lender agrees to make Revolving Loans to Borrower from time to time through the Maturity Date pursuant to the terms hereof in amounts requested by Borrower up to the amount equal to the sum of:

(i) eighty-five percent (85%) of the Net Amount of Eligible Accounts, plus

(ii) (A) until such time as Borrower conducts and receives an Inventory appraisal that is acceptable to Agent (which Agent agrees to review within thirty (30) days of receipt), the lower of (1) up to 10% of the Value of Eligible Inventory, or (2) Seven Million Five Hundred Thousand Dollars (\$7,500,000); and (B) thereafter the lowest of: (1) eighty-five percent (85%) of the net orderly liquidation value (as reasonably determined by Agent) of Eligible Inventory; (2) the sum of fifty percent (50%) of the Value of Eligible Inventory consisting of raw materials plus sixty percent (60%) of the Value of Eligible Inventory consisting of finished goods; or (3) Seven Million Five Hundred Thousand Dollars (\$7,500,000); plus

(iii) the MERE Sublimit; minus

(iv) any Availability Reserves.

(b) Agent may, in its sole discretion, from time to time, upon not less than five $\left(5\right)$

days prior notice to Borrower: (i) reduce the lending formula with respect to Eligible Accounts to the extent that Agent determines in good faith that: (A) the dilution with respect to the Accounts for any period (based on the ratio of the (i) aggregate amount of reductions in Accounts other than as a result of payments in cash to (ii) the aggregate amount of total sales) has increased in any material respect or may be reasonably anticipated to increase in any material respect above historical levels, or (B) the general creditworthiness of Account Debtors has declined or (ii) reduce the lending formula(s) with respect to Eligible Inventory to the extent that Agent determines that: (A) the number of days of the turnover of the Inventory for any period has changed in any material respect or (B) the liquidation value of the Eligible Inventory, or any category thereof, has decreased, or (C) the nature and quality of the Inventory has deteriorated. In determining whether to reduce the lending formula(s), Agent may consider events, conditions, contingencies or risks which are also considered in determining Eligible Accounts, Eligible Inventory or in establishing Availability Reserves.

(c) Except in Agent's sole discretion the aggregate amount of Loans outstanding at any time shall not exceed the Maximum Credit. In the event that the aggregate amount of the outstanding Loans exceed the amounts available under the lending formulas, such event shall not limit, waive or otherwise affect any rights of Agent in that circumstance or on any future occasions and Borrower shall, upon demand by Agent, which may be made at any time or from time to time, immediately repay to Agent and/or Lender the entire amount of any such excess(es) for which payment is demanded.

2.2. Availability Reserves. All Revolving Loans otherwise available to Borrower pursuant to the lending formulas and subject to the Maximum Credit and other applicable limits hereunder shall be subject to Agent's continuing right to establish and revise Availability Reserves. Without limiting any other rights or remedies of Agent under this Agreement or any of the other Financing Agreements with respect to the establishment of Availability Reserves or otherwise, Agent may establish and revise Availability Reserves to reflect: (a) Inventory shrinkage; (b) the aggregate amount of deposits, if any, received by Borrower from its customers in respect of unfilled orders for merchandise; and (c) amounts due or to become due in respect of sales, use and/or withholding taxes.

2.3. MERE Sublimit.

(a) Commencing on June 1, 2002, and continuing on the first day of each calendar month thereafter until the first MERE Adjustment Date, the MERE Sublimit shall be decreased by an amount equal to Eighty Three Thousand Three Hundred Thirty Three Dollars (\$83,333).

(b) On or after October 18, 2002, Borrower may request that Agent reset the MERE Sublimit based upon a Collateral appraisal of Eligible Equipment and the Real Property conducted by Agent, subject to the terms hereof. So long as (a) Borrower's average Excess Availability for the most recently ended fiscal quarter exceeds Two Million Dollars (\$2,000,000), (b) no Event of Default has occurred, or event which, with the passage of time or giving of notice, or both, would constitute an Event of Default has occurred and is continuing and (c) Borrower's Fixed Charge Coverage Ratio is greater than 1.20 to 1, on the first day of the calendar month following such determinations (each a "MERE Adjustment Date") Agent shall, at the request of Borrower, increase the MERE Sublimit to an amount equal to the lesser of (i) the sum of eighty percent (80%)

of the then net orderly liquidation value of Eligible Equipment (as reasonably determined by Agent) plus sixty percent (60%) of the then fair market value of the Real Property (as reasonably determined by Agent) and (ii) Five Million Dollars (\$5,000,000). Agent shall reset the MERE Sublimit within a reasonable period of time after Agent's receipt of such appraisals and receipt of such information required by Agent, in its reasonable discretion.

(c) (i) Commencing the first day of the first month after each MERE Adjustment Date, the MERE Sublimit, as adjusted from time to time, shall be reduced on the first day of each calendar month during the applicable MERE Adjustment Period by an amount equal to one sixtieth (1/60) of the MERE Sublimit as determined on the MERE Adjustment Date of such MERE Adjustment Period; and

(ii) On an annual basis, upon Agent's receipt of the annual audited financial statements of Borrower, the MERE Sublimit shall be further reduced by an amount equal to twenty percent (20%) of Free Cash Flow.

(d) For purposes of this Section 2.3, Agent shall not be required to rely upon an Equipment appraisal greater than thirty (30) days old or a Real Property appraisal greater that one (1) year old and if Borrower has requested that the MERE Sublimit be reset pursuant to this Section 2.3, Borrower shall deliver to Agent, at Agent's reasonable request, a written statement from the appraiser of the Real Property that it is not aware of any circumstances which would materially and adversely affect its most recent appraisal.

SECTION 3. INTEREST AND FEES.

3.1. Interest.

(a) Borrower shall pay to Agent for Lender interest on the outstanding principal amount of the Obligations at the Interest Rate, except that Borrower shall pay to Agent for Lender interest, at Agent's option, without notice, at the Default Rate: (i) on the Obligations for the period from and after the date of termination or non-renewal hereof, or the date of the occurrence of an Event of Default, and for so long as such Event of Default is continuing as determined by Agent and until such time as Agent and Lender have received full and final payment of all Obligations (notwithstanding any entry of judgment against the Borrower); and (ii) on the Revolving Loans at any time outstanding in excess of the amounts available to Borrower under Section 2 (whether or not such excess(es), arise or are made with or without Agent's or Lender's knowledge or consent and whether made before or after an Event of Default). Borrower acknowledges that: (i) such additional rate is a material inducement to Agent and Lender to extend the Revolving Loans; (ii) neither Agent nor Lender would have extended the Revolving Loans in the absence of the agreement of Borrower to pay such additional rate; (iii) such additional rate represents compensation for increased risk to Lender that the Revolving Loans will not be repaid; and (iv) such rate is not a penalty and represents a reasonable estimate of (a) the cost to Agent and Lender in allocating their resources (both personnel and financial) to the ongoing review, monitoring, administration and collection of the Revolving Loans and (b) compensation to Agent and Lender for losses that are difficult to ascertain. All interest accruing hereunder on and after the occurrence of any of the events referred to in Sections 3.1(a)(i) and 3.1 (a)(ii) above shall be payable in demand.

(b) Borrower may from time to time request that a Loan be a Eurodollar Rate Loan, $% \left({{{\left[{{{\rm{T}}_{\rm{T}}} \right]}}} \right)$

that a Base Rate Loans be converted to Eurodollar Rate Loans or that any existing Eurodollar Rate Loans continue for an additional Interest Period. Such request from Borrower shall specify the amount of a Eurodollar Rate Loan, or of the Base Rate Loans which will be converted to Eurodollar Rate Loans (subject to the limits set forth below) and the Interest Period to be applicable to such Eurodollar Rate Loans. Subject to the terms and conditions contained herein, three (3) Business Days after receipt by Agent of such a request from Borrower, such Base Rate Loans shall be converted to Eurodollar Rate Loans or such Eurodollar Rate Loans shall continue as Eurodollar Rate Loans as of the end of such Interest Period, as the case may be, provided, that, (i) no Event of Default exists, or event which with passage of time would constitute an Event of Default exists or has occurred and is continuing, (ii) no party hereto shall have sent any notice of termination or non-renewal of this Agreement, (iii) Borrower shall have complied with such customary procedures as are established by Agent and specified by Agent to Borrower from time to time for requests by Borrower for Eurodollar Rate Loans, (iv) no more than five (5) Interest Periods may be in effect at any one time, (v) the aggregate amount of the Eurodollar Rate Loans must be in an amount not less than One Million Dollars (\$1,000,000) or an integral multiple of Five Hundred Thousand Dollars (\$500,000) in excess thereof, (vi) the maximum amount of the Eurodollar Rate Loans at any time requested by Borrower shall not exceed the amount equal to ninety percent (90%) of the lowest principal amount of the Revolving Loans which it is anticipated will be outstanding during the applicable Interest Period, in each case as determined by Agent (but with no obligation of Lender to make such Revolving Loans); and (vii) Agent shall have determined that the Interest Period or Adjusted Eurodollar Rate is available to Lender through the Reference Bank and can be readily determined as of the date of the request for such Eurodollar Rate Loan by Borrower. Any request by Borrower to convert Base Rate Loans to Eurodollar Rate Loans or to continue any existing Eurodollar Rate Loans shall be irrevocable. Notwithstanding anything to the contrary contained herein, Lender and Reference Bank shall not be required to purchase United States Dollar deposits in the London interbank market or other applicable Eurodollar Rate market to fund any Eurodollar Rate Loans, but the provisions hereof shall be deemed to apply as if Lender and Reference Bank had purchased such deposits to fund the Eurodollar Rate Loans.

Any Eurodollar Rate Loans shall automatically convert to Base Rate (C) Loans upon the last day of the applicable Interest Period, unless Agent has received and approved a request to continue such Eurodollar Rate Loan at least three (3) Business Days prior to such last day in accordance with the terms hereof. Any Eurodollar Rate Loans shall, at Agent's option, upon notice by Agent to Borrower, convert to Base Rate Loans in the event that (i) an Event of Default exists or event which, with the passage of time or giving of notice, or both, would constitute an Event of Default, shall exist, (ii) this Agreement shall terminate or not be renewed, or (iii) the aggregate principal amount of the Base Rate Loans which have previously been converted to Eurodollar Rate Loans or existing Eurodollar Rate Loans continued, as the case may be, at the beginning of an Interest Period shall at any time during such Interest Period exceed (A) the aggregate principal amount of the Loans then outstanding, or (B) the sum of the then outstanding principal amount of the Revolving Loans then available to Borrower under Section 2 hereof. Borrower shall pay to Agent for Lender, upon demand by Agent (or Agent may, at its option, charge any loan account of Borrower) any amounts required to compensate Agent, the Reference Bank or any participant with Agent and/or Lender for any loss (including loss of anticipated profits), cost or expense incurred by such person, as a result of the conversion of Eurodollar Rate Loans to Base Rate Loans pursuant to any of the foregoing.

(d) $% \left(A^{\prime}\right) =0$. Interest shall be payable by Borrower to Agent for Lender monthly in arrears not

later than the first day of each calendar month and shall be calculated on the basis of a three hundred sixty (360) day year and actual days elapsed. The interest rate on Obligations shall increase or decrease by an amount equal to each increase or decrease in the Base Rate effective on the first day of the month after any change in such Base Rate is announced based on the Base Rate in effect on the last day of the month in which any such change occurs. In no event shall charges constituting interest payable by Borrower to Lender exceed the maximum amount or the rate permitted under any applicable law or regulation, and if any such part or provision of this Agreement is in contravention of any such law or regulation, such part or provision shall be deemed amended to conform thereto.

3.2. Closing Fee Borrower shall pay to Agent for Lender as a nonrefundable closing fee the amount of One Hundred Fifty Six Thousand Two Hundred Fifty Dollars (\$156,250), which shall be fully earned on and payable on and as of the date hereof.

3.3. Servicing Fee. Borrower shall pay to Agent, for Agent's own account, monthly a servicing fee in an amount equal to One Thousand Dollars (\$1,000) in respect of Agent's services for each month (or part thereof) while this Agreement remains in effect and for so long thereafter as any of the Obligations are outstanding, which fee shall be fully earned as of and payable in advance on the date hereof and on the first day of each month hereafter.

3.4. Unused Line Fee. Borrower shall pay to Agent for Lender monthly an unused line fee at a rate equal to one-quarter of one percent (0.25%) per annum calculated upon the amount by which \$13,000,000 exceeds the average daily principal balance of the outstanding Revolving Loans during the immediately preceding month (or part thereof) while this Agreement is in effect and for so long thereafter as any of the Obligations are outstanding, which fee shall be payable on the first day of each month in arrears.

3.5. Changes in Laws and Increased Costs of Loans.

(a) Notwithstanding anything to the contrary contained herein, all Eurodollar Rate Loans shall, upon notice by Agent to Borrower, convert to Base Rate Loans in the event that (i) any change in applicable law or regulation (or the interpretation or administration thereof) shall either (A) make it unlawful for Lender, Reference Bank or any participant to make or maintain Eurodollar Rate Loans or to comply with the terms hereof in connection with the Eurodollar Rate Loans, or (B) shall result in the increase in the costs to Lender, Reference Bank or any participant of making or maintaining any Eurodollar Rate Loans by an amount deemed by Lender to be material, or (C) reduce the amounts received or receivable by Lender in respect thereof, by an amount deemed by Lender to be material or (ii) the cost to Lender, Reference Bank or any participant of making or maintaining any Eurodollar Rate Loans shall otherwise increase by an amount deemed by Lender to be material. Borrower shall pay to Agent for Lender, upon demand by Agent (or Agent may, at its option, charge any loan account of Borrower) any amounts required to compensate Lender, the Reference Bank or any participant with Lender for any loss (including loss of anticipated profits), cost or expense incurred by such person as a result of the foregoing, including, without limitation, any such loss, cost or expense incurred by reason of the liquidation or reemployment of deposits or other funds acquired by such person to make or maintain the Eurodollar Rate Loans or any portion thereof. A certificate of Lender setting forth the basis for the determination of such amount necessary to compensate Lender as aforesaid shall be delivered to Borrower and shall be conclusive, absent manifest error.

(b) If any payments or prepayments in respect of the Eurodollar Rate Loans are received by Agent for Lender other than on the last day of the applicable Interest Period (whether pursuant to acceleration, upon maturity or otherwise), including any payments pursuant to the application of collections under Section 6.3 or any other payments made with the proceeds of Collateral, Borrower shall pay to Agent for Lender upon demand by Agent (or Agent may, at its option, charge any loan account of Borrower) any amounts required to compensate Lender, the Reference Bank or any participant with Lender for any additional loss (including loss of anticipated profits), cost or expense incurred by such person as a result of such prepayment or payment, including, without limitation, any loss, cost or expense incurred by reason of the liquidation or reemployment of deposits or other funds acquired by such person to make or maintain such Eurodollar Rate Loans or any portion thereof.

3.6. Adjustments to Interest Rate. At the request of Borrower at any time after October 18, 2002, and so long as (a) Borrower's average Excess Availability for the most recently ended fiscal quarter exceeds Two Million Dollars (\$2,000,000), (b) no Event of Default has occurred and is continuing or event which, with the passage of time or giving of notice, or both, constitutes an Event of Default has occurred and is continuing, and (c) Borrower's Fixed Charge Coverage Ratio is greater than 1.20 to 1, effective on the first day of calendar month following the determination the Applicable Margin shall equal one half of one percent (0.50%) and the Applicable Eurodollar Rate Loans then outstanding).

SECTION 4. CONDITIONS PRECEDENT.

4.1. Conditions Precedent to Initial Loans. Each of the following is a condition precedent to Agent and Lender entering into this Agreement:

(a) Agent shall have received evidence, in form and substance satisfactory to Agent, that Agent has valid perfected and first priority, and only security interests in and liens upon the Collateral and any other property which is intended to be security for the Obligations or the liability of any Obligor in respect thereof, subject only to the security interests and liens permitted herein or in the other Financing Agreements;

(b) All requisite corporate action and proceedings in connection with this Agreement and the other Financing Agreements shall be satisfactory in form and substance to Agent, and Agent shall have received all information and copies of all documents, including records of requisite corporate action and proceedings which Agent may have requested in connection therewith, such documents where requested by Agent or its counsel to be certified by appropriate corporate officers or governmental authorities;

(c) No material adverse change shall have occurred in the assets, business or prospects of Borrower since the date of Agent's latest field examination and no change or event shall have occurred which would impair the ability of Borrower or any Obligor to perform its obligations hereunder or under any of the other Financing Agreements to which it is a party or of Agent to enforce the Obligations or realize upon the Collateral;

 $\mbox{(d)}$ Agent shall have completed a field review of the Records and such other

information with respect to the Collateral as Agent may require to determine the amount of Revolving Loans available to Borrower, the results of which shall be satisfactory to Agent, not more than three (3) business days prior to the date hereof;

(e) Agent shall have received, in form and substance satisfactory to Agent, all consents, waivers, acknowledgments and other agreements from third persons which Agent may deem necessary or desirable in order to permit, protect and perfect its security interests in and liens upon the Collateral or to effectuate the provisions or purposes of this Agreement and the other Financing Agreements, including acknowledgments by lessors, mortgagees and warehousemen of Agent's security interests in the Collateral, waivers by such persons of any security interests, liens or other claims by such persons to the Collateral and agreements permitting Agent access to, and the right to remain on, the premises to exercise its rights and remedies and otherwise deal with the Collateral;

(f) Agent shall have received evidence of insurance and lender loss payee endorsements required hereunder and under the other Financing Agreements, in form and substance satisfactory to Agent, and certificates of insurance policies and/or endorsements naming Agent as lender loss payee;

(g) Agent shall have received, in form and substance satisfactory to Agent, such opinion letters of counsel to Borrower with respect to the Financing Agreements and such other matters as Agent may request;

(h) The Financing Agreements and all instruments and documents hereunder and thereunder shall have been duly executed and delivered to Agent, in form and substance satisfactory to Agent;

(i) Agent shall have received management letter from the independent outside public accounting firm of Borrower for the fiscal year ending October 31, 2001;

(j) The Excess Availability, as determined by Agent on and as of the date hereof, shall be not less than One Million Five Hundred Thousand Dollars (\$1,500,000) after giving effect to any Revolving Loans and Availability Reserves made on the date hereof.

4.2. Conditions Precedent to All Loans. Each of the following is an additional condition precedent to Agent and Lender entering into this Agreement and Lender making Loans to Borrower, including the initial Loans and any future Loans:

(a) All representations and warranties contained herein and in the other Financing Agreements shall be true and correct in all material respects with the same effect as though such representations and warranties had been made on and as of the date of the making of each such Loan and after giving effect thereto, except to the extent that any representation or warranty was made with reference to a specific date, in which case such representation and warranty shall have been true and correct in all material respects as of such date; and

(b) No Event of Default shall exist and no event or condition which, with passage of time or giving of notice, or both, would constitute an Event of Default, shall exist or have occurred and be continuing on and as of the date of the making of such Loan and after giving effect thereto.

4.3. Lien on Real Property. The Deed of Trust securing prompt payment and performance of all of the Obligations shall be executed by Borrower in favor of Agent and shall be duly recorded, at Borrower's expense, in each office where such recording is required to constitute a fully perfected lien on the Real Property covered thereby. Borrower shall deliver to Agent, at Borrower's expense, mortgagee title insurance policies issued by a title insurance company satisfactory to Agent, which policies shall be in form and substance satisfactory to Agent and shall insure a valid first lien in favor of Agent on the Real Property covered thereby, subject only to those exceptions acceptable to Agent and its counsel. Borrower shall also deliver to Agent such other documents and assurances, including, without limitation, existing survey prints and flood plain certificates of the Real Property, as Agent and its counsel may request relating to the Real Property subject to the Deed of Trust.

SECTION 5. GRANT OF SECURITY INTEREST.

To secure payment and performance of all Obligations, Borrower hereby grants to Agent, for the benefit of Lender, a continuing security interest in, a lien upon, and a right of set off against, and hereby assigns to Agent, for the benefit of Lender, as security, the following property and interests in property of Borrower, whether now owned or hereafter acquired or existing, and wherever located (collectively, the "Collateral"):

5.1. Accounts;

5.2. All present and future contract rights, general intangibles (including tax and duty refunds, registered and unregistered patents, trademarks, service marks, copyrights, trade names, applications for the foregoing, trade secrets, goodwill, processes, drawings, blueprints, customer lists, licenses, whether as licensor or licensee, choses in action, payment intangibles and other claims and existing and future leasehold interests in equipment, real estate and fixtures), chattel paper, electronic chattel paper, commercial tort claims, documents, instruments, securities and other investment property, letters of credit, bankers' acceptances and guaranties;

5.3. All present and future monies, securities, credit balances, deposits, deposit accounts, certificates of deposit and other property of Borrower now or hereafter held or received by or in transit to Agent or Lender or its affiliates or at any other depository or other institution from or for the account of Borrower, whether for safekeeping, pledge, custody, transmission, collection or otherwise, and all present and future liens, security interests, rights, remedies, title and interest in, to and in respect of Accounts and other Collateral, including, without limitation, (a) rights and remedies under or relating to guaranties, contracts of suretyship, letters of credit and credit and other insurance related to the Collateral, (b) rights of stoppage in transit, replevin, repossession, reclamation and other rights and remedies of an unpaid vendor, lienor or secured party, (c) goods described in invoices, documents, contracts or instruments with respect to, or otherwise representing or evidencing, Accounts or other Collateral, including returned, repossessed and reclaimed goods, and (d) deposits by and property of Account Debtors or other persons securing the obligations of Account Debtors;

5.4. Equipment;

5.5. Inventory;

- 5.6. Investment Property;
- 5.7. Records;
- 5.8. Commercial Tort Claims;
- 5.9. Letter of Credit Rights; and

5.10. All products and cash and non-cash proceeds of the foregoing, in any form, including, without limitation, insurance policies and proceeds (including, without limitation, business interruption insurance) and all claims against third parties for loss or damage to or destruction of any or all of the foregoing.

SECTION 6. COLLECTION AND ADMINISTRATION.

6.1. Borrower's Loan Account. Agent shall maintain one or more loan account(s) on its books in which shall be recorded (a) all Loans and other Obligations and the Collateral, (b) all payments made by or on behalf of Borrower and (c) all other appropriate debits and credits as provided in this Agreement, including fees, charges, costs, expenses and interest. All entries in the loan account(s) shall be made in accordance with Agent's customary practices as in effect from time to time.

6.2. Statements. Agent shall render to Borrower each month a statement setting forth the balance in Borrower's loan account(s) maintained by Agent for Borrower pursuant to the provisions of this Agreement, including principal, interest, fees, costs and expenses. Each such statement shall be subject to subsequent adjustment by Agent but shall, absent manifest errors or omissions, be considered correct and deemed accepted by Borrower and conclusively binding upon Borrower as an account stated except to the extent that Agent receives a written notice from Borrower of any specific exceptions of Borrower thereto within thirty (30) days after the date such statement has been mailed by Agent. Until such time as Agent shall have rendered to Borrower a written statement as provided above, the balance in Borrower's loan account(s) shall be presumptive evidence of the amounts due and owing to Agent and/or Lender by Borrower.

6.3. Collection of Accounts.

(a) Borrower shall establish and maintain, at its expense, blocked accounts or lockboxes and related blocked accounts (in either case, "Blocked Accounts"), as Agent may specify, with such banks as are acceptable to Agent into which Borrower shall promptly deposit and direct its Account debtors to directly remit all payments on Accounts and all payments constituting proceeds of Inventory or other Collateral in the identical form in which such payments are made, whether by cash, check or other manner. The banks at which the Blocked Accounts are established shall enter into an agreement, in form and substance satisfactory to Agent, providing that all items received or deposited in the Blocked Accounts are the property of Agent, that the depository bank has no lien upon, or right to setoff against, the Blocked Accounts, the items received for deposit therein, or the funds from time to time on deposit therein and that the depository bank will wire, or otherwise transfer, in immediately available funds, on a daily basis, all funds received or deposited into the Blocked Accounts to such bank account of Agent as Agent may from time to time designate for such purpose ("Payment Account"). Borrower agrees that all payments made to such Blocked Accounts or other funds received and collected by Agent, whether on the Accounts or as proceeds

of Inventory or other Collateral or otherwise shall be the property of Agent for so long as any of the Obligations remain outstanding and shall be held or applied in accordance with the terms of this Agreement.

(b) For purposes of calculating the amount of the Loans available to Borrower, such payments will be applied (conditional upon final collection) to the Obligations on the Business Day of receipt by Agent of immediately available funds in the Payment Account provided such payments and notice thereof are received in accordance with Agent's usual and customary practices as in effect from time to time and within sufficient time to credit Borrower's loan account on such day, and if not, then on the next Business Day. For the purposes of calculating interest on the Obligations, such payments or other funds received will be applied (conditional upon final collection) to the Obligations one (1) Business Day following the date of receipt of immediately available funds by Agent in the Payment Account provided such payments or other funds and notice thereof are received in accordance with Agent's usual and customary practices as in effect from time to time and within sufficient time to credit Borrower's loan account on such day, and if not, then on the next Business Day.

(c) Borrower and all of its affiliates, subsidiaries, shareholders, directors, employees or agents shall, acting as trustee for Agent, receive, as the property of Agent, any monies, checks, notes, drafts or any other payment relating to and/or proceeds of Accounts or other Collateral which come into their possession or under their control and immediately upon receipt thereof, shall deposit or cause the same to be deposited in the Blocked Accounts, or remit the same or cause the same to be remitted, in kind, to Agent. In no event shall the same be commingled with Borrower's own funds. Borrower agrees to reimburse Agent and Lender on demand for any amounts owed or paid to any bank at which a Blocked Account is established or any other bank or person involved in the transfer of funds to or from the Blocked Accounts arising out of Agent's payments to or indemnification of such bank or person. The obligation of Borrower to reimburse Agent and Lender for such amounts pursuant to the prior sentence of this Section 6.3 shall survive the termination or non-renewal of this Agreement.

(d) At Agent's request, Borrower shall cause that portion of the proceeds of Collateral representing sales and/or use taxes payable in connection with sales or otherwise, to be deposited into a separate bank account or accounts established for such purpose. In no event shall the same be commingled with Borrower's own funds. Borrower shall provide Agent on a monthly basis with evidence of collection and remittance of sales and/or use taxes.

6.4. Payments. All Obligations shall be payable to the Payment Account as provided in Section 6.3 or such other place as Agent may designate from time to time. Agent may apply payments received or collected from Borrower or for the account of Borrower (including, without limitation, the monetary proceeds of collections or of realization upon any Collateral) to such of the Obligations under this Agreement, whether or not then due, in such order and manner as Agent determines. At Agent's option, all principal, interest, fees, costs, expenses and other charges provided for in this Agreement or the other Financing Agreements may be charged directly to the loan account(s) of Borrower. Borrower shall make all payments to Agent on the Obligations free and clear of, and without deduction or withholding for or on account of, any setoff, counterclaim, defense, duties, taxes, levies, imposts, fees, deductions, withholding, restrictions or conditions of any kind. If after receipt of any payment of, or proceeds of Collateral applied to the payment of, any of the Obligations, Agent is required to surrender or return such payment or proceeds to any Person

for any reason, then the Obligations intended to be satisfied by such payment or proceeds shall be reinstated and continue and this Agreement shall continue in full force and effect as if such payment or proceeds had not been received by Agent. Borrower shall be liable to pay to Agent and Lender, and does hereby indemnify and hold Agent and Lender harmless for the amount of any payments or proceeds surrendered or returned. This Section 6.4 shall remain effective notwithstanding any contrary action which may be taken by Agent and/or Lender in reliance upon such payment or proceeds. This Section 6.4 shall survive the payment of the Obligations and the termination or non-renewal of this Agreement.

6.5. Authorization to Make Loans. Agent and Lender are authorized to make the Loans based upon telephonic or other instructions received from anyone purporting to be any of the authorized officers or representatives of Borrower identified on Schedule 6.5 hereto (as said Schedule 6.5 may be amended or revised from time to time) or, at the discretion of Agent, if such Loans are necessary to satisfy any Obligations. All requests for Loans hereunder shall specify the date on which the requested advance is to be made (which day shall be a Business Day) and the amount of the requested Loan. Requests received after 11:00 a.m. (Eastern Time) on any day shall be deemed to have been made as of the opening of business on the immediately following Business Day. All Loans under this Agreement shall be conclusively presumed to have been made to, and at the request of and for the benefit of, Borrower when deposited to the credit of Borrower or otherwise disbursed or established in accordance with the instructions of Borrower or in accordance with the terms and conditions of this Agreement.

6.6. Use of Proceeds. Borrower shall use the initial proceeds of the Loans provided by Lender to Borrower hereunder only for: (a) payments to each of the persons listed in the disbursement direction letter furnished by Borrower to Agent on or about the date hereof and (b) costs, expenses and fees in connection with the preparation, negotiation, execution and delivery of this Agreement and the other Financing Agreements. All other Loans made by Lender to Borrower pursuant to the provisions hereof shall be used by Borrower only for general operating, working capital and other proper corporate purposes of Borrower not otherwise prohibited by the terms hereof. Borrower may not use the proceeds of the Loans for Settlements without the prior written consent of Agent and Lender, in their exclusive and sole discretion. None of the proceeds will be used, directly or indirectly, for the purpose of purchasing or carrying any margin security or for the purposes of reducing or retiring any indebtedness which was originally incurred to purchase or carry any margin security or for any other purpose which might cause any of the Loans to be considered a "purpose credit" within the meaning of Regulation U of the Board of Governors of the Federal Reserve System, as amended.

SECTION 7. COLLATERAL REPORTING AND COVENANTS.

7.1 Collateral Reporting. Borrower shall provide Agent with the following documents in a form satisfactory to Agent: (a) on a regular basis as required by Agent, a schedule of Accounts, sales made, credits issued and cash received; (b) on a bi-weekly basis or more frequently as Agent may request, (i) perpetual inventory reports, (ii) inventory reports by category and (iii) agings of accounts payable; (c) upon Agent's request, (i) copies of customer statements and credit memos, remittance advices and reports, copies of deposit slips and bank statements, (ii) copies of shipping and delivery documents, and (iii) copies of purchase orders, invoices and delivery documents for Inventory and Equipment acquired by Borrower; (d) on a monthly basis or more frequently as Agent may request, an aging of accounts receivable, reports on sales and use tax collections, deposits

and payments, including monthly sales and use tax accruals; and (e) such other reports as to the Collateral as Agent shall request from time to time. If any of Borrower's records or reports of the Collateral are prepared or maintained by an accounting service, contractor, shipper or other agent, Borrower hereby irrevocably authorizes such service, contractor, shipper or agent at any time that an Event of Default exists or has occurred and is continuing to deliver such records, reports, and related documents to Agent and to follow Agent's instructions with respect to further services.

- 7.2. Accounts Covenants.
 - -----

(a) Borrower shall notify Agent promptly of: (i) any material delay in Borrower's performance of any of its obligations to any Account Debtor or the assertion of any claims, offsets, defenses or counterclaims by any Account Debtor, or any disputes with Account Debtors, or any settlement, adjustment or compromise thereof, (ii) all material adverse information relating to the financial condition of any Account Debtor and (iii) any event or circumstance which, to Borrower's knowledge would cause Agent to consider any then existing Accounts as no longer constituting Eligible Accounts. No credit, discount, allowance or extension or agreement for any of the foregoing shall be granted to any Account Debtor without Agent's consent, except in the ordinary course of Borrower's business in accordance with practices and policies previously disclosed in writing to Agent. So long as no Event of Default exists or has occurred and is continuing, Borrower shall settle, adjust or compromise any claim, offset, counterclaim or dispute with any Account Debtor. At any time that an Event of Default exists or has occurred and is continuing, Agent shall, at its option, have the exclusive right to settle, adjust or compromise any claim, offset, counterclaim or dispute with Account Debtors or grant any credits, discounts or allowances.

(b) Without limiting the obligation of Borrower to deliver any other information to Agent, Borrower shall promptly report to Agent any return of Inventory by any one Account Debtor if the inventory so returned in such case has a sales price in excess of Fifty Thousand Dollars (\$50,000). At any time that Inventory is returned, reclaimed or repossessed, the related Account shall not be deemed an Eligible Account. In the event any Account Debtor returns Inventory when an Event of Default exists or has occurred and is continuing, Borrower shall, upon Agent's request, (i) hold the returned Inventory in trust for Agent, (ii) segregate all returned Inventory from all of its other property, (iii) dispose of the returned Inventory solely according to Agent's instructions, and (iv) not issue any credits, discounts or allowances with respect thereto without Agent's prior written consent.

(c) With respect to each Account: (i) the amounts shown on any invoice delivered to Agent or Lender or schedule thereof delivered to Agent or Lender shall be true and complete, (ii) no payments shall be made thereon except payments immediately delivered to Agent or Lender pursuant to the terms of this Agreement, (iii) no credit, discount, allowance or extension or agreement for any of the foregoing shall be granted to any Account Debtor except as reported to Agent in accordance with this Agreement and except for credits, discounts, allowances or extensions made or given in the ordinary course of Borrower's business in accordance with practices and policies previously disclosed to Agent, (iv) there shall be no setoffs, deductions, contras, defenses, counterclaims or disputes existing or asserted with respect thereto except as reported to Agent in accordance with the terms of this Agreement, (v) none of the transactions giving rise thereto will violate any applicable State or Federal laws or regulations, all documentation relating thereto will be legally sufficient under such laws and regulations and all such documentation will be legally enforceable in accordance with its terms.

(d) Agent shall have the right at any time or times, in Agent's name or in the name of a nominee of Agent, to verify the validity, amount, offsets, chargebacks and claims and any other matter relating to any Account or other Collateral, by mail, telephone, facsimile transmission or otherwise.

(e) Borrower shall deliver or cause to be delivered to Agent, with appropriate endorsement and assignment, with full recourse to Borrower, all chattel paper and instruments which Borrower now owns or may at any time acquire immediately upon Borrower's receipt thereof, except as Agent may otherwise agree.

(f) Agent may, at any time or times that an Event of Default exists or has occurred and is continuing, (i) notify any or all Account Debtors that the Accounts have been assigned to Agent and that Agent has a security interest therein and Agent may direct any or all accounts debtors to make payment of Accounts directly to Agent, (ii) extend the time of payment of, compromise, settle or adjust for cash, credit, return of merchandise or otherwise, and upon any terms or conditions, any and all Accounts or other obligations included in the Collateral and thereby discharge or release the Account Debtor or any other party or parties in any way liable for payment thereof without affecting any of the Obligations, (iii) demand, collect or enforce payment of any Accounts or such other obligations, but without any duty to do so, and Agent shall not be liable for its failure to collect or enforce the payment thereof nor for the negligence of its agents or attorneys with respect thereto and (iv) take whatever other action Agent may deem necessary or desirable for the protection of its interests. At any time that an Event of Default exists or has occurred and is continuing, at Agent's request, all invoices and statements sent to any Account Debtor shall state that the Accounts and such other obligations have been assigned to Agent and are payable directly and only to Agent and Borrower shall deliver to Agent such originals of documents evidencing the sale and delivery of goods or the performance of services giving rise to any Accounts as Agent may require.

7.3. Inventory Covenants. With respect to the Inventory: (a) Borrower shall at all times maintain inventory records reasonably satisfactory to Agent, keeping correct and accurate records itemizing and describing the kind, type, quality and quantity of Inventory, Borrower's cost therefor and daily withdrawals and drawdowns therefrom and additions thereto; (b) Borrower shall conduct a physical count of the Inventory at least once each year, but at any time or times as Agent may request on or after an Event of Default, and promptly following such physical inventory shall supply Agent with a report in the form and with such specificity as may be reasonably satisfactory to Agent concerning such physical count; (c) Borrower shall not remove any Inventory from the locations set forth or permitted herein (other than Inventory received for testing or delivered as samples with a Value not to exceed \$100,000 in the aggregate per annum), without the prior written consent of Agent, except for sales of Inventory in the ordinary course of Borrower's business and except to move Inventory directly from one location set forth or permitted herein to another such location; (d) subject to limitation on expenses set forth in Section 9.15 hereof, Borrower shall, at its expense, at least once every six (6) months, but at any time or times as Agent may request on or after an Event of Default, deliver or cause to be delivered to Agent written reports or appraisals as to the Inventory in form, scope and methodology acceptable to Agent and by an appraiser acceptable to Agent, addressed to Agent or upon which Agent is expressly permitted to rely; (e) Borrower shall produce, use, store and maintain the Inventory with all reasonable care and caution and in accordance with applicable standards of any insurance and in conformity with applicable laws (including the requirements of the Federal Fair Labor Standards Act of 1938, as amended and all

rules, regulations and orders related thereto); (f) Borrower assumes all responsibility and liability arising from or relating to the production, use, sale or other disposition of the Inventory; (g) Borrower shall not sell Inventory to any customer on approval, or any other basis which entitles the customer to return or may obligate Borrower to repurchase such Inventory; (h) Borrower shall keep the Inventory in good and marketable condition; (i) Borrower shall not, without prior written notice to Agent, acquire or accept any inventory on consignment or approval except pursuant to processing or tolling agreements; and (j) Borrower shall not commingle Inventory with goods or materials of customers or other third parties.

7.4. Equipment and Real Property Covenants. With respect to the Equipment: (a) subject to limitation on expenses set forth in Section 9.15 hereof, upon Agent's request, Borrower shall, at its expense, at any time or times as Agent may request, deliver or cause to be delivered to Agent written reports or appraisals as to the Equipment and the Real Property in form, scope and methodology acceptable to Agent and by an appraiser acceptable to Agent; (b) Borrower shall keep the Equipment in good order, repair, running and marketable condition (ordinary wear and tear excepted); (c) Borrower shall use the Equipment with all reasonable care and caution and in accordance with applicable standards of any insurance and in conformity with all applicable laws; (d) the Equipment is and shall be used in Borrower's business and not for personal, family, household or farming use; (e) Borrower shall not remove any Equipment from the locations set forth or permitted herein, except to the extent necessary to have any Equipment repaired or maintained in the ordinary course of the business of Borrower or to move Equipment directly from one location set forth or permitted herein to another such location and except for the movement of motor vehicles used by or for the benefit of Borrower in the ordinary course of business; (f) the Equipment is now and shall remain personal property and Borrower shall not permit any of the Equipment to be or become a part of or affixed to real property; and (g) Borrower assumes all responsibility and liability arising from the use of the Equipment.

7.5. Power of Attorney. Borrower hereby irrevocably designates and appoints Agent (and all persons designated by Agent) as Borrower's true and lawful attorney-in-fact, and authorizes Agent, in Borrower's or Agent's name, to: (a) at any time an Event of Default exists or has occurred or event which, with passage of time or giving of notice, or both, would constitute an Event of Default exists or has occurred and is continuing (i) demand payment on Accounts or other proceeds of Inventory or other Collateral, (ii) enforce payment of Accounts by legal proceedings or otherwise, (iii) exercise all of Borrower's rights and remedies to collect any Account or other Collateral, (iv) sell or assign any Account upon such terms, for such amount and at such time or times as Agent deems advisable, (v) settle, adjust, compromise, extend or renew an Account, (vi) discharge and release any Account, (vii) prepare, file and sign Borrower's name on any proof of claim in bankruptcy or other similar document against an Account Debtor, (viii) notify the post office authorities to change the address for delivery of Borrower's mail to an address designated by Agent, and open and dispose of all mail addressed to Borrower, and (ix) do all acts and things which are necessary, in Agent's determination, to fulfill Borrower's obligations under this Agreement and the other Financing Agreements and (b) at any time to (i) take control in any manner of any item of payment or proceeds thereof, (ii) have access to any lockbox or postal box into which Borrower's mail is deposited, (iii) endorse Borrower's name upon any items of payment or proceeds thereof and deposit the same in Agent's account for application to the Obligations, (iv) endorse Borrower's name upon any chattel paper, document, instrument, invoice, or similar document or agreement relating to any Account or any goods pertaining thereto or any other Collateral, (v) sign Borrower's name on any verification of Accounts and notices thereof to Account Debtors and (vi) execute in

Borrower's name (if applicable law requires such signature) and otherwise file any UCC financing statements or amendments thereto. Borrower hereby releases Agent and its officers, employees and designees from any liabilities arising from any act or acts under this power of attorney and in furtherance thereof, whether of omission or commission, except as a result of Agent's own gross negligence or wilful misconduct as determined pursuant to a final non-appealable order of a court of competent jurisdiction.

7.6. Right to Cure. Agent may, at its option, (a) after the occurrence and continuance of an Event of Default, cure any default by Borrower under any agreement with a third party or pay or bond on appeal any judgment entered against Borrower or liens permitted pursuant to Section 9.8, (b) at any other time, except for liens permitted pursuant to Section 9.8, discharge taxes, liens, security interests or other encumbrances at any time levied on or existing with respect to the Collateral and (c) pay any amount, incur any expense or perform any act which, in Agent's judgment, is necessary or appropriate to preserve, protect, insure or maintain the Collateral and the rights of Agent and Lender with respect thereto. Agent may add any amounts so expended to the Obligations and charge Borrower's account therefor, such amounts to be repayable by Borrower on demand. Agent shall be under no obligation to effect such cure, payment or bonding and shall not, by doing so, be deemed to have assumed any obligation or liability of Borrower. Any payment made or other action taken by Agent under this Section shall be without prejudice to any right to assert an Event of Default hereunder and to proceed accordingly.

7.7. Access to Premises. From time to time as requested by Agent, at the cost and expense of Borrower, (a) Agent or its designee shall have complete access to all of Borrower's premises during normal business hours and after notice to Borrower, or at any time and without notice to Borrower if an Event of Default exists or has occurred and is continuing, for the purposes of inspecting, verifying and auditing the Collateral and all of Borrower's books and records, including the Records, and (b) Borrower shall promptly furnish to Agent such copies of such books and records or extracts therefrom as Agent may request, and (c) use during normal business hours such of Borrower's personnel, equipment, supplies and premises as may be reasonably necessary for the foregoing and if an Event of Default exists or has occurred and is continuing for the collection of Accounts and realization of other Collateral.

SECTION 8. REPRESENTATIONS AND WARRANTIES.

Borrower hereby represents and warrants to Agent and Lender the following (which shall survive the execution and delivery of this Agreement), the truth and accuracy of which are a continuing condition of the making of Loans by Agent to Borrower:

8.1. Corporate Existence, Power and Authority; Subsidiaries. Borrower is a corporation duly organized and in good standing under the laws of its state of incorporation and is duly qualified as a foreign corporation and in good standing in all states or other jurisdictions where the nature and extent of the business transacted by it or the ownership of assets makes such qualification necessary, except for those jurisdictions in which the failure to so qualify would not have a material adverse effect on Borrower's financial condition, results of operation or business or the rights of Agent in or to any of the Collateral. The execution, delivery and performance of this Agreement, the other Financing Agreements and the transactions contemplated hereunder and thereunder are all within Borrower's corporate powers, have been duly authorized and are not in contravention of law or the terms of Borrower's certificate of incorporation, by-laws, or other organizational documentation, or

any indenture, agreement or undertaking to which such Borrower is a party or by which Borrower or its property are bound. This Agreement and the other Financing Agreements constitute legal, valid and binding obligations of Borrower enforceable in accordance with their respective terms. Borrower does not have any non-Borrower subsidiaries other than Optical Cable Foreign Sales Corporation, a corporation organized under the laws of the U.S. Virgin Islands.

8.2. Financial Statements; No Material Adverse Change. All financial statements relating to Borrower which have been or may hereafter be delivered by Borrower to Agent have been prepared in accordance with GAAP and fairly present the financial condition and the results of operation of Borrower as at the dates and for the periods set forth therein. Except as disclosed in any interim financial statements furnished by Borrower to Agent prior to the date of this Agreement, there has been no material adverse change as of the date of this Agreement in the assets, liabilities, properties and condition, financial or otherwise, of Borrower, since the date of the most recent audited financial statements furnished by Borrower to Agent prior to the date of this Agreement.

8.3. Chief Executive Office; Collateral Locations. The chief executive office of Borrower and Borrower's Records concerning Accounts are located only at the address set forth below on the signature page hereof and its only other places of business, subject to the right of Borrower to establish new locations in accordance with Section 9.2 below. The Information Certificate correctly identifies any of such locations which are not owned by Borrower and sets forth the owners and/or operators thereof and the holders of any mortgages on such locations.

8.4. Priority of Liens; Title to Properties. The security interests and liens granted to Agent under this Agreement and the other Financing Agreements constitute valid and perfected first priority liens and security interests in and upon the Collateral subject only to the liens shown on Schedule 8.4 attached hereto and made part hereof and the other liens permitted under Section 9.8 hereof. Borrower has good and marketable title to all of its properties and assets subject to no liens, mortgages, pledges, security interests, encumbrances or charges of any kind, except those granted to Agent and such others as are specifically listed on Schedule 8.4 or permitted under Section 9.8 hereof.

8.5. Tax Returns. Borrower has filed, or caused to be filed, in a timely manner all tax returns, reports and declarations which are required to be filed by it (without requests for extension except as previously disclosed in writing to Agent) except where the failure to file the same could not be reasonably expected to have a Material Adverse Effect. All information in such tax returns, reports and declarations is complete and accurate in all material respects. Borrower has paid or caused to be paid all taxes due and payable or claimed due and payable in any assessment received by it, except taxes the validity of which are being contested in good faith by appropriate proceedings diligently pursued and available to Borrower and with respect to which adequate reserves have been set aside on its books except where the failure to pay the same could not be reasonably expected to have a Material Adverse Effect. Adequate provision has been made for the payment of all accrued and unpaid federal, state, county, local, foreign and other taxes whether or not yet due and payable and whether or not disputed.

8.6. Litigation. Except as set forth on the Information Certificate or Schedule 8.6 hereto, there is no present investigation by any governmental agency pending, or to the best of Borrower's knowledge threatened, against or affecting Borrower, its assets or business and there is no action, suit, proceeding or claim by any Person pending, or to the best of Borrower's knowledge

threatened, against Borrower or its assets or goodwill, or against or affecting any transactions contemplated by this Agreement, which if adversely determined against Borrower would result in a Material Adverse Effect.

8.7. Compliance with Other Agreements and Applicable Laws. Borrower is not in default in any material respect under, or in violation in any material respect of any of the terms of, any agreement, contract, instrument, lease or other commitment to which it is a party or by which it or any of its assets are bound and Borrower is in compliance in all material respects with all applicable provisions of laws, rules, regulations, licenses, permits, approvals and orders of any foreign, federal, state or local governmental authority.

8.8. Bank Accounts. All of the deposit accounts, investment accounts or other accounts in the name of or used by Borrower maintained at any bank or other financial institution are shown on Schedule 8.8 attached hereto and made part hereof, subject to the right of Borrower to establish new accounts in accordance with Section 9.13 below.

8.9. Accuracy and Completeness of Information. All information furnished by or on behalf of Borrower in writing to Agent in connection with this Agreement or any of the other Financing Agreements or any transaction contemplated hereby or thereby, is true and correct in all material respects on the date as of which such information is dated or certified and does not omit any material fact necessary in order to make such information not misleading. No event or circumstance has occurred which has had or could reasonably be expected to have a Material Adverse Effect which has not been fully and accurately disclosed to Agent in writing.

8.10. Environmental Compliance.

(a) Except as shown on Schedule 8.10 attached hereto and made a part hereof or in the Phase One environmental site assessment report prepared for Agent in connection with the Loans (the "Phase One Report"), Borrower has not generated, used, stored, treated, transported, manufactured, handled, produced or disposed of any Hazardous Materials, on or off its premises (whether or not owned by it) in any manner which at any time violates any applicable Environmental Law or any license, permit, certificate, approval or similar authorization thereunder and the operation of Borrower complies in all respects with all Environmental Laws and all licenses, permits, certificates, approvals and similar authorizations thereunder.

(b) Except as shown on Schedule 8.10 attached hereto and made a part hereof or the Phase One Report, there has been no investigation, proceeding, complaint, order, directive, claim, citation or notice by any governmental authority or any other person nor is any pending or to the best of Borrower's knowledge threatened, with respect to any non-compliance with or violation of the requirements of any Environmental Law by Borrower or the release, spill or discharge, threatened or actual, of any Hazardous Material or the generation, use, storage, treatment, transportation, manufacture, handling, production or disposal of any Hazardous Materials or any other environmental, health or safety matter, which affects Borrower or its business, operations or assets or any properties at which Borrower has transported, stored or disposed of any Hazardous Materials.

(c) Except as shown on Schedule 8.10 attached hereto and made a part hereof or the Phase One Report, Borrower has no liability, contingent or otherwise for violation of

Environmental Laws, in connection with a release, spill or discharge, threatened or actual, of any Hazardous Materials or the generation, use, storage, treatment, transportation, manufacture, handling, production or disposal of any Hazardous Materials.

(d) Borrower has all licenses, permits, certificates, approvals or similar authorizations required to be obtained or filed in connection with the operations of Borrower under any Environmental Law and all of such licenses, permits, certificates, approvals or similar authorizations are valid and in full force and effect.

(e) Borrower has no actual knowledge of:

(i) the presence of any Hazardous Materials on any of the Real Property or on which the Collateral is located in violation of Environmental Laws, or

 $({\rm ii})~$ any on-site spills, releases, discharges, disposal or storage of Hazardous Materials that have occurred or are presently occurring on any of such real property, or

(iii) any spills, releases, discharges or disposal of Hazardous Materials in violation of Environmental Laws that have occurred or are presently occurring on such real properties as a result of the activities of Borrower, or

(iv) except as shown on Schedule 8.10 attached hereto and made a part hereof, any notice, summons, citation or other communication sent to Borrower from any state or federal agency concerning any intentional or unintentional action or conduct, inaction or omission, past or present which is or may be in violation of Environmental Laws.

8.11. Solvency. Borrower is able to pay its debts as they become due, has sufficient capital to carry on its business operations, and presently owns property having a fair salable value which is greater than the amount required to pay all of Borrower's debts as they become due, after giving effect to the application of the initial Loan proceeds as provided in Section 6.6 hereof.

8.12. Employee Benefits.

(a) Borrower has not engaged in any transaction in connection with which Borrower or any of its ERISA Affiliates could be subject to either a civil penalty assessed pursuant to Section 502(i) of ERISA or a tax imposed by Section 4975 of the Code, including any accumulated funding deficiency described in Section 8.12(c) hereof and any deficiency with respect to vested accrued benefits described in Section 8.12(d) hereof.

(b) No liability to the Pension Benefit Guaranty Corporation (other than routine premium payments due from time to time in accordance with 4007(a) of ERISA) has been or is expected by Borrower to be incurred with respect to any employee benefit plan of Borrower or any of its ERISA Affiliates. There has been no reportable event (within the meaning of Section 4043(c) of ERISA) or any other event or condition with respect to any employee pension benefit plan of Borrower or any of its ERISA Affiliates a risk of termination of any such plan by the Pension Benefit Guaranty Corporation.

(c) Full payment has been made of all amounts which Borrower or any of their ERISA Affiliates is required under Section 302 of ERISA and Section 412 of the Code to have paid

under the terms of each employee benefit plan as contributions to such plan as of the last day of the most recent fiscal year of such plan ended prior to the date hereof, and no accumulated funding deficiency (as defined in Section 302 of ERISA and Section 412 of the Code), whether or not waived, exists with respect to any employee benefit plan, including any penalty or tax described in Section 8.12(a) hereof and any deficiency with respect to vested accrued benefits described in Section 8.12(d) hereof.

(d) The current value of all vested accrued benefits under all employee benefit plans maintained by Borrower that are subject to Title IV of ERISA does not exceed the current value of the assets of such plans allocable to such vested accrued benefits, including any penalty or tax described in Section 8.12(a) hereof and any accumulated funding deficiency described in Section 8.12(c) hereof. The terms "current value" and "accrued benefit" have the meanings specified in ERISA.

(e) Neither Borrower nor any of its ERISA Affiliates is or has ever been obligated to contribute to any "multiemployer plan" (as such term is defined in Section 4001(a) (3) of ERISA) that is subject to Title IV of ERISA.

8.13. Commercial Tort Claims. Borrower has no Commercial Tort Claims in which Borrower is the claimant except as shown on Schedule 8.13 attached hereto and made a part hereof or as disclosed in writing to Agent from time to time.

8.14. Letter of Credit Rights. Borrower has no Letter of Credit Rights except as shown on Schedule 8.14 attached hereto and made a part hereof or as disclosed in writing to Agent from time to time.

8.15. Survival of Warranties; Cumulative. All representations and warranties contained in this Agreement or any of the other Financing Agreements shall survive the execution and delivery of this Agreement and shall be deemed to have been made again to Agent and Lender on the date of each additional borrowing or other credit accommodation hereunder, except to the extent that any representation or warranty was made with reference to a specific date, in which case such representation and warranty shall have been true and correct in all material respects as of such date, and shall be conclusively presumed to have been relied on by Agent and Lender regardless of any investigation made or information possessed by Agent and/or Lender. The representations and warranties set forth herein shall be cumulative and in addition to any other representations or warranties which Borrower shall now or hereafter give, or cause to be given, to Agent and Lender.

SECTION 9. AFFIRMATIVE AND NEGATIVE COVENANTS.

9.1. Maintenance of Existence. Borrower shall at all times preserve, renew and keep in full, force and effect its corporate existence and rights and franchises with respect thereto and maintain in full force and effect all permits, licenses, trademarks, tradenames, approvals, authorizations, leases and contracts necessary to carry on the business as presently or proposed to be conducted. Borrower shall give Agent thirty (30) days prior written notice of any proposed change in its corporate name, which notice shall set forth the new name and Borrower shall deliver to Agent a copy of the amendment to the Certificate of Incorporation of Borrower providing for the name change certified by the Secretary of State of the jurisdiction of incorporation of Borrower as soon as it is available.

9.2. New Collateral Locations. Borrower shall not directly or indirectly open any new location or place, or store or warehouse Inventory, in any new location, or enter into any lease for any new location, within the continental United States or otherwise unless Borrower shall give Agent thirty (30) days prior written notice of any such new location and first executes and delivers, or causes to be executed and delivered, to Agent such agreements, documents, and instruments, as Agent may deem reasonably necessary or desirable to protect its interests in the Collateral at such location, including UCC financing statements.

9.3. Compliance with Laws, Regulations, etc.

(a) Borrower shall, at all times, comply in all respects with all laws, rules, regulations, licenses, permits, approvals and orders applicable to it and duly observe all requirements of any federal, state or local governmental authority, including ERISA, the Occupational Safety and Health Act of 1970, as amended, the Fair Labor Standards Act of 1938, as amended, and all statutes, rules, regulations, orders, permits and stipulations relating to environmental pollution and employee health and safety, including all Environmental Laws.

(b) Borrower shall monitor its continued compliance with all Environmental Laws in all of its operations. Copies of all environmental surveys, audits, assessments, feasibility studies and results of remedial investigations shall be promptly furnished, or caused to be furnished, by Borrower to Agent. Borrower shall take prompt and appropriate action to respond to any non-compliance with any of the Environmental Laws and shall regularly report to Agent on such response.

(c) Borrower shall give written notice to Agent promptly upon Borrower's receipt of any notice of, or Borrower's otherwise obtaining knowledge of, (i) the occurrence of any event involving the release, spill or discharge, threatened or actual, of any Hazardous Material relating to Borrower, the Collateral or Borrower's premises in violation of any Environmental Laws or (ii) any investigation, proceeding, complaint, order, directive, claims, citation or notice with respect to any non-compliance with or violation of any Environmental Law by Borrower, or any other environmental, health or safety matter, which affects Borrower or its business and operations.

(d) Without limiting the generality of the foregoing, whenever Agent reasonably determines that there is non-compliance, or any condition which requires any action by or on behalf of Borrower in order to avoid any non-compliance, with any Environmental Law, Borrower shall, at Agent's request and Borrower's expense: (i) cause an independent environmental engineer acceptable to Agent to conduct such tests of the site where Borrower's non-compliance or alleged non-compliance with such Environmental Laws has occurred as to such non-compliance and prepare and deliver to Agent a report as to such non-compliance setting forth the results of such tests, a proposed plan for responding to any environmental problems described therein, and an estimate of the costs thereof and (ii) provide to Agent a supplemental report of such engineer whenever the scope of such non-compliance, or Borrower's response thereto or the estimated costs thereof, shall change in any respect.

(e) Borrower shall indemnify and hold harmless Agent and Lender, their directors, officers, employees, agents, invitees, representatives, successors and assigns, from and against any and all losses, claims, damages, liabilities, costs, and expenses (including attorneys' fees and legal expenses) directly or indirectly arising out of or attributable to the use, generation, manufacture,

reproduction, storage, release, threatened release, spill, discharge, disposal or presence of a Hazardous Material, including the costs of any required or necessary repair, cleanup or other remedial work with respect to any property of Borrower and the preparation and implementation of any closure, remedial or other required plans. All representations, warranties, covenants and indemnifications in this Section 9.3 shall survive the payment of the Obligations and the termination or non-renewal of this Agreement.

9.4. Payment of Taxes and Claims. Borrower shall duly pay and discharge all taxes, assessments, contributions and governmental charges upon or against it or its properties or assets, except for taxes the validity of which are being contested in good faith by appropriate proceedings diligently pursued and available to Borrower and with respect to which adequate reserves have been set aside on its books. Borrower shall be liable for any tax or penalties imposed on Agent and/or Lender as a result of the financing arrangements provided for herein and Borrower agrees to indemnify and hold Agent and Lender harmless with respect to the foregoing, and to repay to Agent and Lender on demand the amount thereof, and until paid by Borrower such amount shall be added and deemed part of the Loans, provided, that, nothing contained herein shall be construed to require Borrower to pay any income or franchise taxes attributable to the income of Agent and Lender from any amounts charged or paid hereunder to Agent and Lender. The foregoing indemnity shall survive the payment of the Obligations and the termination or non-renewal of this Agreement.

9.5. Insurance. Borrower shall, at all times, maintain with financially sound and reputable insurers insurance with respect to the Collateral against loss or damage and all other insurance of the kinds and in the amounts customarily insured against or carried by corporations of established reputation engaged in the same or similar businesses and similarly situated. Said policies of insurance shall be satisfactory to Agent as to form, amount and insurer. Borrower shall furnish certificates, policies or endorsements to Agent as Agent shall require as proof of such insurance, and, if Borrower fails to do so, Agent is authorized, but not required, to obtain such insurance at the expense of Borrower. All policies shall provide for at least thirty (30) days prior written notice to Agent of any cancellation or reduction of coverage and that Agent may act as attorney for Borrower in obtaining, and at any time an Event of Default exists or has occurred and is continuing, adjusting, settling, amending and canceling such insurance. Borrower shall cause Agent and Lender to be named as a lender loss pavee and an additional insured (but without any liability for any premiums) under such insurance policies and Borrower shall obtain non-contributory lender's loss payable endorsements to all insurance policies in form and substance satisfactory to Agent. Such lender's loss payable endorsements shall specify that the proceeds of such insurance shall be payable to Agent as its interests may appear and further specify that Agent shall be paid regardless of any act or omission by Borrower or any of its affiliates. At its option, Agent may apply any insurance proceeds received by Agent at any time to the cost of repairs or replacement of Collateral and/or to payment of the Obligations, whether or not then due, in any order and in such manner as Agent may determine or hold such proceeds as cash collateral for the Obligations.

9.6. Financial Statements and Other Information.

(a) Borrower shall keep proper books and records in which true and complete entries shall be made of all dealings or transactions of or in relation to the Collateral and the business of Borrower and its subsidiaries (if any) in accordance with GAAP and Borrower shall furnish or cause to be furnished to Agent: (i) within thirty (30) days after the end of each fiscal month, monthly unaudited consolidated financial statements, (including balance sheets, statements

of income and loss, statements of cash flow, and statements of shareholders' equity), all in reasonable detail, fairly presenting the financial position and the results of the operations of Borrower and its subsidiaries as of the end of and through such fiscal month subject to such revisions as may be reflected on the quarterly unaudited consolidated financial statements filed by Borrower with the Securities and Exchange Commission; and (ii) within ninety (90) days after the end of each fiscal year, audited consolidated financial statements (including balance sheets, statements of income and loss, statements of cash flow and statements of shareholders' equity), and the accompanying notes thereto, all in reasonable detail, fairly presenting the financial position and the results of the operations of Borrower and its subsidiaries as of the end of and for such fiscal year, together with the unqualified opinion of independent certified public accountants, which accountants shall be an independent accounting firm selected by Borrower and reasonably acceptable to Agent, that such financial statements have been prepared in accordance with GAAP, and present fairly the results of operations and financial condition of Borrower and its subsidiaries as of the end of and for the fiscal year then ended.

(b) Borrower shall promptly notify Agent in writing of the details of (i) any loss, damage, investigation, action, suit, proceeding or claim relating to the Collateral or any other property which is security for the Obligations or which would result in any Material Adverse Effect and (ii) the occurrence of any Event of Default or event which, with the passage of time or giving of notice, or both, would constitute an Event of Default.

(c) Borrower shall promptly after the sending or filing thereof furnish or cause to be furnished to Agent copies of all reports which Borrower sends to its stockholders generally and copies of all reports and registration statements which Borrower files with the Securities and Exchange Commission, any national securities exchange or the National Association of Securities Dealers, Inc.

(d) Borrower shall furnish or cause to be furnished to Agent such budgets, forecasts, projections and other information respecting the Collateral and the business of Borrower, as Agent may, from time to time, reasonably requests. Agent is hereby authorized to deliver a copy of any financial statement or any other information relating to the business of Borrower to any court or other government agency if required by law or order or to any participant or assignee or prospective participant or assignee. Borrower hereby irrevocably authorizes and directs all accountants or auditors to deliver to Agent, at Borrower's expense, copies of the financial statements of Borrower and any reports or management letters prepared by such accountants or auditors on behalf of Borrower and to disclose to Agent such information as they may have regarding the business of Borrower. Any documents, schedules, invoices or other papers delivered to Agent may be destroyed or otherwise disposed of by Agent one (1) year after the same are delivered to Agent, except as otherwise designated by Borrower to Agent in writing.

(e) Along with the set of financial statements delivered to Agent pursuant to Section 9.6(a) (ii) hereof, Borrower shall deliver to Agent a certificate from its chief financial officer or treasurer setting forth that the signer has reviewed the relevant terms of this Agreement, and has made (or caused to be made under his supervision) a review of the transactions and condition of Borrower from the beginning of the fiscal year covered by the statement being delivered therewith to the date of the certificate, and that such review has not disclosed the existence during such period of any condition or event which constitutes an Event of Default or if any such condition or event existed or exists, specifying the nature and period of existence thereof and what action Borrower has

taken or proposes to take with respect thereto.

(f) Agent acknowledges its execution of that certain Non-Disclosure and Confidentiality Agreement dated December 5, 2001 and Lender acknowledges that Agent has delivered a copy of same to Lender, which agrees to be bound as if it were a signatory thereto.

9.7. Sale of Assets, Consolidation, Merger, Dissolution, Etc. Borrower shall not, directly or indirectly, (a) merge into or with or consolidate with any other Person or permit any other Person to merge into or with or consolidate with it, or (b) sell, assign, lease, transfer, abandon or otherwise dispose of any indebtedness to any other Person or any of its assets to any other Person (except for (i) sales of Inventory in the ordinary course of business, (ii) dispositions expressly permitted by Agent's written consent, and (iii) the disposition of worn-out or obsolete Equipment or Equipment no longer used in the business of Borrower so long as (A) if an Event of Default exists or has occurred and is continuing, any proceeds are paid to Agent and (B) such sales do not involve Equipment having an aggregate fair market value in excess of Fifty Thousand Dollars (\$50,000) for all such Equipment disposed of in any fiscal year of Borrower), or (c) form or acquire any subsidiaries, or (d) wind up, liquidate or dissolve or (e) agree to do any of the foregoing. Notwithstanding the foregoing restrictions, Borrower may issue capital stock in the aggregate par or stated value not to exceed twenty percent (20%) of the market capitalization of Borrower as of the close of business on the closing date of this Agreement.

9.8. Encumbrances. Borrower shall not create, incur, assume or suffer to exist any security interest, mortgage, pledge, lien, charge or other encumbrance of any nature whatsoever on any of its assets or properties, including the Collateral, except: (a) the liens and security interests of Agent; (b) liens securing the payment of taxes, either not yet overdue or the validity of which are being contested in good faith by appropriate proceedings diligently pursued and available to Borrower and with respect to which adequate reserves have been set aside on its books; (c) non-consensual statutory liens (other than liens securing the payment of taxes) arising in the ordinary course of Borrower's business to the extent: (i) such liens secure indebtedness which is not overdue or (ii) such liens secure indebtedness relating to claims or liabilities which are fully insured and being defended at the sole cost and expense and at the sole risk of the insurer or being contested in good faith by appropriate proceedings diligently pursued and available to Borrower, in each case prior to the commencement of foreclosure or other similar proceedings and with respect to which adequate reserves have been set aside on its books; (d) zoning restrictions, easements, licenses, covenants and other restrictions affecting the use of real property which do not interfere in any material respect with the use of such real property or ordinary conduct of the business of Borrower as presently conducted thereon or materially impair the value of the real property which may be subject thereto; (e) purchase money liens and security interests as permitted in Section 9.9(c) below; and (f) the security interests and liens set forth on Schedule 8.4 hereto.

9.9. Indebtedness. Borrower shall not incur, create, assume, become or be liable in any manner with respect to, or permit to exist, any obligations or indebtedness, except:

(a) The Obligations;

(b) Trade obligations and normal accruals in the ordinary course of business not yet due and payable, or with respect to which Borrower is contesting in good faith the amount or validity thereof by appropriate proceedings diligently pursued and available to Borrower and with

respect to which adequate reserves have been set aside on its books;

(c) Purchase money indebtedness (including capital leases) hereafter incurred by Borrower to finance fixed assets provided that (A) such indebtedness when incurred shall not exceed the purchase price of the asset(s) financed or the lesser of the purchase price or the value of the asset(s) if subsequently financed; and (B) no such indebtedness shall be refinanced for a principal amount in excess of the principal balance outstanding thereon at the time of such refinancing;

(d) Obligations or indebtedness set forth in Schedule 9.9 hereto; provided, that: (i) Borrower may only make regularly scheduled payments of principal and interest in respect of such indebtedness in accordance with the terms of the agreement or instrument evidencing or giving rise to such indebtedness as in effect on the date of the execution thereof, (ii) Borrower shall not, directly or indirectly, (A) make any prepayments or other non-mandatory payments in respect of such indebtedness, or (B) amend, modify, alter or change the terms of such indebtedness or any agreement, document or instrument related thereto, or (C) redeem, retire, defease, purchase or otherwise acquire such indebtedness, or set aside or otherwise deposit or invest any sums for such purpose, and (iii) Borrower shall furnish to Agent all notices, demands or other materials in connection with such indebtedness either received by Borrower or on their behalf, promptly after the receipt thereof, or sent by Borrower or on their behalf, concurrently with the sending thereof, as the case may be; and

(e) Unsecured indebtedness at any time not exceeding Two Hundred Thousand Dollars (200,000) in the aggregate.

9.10. Loans, Investments, Guarantees, Etc. Borrower shall not, directly or indirectly, make any loans or advance money or property to any Person, or invest in (by capital contribution, dividend or otherwise) or purchase or repurchase the stock or indebtedness or all or a substantial part of the assets or property of any Person, or guarantee, assume, endorse, or otherwise become responsible for (directly or indirectly) the indebtedness, performance, obligations or dividends of any Person or agree to do any of the foregoing, except: (a) the endorsement of instruments for collection or deposit in the ordinary course of business; (b) investments in: short-term direct obligations of the United States Government, negotiable certificates of deposit issued by any bank satisfactory to Agent, payable to the order of Borrower or to bearer and delivered to Agent, and commercial paper rated A1 or P1; provided, that, as to any of the foregoing, unless waived in writing by Agent, Borrower shall take such actions as are deemed necessary by Agent to perfect the security interest of Agent in such investments; and (c) the loans, advances and guarantees set forth on Schedule 9.10 hereto; provided, that, as to such loans, advances and guarantees set forth on Schedule 9.10, Borrower shall not, directly or indirectly, amend, modify, alter or change the terms of such loans, advances or guarantees or any agreement, document or instrument related thereto, or as to such guarantees, redeem, retire, defease, purchase or otherwise acquire the obligations arising pursuant to such guarantees, or set aside or otherwise deposit or invest any sums for such purpose, and Borrower shall furnish to Agent all notices or demands in connection with such loans, advances or guarantees or other indebtedness subject to such guarantees either received by Borrower or on its behalf, promptly after the receipt thereof, or sent by Borrower or on its behalf, concurrently with the sending thereof, as the case may be.

9.11. Dividends and Redemptions. Except as set forth on Schedule 9.11 hereto, Borrower shall not, directly or indirectly, declare or pay any cash dividends on account of any shares

of class of capital stock of Borrower now or hereafter outstanding, or set aside or otherwise deposit or invest any sums for such purpose, or redeem, retire, defease, purchase or otherwise acquire any shares of any class of capital stock (or set aside or otherwise deposit or invest any sums for such purpose) for any consideration other than common stock or apply or set apart any sum, or make any other distribution (by reduction of capital or otherwise) in respect of any such shares or agree to do any of the foregoing, but nothing herein contained shall prevent Borrower from declaring and completing stock splits or reverse stock splits.

9.12. Transactions with Affiliates. Borrower shall not, directly or indirectly, purchase, acquire or lease any property from, or sell, transfer or lease any property to, any officer, director, agent or other person affiliated with Borrower, except in the ordinary course of and pursuant to the reasonable requirements of Borrower's business and upon fair and reasonable terms no less favorable to Borrower than Borrower would obtain in a comparable arm's length transaction with an unaffiliated person or make any payments of management, consulting or other fees for management or similar services, or of any indebtedness owing to any officer, employee, shareholder, director or other person affiliated with Borrower except reasonable compensation to officers, employees and directors for services rendered to Borrower in the ordinary course of business.

9.13. Additional Bank Accounts. Borrower shall not, directly or indirectly, open, establish or maintain any deposit account, investment account or any other account with any bank or other financial institution, other than the Blocked Accounts and the accounts set forth in Schedule 8.8 except: as to any new or additional Blocked Accounts and other such new or additional accounts which contain any Collateral or proceeds thereof, with the prior written consent of Agent and subject to such conditions thereto as Agent may establish and as to any accounts used by Borrower to make payments of payroll, taxes or other obligations to third parties, after prior written notice to Agent.

9.14. Adjusted Net Worth. Borrower shall, at all times, maintain Adjusted Net Worth of not less than Twenty Five Million Dollars (\$25,000,000).

9.15. Costs and Expenses. Borrower shall pay to Agent and Lender on demand all costs, expenses, filing fees and taxes paid or payable in connection with the preparation, negotiation, execution, delivery, recording, administration, collection, liquidation, enforcement and defense of the Obligations, Agent's rights in the Collateral, this Agreement, the other Financing Agreements and all other documents related hereto or thereto, including any amendments, supplements or consents which may hereafter be contemplated (whether or not executed) or entered into in respect hereof and thereof, including, but not limited to: (a) all costs and expenses of filing or recording (including Uniform Commercial Code financing statement filing taxes and fees, documentary taxes, intangibles taxes and mortgage recording taxes and fees, if applicable); (b) all title and other insurance premiums, appraisal fees and search fees; (c) costs and expenses of remitting loan proceeds, collecting checks and other items of payment, and establishing and maintaining the Blocked Accounts, together with Agent's and Lender's customary charges and fees with respect thereto; (d) costs and expenses of preserving and protecting the Collateral; (e) costs and expenses paid or incurred in connection with obtaining payment of the Obligations, enforcing the security interests and liens of Agent, selling or otherwise realizing upon the Collateral, and otherwise enforcing the provisions of this Agreement and the other Financing Agreements or defending any claims made or threatened against Agent and/or Lender arising out of the transactions contemplated hereby and thereby (including, without limitation, preparations for and consultations concerning any

such matters); (f) all out-of-pocket expenses and costs heretofore and from time to time hereafter incurred by Agent and/or Lender during the course of periodic field examinations of the Collateral and Borrower's operations, plus a per diem charge at the rate of Seven Hundred Fifty Dollars (\$750) per person per day for Agent's examiners in the field and office; and (q) the fees and disbursements of counsel (including legal assistants) to Agent and/or Lender in connection with any of the foregoing. Notwithstanding the foregoing or anything in this Agreement to the contrary, so long as no Event of Default has occurred or event which, with the passage of time or giving of notice, or both, would constitute an Event of Default has occurred and is continuing, Borrower shall be responsible for the costs associated with no more than one (1) Equipment appraisal during any six (6) month period, one (1) Inventory appraisal during any six (6) month period and one (1) Real Estate appraisal during any twelve (12) month period plus at Agent's reasonable request if Borrower has requested that the MERE Sublimit be reset pursuant to Section 2.3, a written statement from the appraiser of the Real Property that it is not aware of any circumstances which would materially and adversely affect its most recent appraisal. In addition, so long as no Event of Default has occurred and no event which, with the passage of time or giving of notice, or both, would constitute an Event of Default has occurred and is continuing, Borrower shall not be responsible for the costs associated with an Inventory appraisal if the average amount calculated under Section 2.1(a) (ii) of this Agreement during the preceding six month period does not exceed One Million Five Hundred Thousand Dollars (\$1,500,000).

9.16. Compliance with ERISA.

Borrower shall not with respect to any "employee benefit (a) plans" maintained by Borrower or any of its ERISA Affiliates: (i) terminate any of such employee benefit plans so as to incur any liability to the Pension Benefit Guaranty Corporation established pursuant to ERISA, (ii) allow or suffer to exist any prohibited transaction involving any of such employee benefit plans or any trust created thereunder which would subject Borrower or such ERISA Affiliate to a tax or penalty or other liability on prohibited transactions imposed under Section 4975 of the Code or under ERISA, (iii) fail to pay to any such employee benefit plan any contribution which it is obligated to pay under Section 302 of ERISA, Section 412 of the Code or the terms of such plan, (iv) allow or suffer to exist any accumulated funding deficiency, whether or not waived, with respect to any such employee benefit plan, (v) allow or suffer to exist any occurrence of a reportable event or any other event or condition which presents a material risk of termination by the Pension Benefit Guaranty Corporation of any such employee benefit plan that is a single employer plan, which termination could result in any liability to the Pension Benefit Guaranty Corporation or (vi) incur any withdrawal liability with respect to any multiemployer pension plan.

(b) As used in this Section 9.16, the terms "employee benefit plans", "accumulated funding deficiency" and "reportable event" shall have the respective meanings assigned to them in ERISA, and the term "prohibited transaction" shall have the meaning assigned to it in Section 4975 of the Code and in ERISA.

9.17. Further Assurances. At the request of Agent at any time and from time to time, Borrower shall, at its expense, duly execute and deliver, or cause to be duly executed and delivered, such further agreements, documents and instruments, and do or cause to be done such further acts as may be necessary or proper to evidence, perfect, maintain and enforce the security interests and the priority thereof in the Collateral and to otherwise effectuate the provisions or purposes of this Agreement or any of the other Financing Agreements. Agent may at any time and from time to

time request a certificate from an officer of Borrower representing that all conditions precedent to the making of Loans contained herein are satisfied. In the event of such request by Agent, Agent and Lender may, at Agent's option, cease to make any further Loans until Agent has received such certificate and, in addition, Agent has determined that such conditions are satisfied. Borrower hereby authorizes Agent to file one or more UCC financing statements.

9.18. Commercial Tort Claims. Borrower shall provide written notice to Agent of any Commercial Tort Claim to which Borrower is or becomes a party or which otherwise inures to the benefit of Borrower. Such notice shall contain a sufficient description of the Commercial Tort Claim including the parties, the court in which the claim was commenced (if applicable), the docket number assigned to the case (if applicable) and a detailed explanation of the events giving rise to such claim. Borrower shall grant Agent a security interest in such Commercial Tort Claim to secure payment of the Obligations. Borrower shall execute and deliver such instruments, documents and agreements as Agent may reasonably require in order to obtain and perfect such security interest including, without limitation, a security agreement or amendment to this Agreement all in form and substance satisfactory to Agent. Borrower authorizes Agent to file (without Borrower's signature), financing statements or amendments to existing financing statements as Agent deems necessary to perfect the security interest.

9.19. Letter of Credit Rights. Borrower shall provide Agent with written notice of any Letters of Credit for which Borrower is the beneficiary. Borrower shall execute and deliver such instruments, documents and agreements and take such actions as Agent reasonably may require in order to obtain and perfect its security interest in such Letter of Credit Rights.

SECTION 10. EVENTS OF DEFAULT AND REMEDIES.

10.1. Events of Default. The occurrence or existence of any one or more of the following events are referred to herein individually as an "Event of Default", and collectively as "Events of Default":

(a) Borrower fails to pay when due any of the Obligations, or fails to perform any of the terms, covenants, conditions or provisions contained in this Agreement or any of the other Financing Agreements within ten (10) days after the earlier of any of the executive management of Borrower or any officer listed on Schedule 6.5 hereto becoming aware of such failure or Agent has given Borrower notice of such event (except that these shall be no requirement of notice and there shall be no cure period with respect to the provisions of Sections 9.1, 9.4, 9.6, 9.7, 9.8, 9.9, 9.10, 9.11, 9.12, and 9.14);

(b) Any representation, warranty or statement of fact made by Borrower to Agent and/or Lender in this Agreement, the other Financing Agreements or any other agreement, schedule, confirmatory assignment or otherwise shall when made or deemed made be false or misleading in any material respect;

(c) Any Obligor revokes, terminates or fails to perform any of the terms, covenants, conditions or provisions of any guarantee, endorsement or other agreement of such party in favor of Agent and/or Lender;

 $(d) \qquad \text{Any judgment for the payment of money is rendered against} \\ \text{Borrower or any} \\$

Obligor in excess of One Hundred Thousand Dollars (\$100,000) in any one case or in excess of Two Hundred Thousand Dollars (\$200,000) in the aggregate and shall remain undischarged or unvacated for a period in excess of thirty (30) days or execution shall at any time not be effectively stayed, or any judgment other than for the payment of money, or injunction, attachment, garnishment or execution is rendered against Borrower or any Obligor or any of their assets;

(e) Any Obligor (being a natural person or a general partner of an Obligor which is a partnership) dies or Borrower, or any Obligor which is a partnership, limited liability company, limited liability partnership or a corporation, dissolves or suspends or discontinues doing business;

(f) Borrower or any Obligor becomes insolvent (however defined or evidenced), makes an assignment for the benefit of creditors, makes or sends notice of a bulk transfer or calls a meeting of its creditors or principal creditors;

(g) A case or proceeding under the bankruptcy laws of the United States of America now or hereafter in effect or under any insolvency, reorganization, receivership, readjustment of debt, dissolution or liquidation law or statute of any jurisdiction now or hereafter in effect (whether at law or in equity) is filed against Borrower or any Obligor or all or any part of its properties and such petition or application is not dismissed within sixty (60) days after the date of its filing or Borrower or any Obligor shall file any answer admitting or not contesting such petition or application or indicates its consent to, acquiescence in or approval of, any such action or proceeding or the relief requested is granted sooner;

(h) A case or proceeding under the bankruptcy laws of the United States of America now or hereafter in effect or under any insolvency, reorganization, receivership, readjustment of debt, dissolution or liquidation law or statute of any jurisdiction now or hereafter in effect (whether at a law or equity) is filed by Borrower or any Obligor or for all or any part of its property; or

(i) Any default by Borrower or any Obligor under any agreement, document or instrument relating to any indebtedness for borrowed money owing to any person other than Agent and/or Lender, or any capitalized lease obligations, contingent indebtedness in connection with any guarantee, letter of credit, indemnity or similar type of instrument in favor of any person other than Agent and/or Lender, in any case in an amount in excess of One Hundred Thousand Dollars (\$100,000), which default continues for more than the applicable cure period, if any, with respect thereto, or any default by Borrower or any Obligor under any contract, lease, license or other obligation to any person other than Agent and/or Lender, which default continues for more than the applicable cure period, if any, with respect thereto;

(j) The occurrence of a Change in Control;

(k) The indictment or threatened indictment of Borrower or any Obligor under any criminal statute, or commencement or threatened commencement of criminal or civil proceedings against Borrower or any Obligor, pursuant to which statute or proceedings the penalties or remedies sought or available include forfeiture of any of the property of Borrower or such Obligor;

(1) There shall be a material adverse change in the business, assets or prospects of Borrower or any Obligor after the date hereof; or

 $({\tt m})$ $\,$ There shall be an Event of Default under, and as defined in, any of the other $\,$

10.2. Remedies.

(a) At any time an Event of Default exists or has occurred and is continuing, Agent shall have all rights and remedies provided in this Agreement, the other Financing Agreements, the Uniform Commercial Code and other applicable law, all of which rights and remedies may be exercised without notice to or consent by Borrower or any Obligor, except as such notice or consent is expressly provided for hereunder or required by applicable law. All rights, remedies and powers granted to Agent hereunder, under any of the other Financing Agreements, the Uniform Commercial Code or other applicable law, are cumulative, not exclusive and enforceable, in Agent's discretion, alternatively, successively, or concurrently on any one or more occasions, and shall include, without limitation, the right to apply to a court of equity for an injunction to restrain a breach or threatened breach by Borrower of this Agreement or any of the other Financing Agreements. Agent may, at any time or times, proceed directly against Borrower or any Obligor to collect the Obligations without prior recourse to the Collateral.

Without limiting the foregoing, at any time an Event of (b) Default exists or has occurred and is continuing, Agent may, in its discretion and without limitation, (i) accelerate the payment of all Obligations and demand immediate payment thereof to Agent (provided, that, upon the occurrence of any Event of Default described in Sections 10.1(g) and 10.1(h), all Obligations shall automatically become immediately due and payable), (ii) with or without judicial process or the aid or assistance of others, enter upon any premises on or in which any of the Collateral may be located and take possession of the Collateral or complete processing, manufacturing and repair of all or any portion of the Collateral, (iii) require Borrower, at Borrower's expense, to assemble and make available to Agent any part or all of the Collateral at any place and time designated by Agent, (iv) collect, foreclose, receive, appropriate, setoff and realize upon any and all Collateral, (v) remove any or all of the Collateral from any premises on or in which the same may be located for the purpose of effecting the sale, foreclosure or other disposition thereof or for any other purpose, (vi) sell, lease, transfer, assign, deliver or otherwise dispose of any and all Collateral (including entering into contracts with respect thereto, public or private sales at any exchange, broker's board, at any office of Agent or elsewhere) at such prices or terms as Agent may deem reasonable, for cash, upon credit or for future delivery, with Agent having the right to purchase the whole or any part of the Collateral at any such public sale, all of the foregoing being free from any right or equity of redemption of Borrower, which right or equity of redemption is hereby expressly waived and released by Borrower and/or (vii) terminate this Agreement. If any of the Collateral is sold or leased by Agent and/or Lender upon credit terms or for future delivery, the Obligations shall not be reduced as a result thereof until payment therefore is finally collected by Agent and/or Lender. If notice of disposition of Collateral is required by law, five (5) days prior notice by Agent to Borrower designating the time and place of any public sale or the time after which any private sale or other intended disposition of Collateral is to be made, shall be deemed to be reasonable notice thereof and Borrower waives any other notice. In the event Agent institutes an action to recover any Collateral or seeks recovery of any Collateral by way of prejudgment remedy, Borrower waives the posting of any bond which might otherwise be required.

(c) Agent may apply the cash proceeds of Collateral actually received by Agent and/or Lender from any sale, lease, foreclosure or other disposition of the Collateral to payment of the Obligations, in whole or in part and in such order as Agent may elect, whether or not then due.

Borrower shall remain liable to Agent and Lender for the payment of any deficiency with interest at the highest rate provided for herein and all costs and expenses of collection or enforcement, including attorneys' fees and legal expenses.

(d) Without limiting the foregoing, upon the occurrence of an Event of Default or an event which with passage of time would constitute an Event of Default, Agent may, at its option, without notice, (i) cease making Loans or reduce the lending formulas or amounts of Revolving Loans available to Borrower and/or terminate any provision of this Agreement providing for any future Loans to be made by Lender to Borrower.

SECTION 11. JURY TRIAL WAIVER; OTHER WAIVERS AND CONSENTS; GOVERNING LAW.

_ ____

(a) The validity, interpretation and enforcement of this Agreement and the other Financing Agreements and any dispute arising out of the relationship between the parties hereto, whether in contract, tort, equity or otherwise, shall be governed by the internal laws of the State of New York (without giving effect to principles of conflicts of law).

Borrower, Agent and Lender irrevocably consent and submit to (b) the non-exclusive jurisdiction of the state courts of New York County and the United States District Court for the Southern District of New York and waive any objection based on venue or forum non conveniens with respect to any action instituted therein arising under this Agreement or any of the other Financing Agreements or in any way connected with or related or incidental to the dealings of the parties hereto in respect of this Agreement or any of the other Financing Agreements or the transactions related hereto or thereto, in each case whether now existing or hereafter arising, and whether in contract, tort, equity or otherwise, and agree that any dispute with respect to any such matters shall be heard only in the courts described above (except that Agent and Lender shall have the right to bring any action or proceeding against Borrower or its property in the courts of any other jurisdiction which Agent deems necessary or appropriate in order to realize on the Collateral or to otherwise enforce its rights against Borrower or its property).

(c) Borrower hereby waives personal service of any and all process upon it and consents that all such service of process may be made by certified mail (return receipt requested) directed to its address set forth on the signature pages hereof and service so made shall be deemed to be completed five (5) days after the same shall have been so deposited in the U.S. mails, or, at Agent's option, by service upon Borrower in any other manner provided under the rules of any such courts. Within thirty (30) days after such service, Borrower shall appear in answer to such process, failing which Borrower shall be deemed in default and judgment may be entered by Agent and/or Lender against Borrower for the amount of the claim and other relief requested.

(d) BORROWER, AGENT AND LENDER EACH HEREBY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION (i) ARISING UNDER THIS AGREEMENT OR ANY OF THE OTHER FINANCING AGREEMENTS OR (ii) IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO IN RESPECT OF THIS AGREEMENT OR ANY OF THE OTHER FINANCING AGREEMENTS OR THE

TRANSACTIONS RELATED HERETO OR THERETO IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER IN CONTRACT, TORT, EQUITY OR OTHERWISE. BORROWER, AGENT AND LENDER EACH HEREBY AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY AND THAT BORROWER, AGENT OR LENDER MAY FILE AN ORIGINAL COUNTERPART OF A COPY OF THIS AGREEMENT WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE PARTIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

(e) Neither Agent nor Lender shall have any liability to Borrower (whether in tort, contract, equity or otherwise) for losses suffered by Borrower in connection with, arising out of, or in any way related to the transactions or relationships contemplated by this Agreement, or any act, omission or event occurring in connection herewith, unless it is determined by a final and non-appealable judgment or court order binding on Agent and/or Lender, that the losses were the result of acts or omissions constituting gross negligence or willful misconduct. In any such litigation, Agent and Lender shall be entitled to the benefit of the rebuttable presumption that it acted in good faith and with the exercise of ordinary care in the performance by it of the terms of this Agreement.

11.2. Waiver of Notices. Borrower hereby expressly waives demand, presentment, protest and notice of protest and notice of dishonor with respect to any and all instruments and commercial paper, included in or evidencing any of the Obligations or the Collateral, and any and all other demands and notices of any kind or nature whatsoever with respect to the Obligations, the Collateral and this Agreement, except such as are expressly provided for herein. No notice to or demand on Borrower which Agent and/or Lender may elect to give shall entitle Borrower to any other or further notice or demand in the same, similar or other circumstances.

11.3. Amendments and Waivers. Neither this Agreement nor any provision hereof shall be amended, modified, waived or discharged orally or by course of conduct, but only by a written agreement signed by an authorized officer of Agent and Lender (subject to Section 13.14 hereof), and as to amendments, as also signed by an authorized officer of Borrower. Neither Agent nor Lender shall, by any act, delay, omission or otherwise be deemed to have expressly or impliedly waived any of its rights, powers and/or remedies unless such waiver shall be in writing and signed by an authorized officer of Agent and Lender. Any such waiver shall be enforceable only to the extent specifically set forth therein. A waiver by Agent and/or Lender of any right, power and/or remedy on any one occasion shall not be construed as a bar to or waiver of any such right, power and/or remedy which Agent and Lender would otherwise have on any future occasion, whether similar in kind or otherwise.

11.4. Waiver of Counterclaims. Borrower waives all rights to interpose any claims, deductions, setoffs or counterclaims of any nature (other then compulsory counterclaims) in any action or proceeding with respect to this Agreement, the Obligations, the Collateral or any matter arising therefrom or relating hereto or thereto.

11.5. Indemnification. Borrower shall indemnify and hold Agent and Lender, and their directors, agents, employees and counsel, harmless from and against any and all losses, claims, damages, liabilities, costs or expenses imposed on, incurred by or asserted against any of them in connection with any litigation, investigation, claim or proceeding commenced or threatened related to the negotiation, preparation, execution, delivery, enforcement, performance or administration of

this Agreement, any other Financing Agreements, or any undertaking or proceeding related to any of the transactions contemplated hereby or any act, omission, event or transaction related or attendant thereto, including amounts paid in settlement, court costs, and the fees and expenses of counsel. To the extent that the undertaking to indemnify, pay and hold harmless set forth in this Section may be unenforceable because it violates any law or public policy, Borrower shall pay the maximum portion which it is permitted to pay under applicable law to Agent and Lender in satisfaction of indemnified matters under this Section. The foregoing indemnity shall survive the payment of the Obligations and the termination or non-renewal of this Agreement.

11.6. Waiver and Consent. Borrower waives notice of and hereby consents to, (i) extensions of time of payment of or increase or decrease in the amount of any of the Obligations, interest rate, fees, other charges, or the Collateral, (ii) the taking, exchange, surrender and releasing of the Collateral or guarantees now or at any time held by or available to Agent and/or Lender for the Obligations of any other party at any time liable on or in respect of the Obligations, (iii) the exercise of, or refraining from the exercise of any rights against any Borrower or the Collateral, (iv) the settlement, compromise or release of, or the waiver of an Event of Default with respect to, any of the Obligations and (v) any other act, event or condition that may, in law or equity, constitute a defense under applicable law to the obligation of a surety or guarantor. Borrower hereby irrevocably and unconditionally waives and relinquishes all statutory, contractual, common law, equitable and all other claims against Agent and/or Lender for subrogation, reimbursement, exoneration, contribution, indemnification, setoff or other recourse in respect to sums paid or payable to Agent and/or Lender by Borrower hereunder.

SECTION 12. TERM OF AGREEMENT; MISCELLANEOUS.

12.1. Term.

(a) This Agreement and the other Financing Agreements shall become effective as of the date set forth on the first page hereof and shall continue in full force and effect for a term ending on the date three (3) years from the date hereof (the "Initial Maturity Date"), and from year to year thereafter (said Initial Maturity Date and any anniversary thereof being referred to herein as the "Maturity Date"), unless sooner terminated pursuant to the terms hereof; provided, that, Agent and Lender may, at their option, extend the Maturity Date to the date four (4) years from the date hereof by giving Borrower notice at least sixty (60) days prior to the third anniversary of this Agreement. Agent and Lender or Borrower (subject to Agent's and Lender's right to extend the Maturity Date as provided above) may terminate this Agreement and the other Financing Agreements effective on the Maturity Date or on the anniversary of the Maturity Date in any year by giving to the other party at least sixty (60) days prior written notice; provided, that, this Agreement and all other Financing Agreements must be terminated simultaneously. Upon the effective date of termination or non-renewal of the Financing Agreements, Borrower shall pay to Agent for Lender, in full, all outstanding and unpaid Obligations and shall furnish cash collateral to Agent in such amounts as Agent determines are reasonably necessary to secure Agent and Lender from loss, cost, damage or expense, including attorneys' fees and legal expenses, in connection with any contingent Obligations, including any checks or other payments provisionally credited to the Obligations and/or as to which Agent and/or Lender has not yet received final and indefeasible payment. Such payments in respect of the Obligations and cash collateral shall be remitted by wire transfer in federal funds to such bank account of Agent, as Agent may, in its sole discretion, designate in writing to Borrower for such purpose. Interest shall be due until and including the next Business

Day, if the amounts so paid by Borrower to the bank account designated by Agent are received in such bank account later than 12:00 noon, Eastern time.

(b) Borrower may terminate this Agreement at any time before the Maturity Date upon not less than sixty (60) days prior written notice and compliance with the other requirements of this Section 12.1 (including payment of the fee described in Section 12.1(c)). No termination of this Agreement or the other Financing Agreements shall relieve or discharge Borrower of its respective duties, obligations and covenants under this Agreement or the other Financing Agreements until all Obligations have been fully and finally discharged and paid, and Agent's continuing security interest in the Collateral and the rights and remedies of Agent and Lender hereunder, under the other Financing Agreements and applicable law, shall remain in effect until all such Obligations have been fully and finally discharged and paid.

(c) If for any reason this Agreement is terminated prior to the Maturity Date, in view of the impracticality and extreme difficulty of ascertaining actual damages and by mutual agreement of the parties as to a reasonable calculation of Lender's lost profits as a result thereof, Borrower agrees to pay to Agent for Lender, upon the effective date of such termination, an early termination fee in the amount set forth below if such termination is effective in the period indicated:

	Amount 	Period
(i)	2.0% of the Maximum Credit.	From the date hereof to and including the first anniversary of this Agreement.
(ii)	1.0% of the Maximum Credit.	From the first anniversary of this Agreement to and including the second anniversary of this Agreement.
(iii)	.5% of the Maximum Credit	Any time after the second anniversary of this Agreement and prior to the then applicable Maturity Date.

Such early termination fee shall be presumed to be the amount of damages sustained by Lender as a result of such early termination and Borrower agrees that it is reasonable under the circumstances currently existing. In addition, Lender shall be entitled to such early termination fee upon the occurrence of any Event of Default described in Sections 10.1(g) and 10.1(h) hereof, even if Agent and Lender do not exercise their right to terminate this Agreement, but elects, at its option, to provide financing to Borrower or permit the use of cash collateral under the United States Bankruptcy Code. The early termination fee provided for in this Section 12.1 shall be deemed included in the Obligations. Notwithstanding the foregoing, Agent and Lender shall waive the early termination fee if Borrower refinances the Credit Facilities under this Agreement with Lender.

12.2. Notices. All notices, requests and demands hereunder shall be in writing and (a) made to Agent and Lender at its respective address set forth below and to Borrower at its chief executive office set forth below, or to such other address as either party may designate by written notice to the other in accordance with this provision, and (b) deemed to have been given or made: if delivered in person, immediately upon delivery; if by telex, telegram or facsimile transmission, immediately upon sending and upon confirmation of receipt; if by nationally recognized overnight courier service with instructions to deliver the next Business Day, one (1) Business Day after

sending; and if by certified mail, return receipt requested, five (5) days after mailing.

12.3. Partial Invalidity. If any provision of this Agreement is held to be invalid or unenforceable, such invalidity or unenforceability shall not invalidate this Agreement as a whole, but this Agreement shall be construed as though it did not contain the particular provision held to be invalid or unenforceable and the rights and obligations of the parties shall be construed and enforced only to such extent as shall be permitted by applicable law.

12.4. Successors. This Agreement, the other Financing Agreements and any other document referred to herein or therein shall be binding upon and inure to the benefit of and be enforceable by Agent and Lender, Borrower and their respective successors and assigns, except that Borrower may not assign its rights under this Agreement, the other Financing Agreements and any other document referred to herein or therein without the prior written consent of Agent. Agent and Lender may, after notice to Borrower, assign its respective rights and delegate its respective obligations under this Agreement and the other Financing Agreements and further may assign, or sell participations in, all or any part of the Loans, or any other interest herein to another financial institution or other person, in which event, the assignee or participant shall have, to the extent of such assignment or participation, the same rights and benefits as it would have if it were Agent or Lender, as applicable, hereunder, except as otherwise provided by the terms of such assignment or participation.

12.5. Entire Agreement. This Agreement, the other Financing Agreements, any supplements hereto or thereto, and any instruments or documents delivered or to be delivered in connection herewith or therewith represents the entire agreement and understanding concerning the subject matter hereof and thereof between the parties hereto, and supersede all other prior agreements, understandings, negotiations and discussions, representations, warranties, commitments, proposals, offers and contracts concerning the subject matter hereof, whether oral or written. In the event of any inconsistency between the terms of this Agreement and any schedule or exhibit hereto, the terms of this Agreement shall govern.

SECTION 13. AGENT

As between Agent, on one hand, and Lender, on the other hand, Agent and Lender agree as follows (with the consent and approval of Borrower):

13.1. Appointment and Authorization. Lender hereby irrevocably appoints and authorizes Agent to take such action on its behalf and to exercise such powers under this Agreement as are delegated to Agent by the terms hereof, together with such powers as are reasonably incidental thereto. Subject to the provisions of this Agreement, Agent will handle all transactions relating to the Loans and all other Obligations, including, without limitation, all transactions with respect to this Agreement, the Financing Agreements and all related documents in accordance with its usual banking practices. Except as otherwise expressly provided herein, Borrower is hereby authorized by Lender to deal solely with Agent in all transactions which affect Lender under this Agreement and the Financing Agreements. The rights, privileges and remedies accorded to Agent hereunder shall be exercised by Agent on behalf of Lender.

13.2. General Immunity. In performing its duties as Agent hereunder, Agent will take the same care as it takes in connection with loans in which it alone is interested. Subject to Section 13.6

of this Agreement, neither Agent nor any of its directors, officers, agents or employees shall be liable for any action taken or omitted to be taken by it or them hereunder or in connection herewith except as such action or omission are caused solely from its or their own gross negligence or willful misconduct unless such action was taken or omitted to be taken by Agent at the direction of Lender.

13.3. Consultation with Counsel. Agent may consult with legal counsel and any other professional advisors or consultants deemed necessary or appropriate and selected by Agent and shall not be liable for any action taken or suffered in good faith by it in accordance with the advice of such counsel.

13.4. Documents. Agent shall not be under a duty to examine into or pass upon the effectiveness, genuineness or validity of this Agreement or any of the Financing Agreements or any other instrument or document furnished pursuant hereto or in connection herewith, and Agent shall be entitled to assume that the same are valid, effective and genuine and what they purport to be. In addition Agent shall not be liable for failing to make any inquiry concerning the accuracy, performance or observance of any of the terms, provisions or conditions of such instrument or document.

13.5. Rights as a Lender. Agent shall have the same rights and powers hereunder as a Lender and may exercise the same as though it were not Agent. The term "Lender" or "Lenders" shall, unless the context otherwise indicates, include Agent in its individual capacity.

13.6. Responsibility of Agent. It is expressly understood and agreed that the obligations of Agent hereunder are only those expressly set forth in this Agreement and that Agent shall be entitled to assume that no Event of Default has occurred and is continuing, unless Agent has actual knowledge of such fact. Except to the extent Agent is required by Lender pursuant to the express terms hereof to take a specific action, Agent shall be entitled to use its discretion with respect to exercising or refraining from exercising any rights which may be vested in it by, or with respect to taking or refraining from taking any action or actions that it may be able to take under or in respect of, this Agreement and the Financing Agreements. Agent shall incur no liability under or in respect of this Agreement and the Financing Agreements by acting upon any notice, consent, certificate, warranty or other paper or instrument believed by it to be genuine or authentic or to be signed by the proper party or parties, or with respect to anything that it may do or refrain from doing in the reasonable exercise of its judgment, or that may seem to it to be necessary or desirable under the circumstances. It is agreed among Agent and Lender that Agent shall have no responsibility to carry out audits or otherwise examine the books and records or properties of Borrower, except as Agent in its sole discretion deems appropriate or as otherwise expressly required hereunder. The relationship between Agent and each Lender is and shall be that of agent and principal only and nothing herein shall be construed to constitute Agent a joint venturer with Lender, a trustee or fiduciary for Lender or for the holder of a participation therein nor impose on Agent duties and obligations other than those set forth herein.

13.7. Collections and Disbursements.

(a) Agent will have the right to collect and receive all payments of the Obligations, together with all fees, charges or other amounts due under this Agreement and the Financing Agreements, and Agent will remit to Lender all such payments actually received by Agent (subject to any required clearance procedures) in accordance with the settlement procedures established by

Agent from time to time.

(b) If any such payment received by Agent is rescinded or otherwise required to be returned for any reason at any time, whether before or after termination of this Agreement and the Financing Agreements, Lender will, upon written notice from Agent, promptly pay over to Agent the amount so rescinded or returned, together with interest and other fees thereon if also required to be rescinded or returned.

(c) All payments by Agent and Lender to each other hereunder shall be in immediately available funds. Agent will at all times maintain proper books of account and records reflecting the interest of Lender in the Loans, in a manner customary to Agent's keeping of such records, which books and records shall be available for inspection by Lender at reasonable times during normal business hours, at Lender's sole expense.

13.8. Indemnification. Lender hereby indemnifies Agent (and any issuer with respect to letters of credit) from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever that may be imposed on, incurred by or asserted against Agent (or issuer, as the case may be) in any way relating to or arising out of this Agreement or any other Financing Agreements or any action taken or omitted by Agent (or issuer, as the case may be) under or related to this Agreement or the Loans, provided that Lender shall not be liable for any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements resulting solely from Agent's (or issuer's, as the case may be) gross negligence or willful misconduct. Agent shall have the right to deduct, from any amounts to be paid by Agent to any Lender hereunder, any amounts owing to Agent by Lender by virtue of this paragraph.

13.9. Expenses. (a) All out-of-pocket costs and out-of-pocket expenses incurred by Agent and not reimbursed on demand by Borrower, in connection with the analysis, negotiation, preparation, consummation, creation, amendment, administration, termination, work-out, forbearance and enforcement of the Obligations (including, without limitation, audit expenses, counsel fees and expenditures to protect, preserve and defend Agent's and Lender's rights and interest under the Financing Agreements) shall be shared and paid on demand by Lender; (b) Agent may deduct from payments or distributions to be made to Lender such funds as may be necessary to pay or reimburse Agent for such costs or expenses.

13.10. No Reliance By execution of or joining in this Agreement, each Lender acknowledges that it has entered into this Agreement and the Financing Agreements solely upon its own independent investigation and is not relying upon any information supplied by or any representations made by Agent. Each Lender shall continue to make its own analysis and evaluation of Borrower. Agent makes no representation or warranty and assumes no responsibility with respect to the financial condition of Borrower, any Obligor or any Account debtor of Borrower; the accuracy, sufficiency or currency of any information concerning the financial condition, prospects or results of operations of Borrower; or for sufficiency, authenticity, legal effect, validity or enforceability of the Financing Agreements. Agent assumes no responsibility or liability with respect to the collectibility of the Obligations or the performance by Borrower under the Financing Agreements.

13.11. Reporting. During the term of this Agreement, and subject to Section 9.6, Agent will

promptly furnish each Lender with copies of all financial statements of Borrower to be delivered or obtained hereunder and such other financial statements and reports as Lender may reasonably request. Agent will immediately notify Lender when it receives actual knowledge of any Event of Default under the Financing Agreements.

13.12. Removal of Agent. Agent may resign at any time upon giving forty-five (45) days prior written notice thereof to Lender and Borrower. Agent may be removed as Agent hereunder upon the written consent of all Lenders exclusive of Agent upon the following: (i) willful misconduct in the performance of Agent's duties or responsibilities under this Agreement; or (ii) if a receiver, trustee or conservator is appointed for Agent or any state or federal regulatory authority assumes management or control of Agent or if, under applicable law, the administrative or discretionary duties and responsibilities of Agent hereunder become controlled by or subject to the approval of any state or federal regulatory authority. Upon any resignation or permitted removal of Agent, Lender may appoint a successor agent. Upon the acceptance of the appointment as a successor agent hereunder by such successor agent, such successor agent shall thereupon succeed to and become vested with all rights, powers, obligations and duties of the retiring Agent and the retiring Agent shall be discharged from its duties and obligations hereunder.

13.13. Action on Instructions of Lender. With respect to any provision of this Agreement, or any issue arising thereunder, concerning which Agent is authorized to act or withhold action by direction of Lender, Agent shall in all cases be fully protected in so acting, or in so refraining from acting, hereunder in accordance with written instructions signed by Lender. Such instructions and any action taken or failure to act pursuant thereto shall be binding on Lender.

13.14. Consent of Lender.

(a) Except as expressly provided herein, Agent shall have the sole and exclusive right to service, administer and monitor the Loans and the Financing Agreements, including without limitation, the right to exercise all rights, remedies, privileges and options under the Financing Agreements, and including without limitation the credit judgment with respect to the making of advances and the determination as to the basis on which and extent to which advances may be made.

(b) Notwithstanding anything to the contrary contained in subparagraph (a) above, Agent shall not, without the prior consent of Lender: (i) extend the Maturity Date, (ii) except as contemplated under this Agreement, reduce any interest rate applicable to the Loans, any fee payable hereunder or any fee for any letter of credit, (iii) increase the Maximum Credit, (iv) compromise or settle all or a portion of the Obligations, (v) release any Obligor from the Obligations except in connection with termination of the this Agreement and full payment and satisfaction of all Obligations, (vi) amend this Section 13.14(b), (vii) enter into any written amendment to any of the Financing Agreements; (viii) waive Borrower's compliance with the terms and conditions of the Financing Agreements or any Event of Default hereunder or thereunder, or (ix) consent to Borrower's taking any action which, if taken, would constitute an Event of Default under this Agreement or under any of the Financing Agreements.

(c) After an acceleration of the Obligations, Agent shall have the sole and exclusive right, after consultation (to the extent reasonably practicable under the circumstances) with Lender, and, unless otherwise directed by the Lender, to exercise or refrain from exercising any and all right,

remedies, privileges and options under the Financing Agreements and actions, including, without limitation, instituting and pursuing all legal actions brought against Borrowers or to collect the Obligations, or defending any and all actions brought by Borrower or other Person; or incurring expenses or otherwise making expenditures to protect the Loans, the Collateral or Lenders' rights or remedies.

(d) To the extent Agent is required to obtain or otherwise elects to seek the consent of Lenders to an action Agent desires to take, if Lender fails to notify Agent, in writing, of its consent or dissent to any request of Agent hereunder within five (5) days of Lender's receipt of such request, Lender shall be deemed to have given its consent thereto.

SIGNATURES ON FOLLOWING PAGE

]	IN WI	FNESS	WHEF	REOF	, P	Agent	:, Le	ender	and	Borrow	ver h	nave	caused	these	presents
to	be	duly	execu	uted	as	of	the	day	and	year	first	abov	ve wr	ritten.		

BORROWER:	OPTICAL CABLE CORPORATION					
Address for Notices:						
5290 Concourse Drive Roanoke, VA 24019 Attn: Mr. Neil D. Wilkin Facsimile:	By: Name: Title:					
AGENT:	CONGRESS AS AGENT	FINANCIAL CORPORATION,				
Address for Notices:						
1133 Avenue of the Americas New York, New York 10036 Attn: Ms. Terese Gatto Facsimile: 212.545.4283	Ву:	Thomas Ennis, Vice President				
LENDER:	WACHOVIA	BANK, NATIONAL ASSOCIATION				
Address for Notices:						
Wachovia Bank, National Association One First Union Center 301 South College Street Charlotte, NC 28288-0658 Attn: Mr. Forest Steele Facsimile: 704-374-6249	By: Name: Title:					